

Special Town Meeting

October 22, 2016

At the close of registration on October 12, 2016 there were 5876 registered voters.

Precinct 1 - 101

Precinct 2 - 48

Precinct 3 - 48

Head Teller: Jane M. Wetson

Appointed Tellers:

Left:

David G. Neill

104 Asbury St

William F. Sullivan

66 Woodbury St

Center:

Darcyll L. Dale

58 Woodbury St

Jane M. Wetson

261 Asbury St

Right:

Charles A Chivakos

27 Village Ln

Stephen A. Walsh

17 Hedding Ave

The Moderator declared a quorum present (75) and the Warrant returned showing it had been properly served, opened the Special Town Meeting at 9:05 A.M. with 197 voters checked and present.

Pledge of Allegiance

The Moderator announced that October 24, 2016 through November 4, 2016 early voting would occur from 8:00 to 4:30 pm. The election of November 8, 2016 would be at the Recreation Center. The overflow room was the school cafeteria with televised recording and voting ability.

The Moderator introduced Town officials: Town Clerk, Andrea Carlson; Assistant to the Selectmen, Annabella Baptista; Town Counsel, Donna Brewer; Town Manager, Michael Lombardo; Board of Selectmen: Chairman, Scott Maddern; Bill Wilson; and Jeff Hubbard; Finance and Advisory Committee: Chairman, David Wanger; John Pruellage; Philip Stearns; and Nick Tensen; and Planning Board Chairman, Jeff Meick. The Moderator acknowledged, the checkers and tellers as well as Chris Shailor, Bill Melville and the HWcam crew.

The Moderator read: "I have asked non-resident Town officials and those assisting the Selectmen or Finance Committee to be present here so they may be available to answer questions. Members of the press have also been allowed to be present. All other non-registered visitors must be seated in the visitors' area located in the back rows of the left section of the auditorium. The back rows have been taped off to encourage voters to sit closer so the counting of votes would be easier.

These proceedings were recorded so as to have an accurate record of the meeting and to help everybody know who is speaking and hear what is said. Therefore, if you want to speak, use the microphones provided and unless I recognize you by name, please give your name and address when you address the moderator and remarks cannot be made to other Town officials or voters.

Please keep comments precise and don't repeat comments of others. Keep comments within three minutes. There is an ipad stopwatch available and once everyone else has spoken, the speaker will be given another one minute to clarify a point or address a point after you have spoken. Feel free to come and ask question, but do not use questions to aggregate your position. Do not deal with personalities. No personal attacks and any personal references should be only to let everyone know to whom you are responding. Other than privileged motions or point of order, such as a voter cannot hear, I will not recognize anything shouted form the audience, if you wish to make a motion or ask questions, you must come to the microphone, identify yourself and make the motion. Following our tradition, I will not accept the motion unless I feel there is sufficient debate

You should pick up a voting card. Please note that some of the articles and motions require a two-thirds vote and one requires a nine-tenths vote. As with majority votes if there is any doubt as to if it is a majority vote, I would call for a hand held vote by the tellers. With regard to the nine-tenths vote, unless we have a unanimous response, I am required to count the vote. For Article 2-5 as it pertains to the new roof at the Patton Homestead, I have chosen to not serve as Moderator for that Motion. As Selectmen, I served as co-chair of the Patton Advisory Committee and conversed directly with Joanne Patton and feel after talking with Town Counsel, reviewing Town Meeting Time, and speaking with other moderators, that I should recuse myself in that situation. I think it is best to appoint a temporary moderator to oversee the article and this will allow me to speak on the floor and engage in the debate. This is a common practice of moderators and I am appointing Bruce Ramsey to serve as moderator for Article 2-5.

The procedures we will follow today, follow the Town By-laws, Town Meeting Time, and the traditions of the Town. I ask you to turn off or mute cell phones or other electronic devices. Finally, if at all possible, I ask that you stay with us for the entire session so you can complete our business with as much voter participation as possible."

SECTION 1: REPORTS AND PROCEDURES

Affordable Housing Trust Report

The Moderator said she recognized Peter Britton, Chair of the Affordable Housing Trust to give a report and noted that this was an information report only. No questions could be asked. Afterward, Peter Britton and the other members of the Affordable Housing Trust offered to stay after the meeting for a Q&A session if anyone was interested

Peter Britton said one of the Affordable Housing Trust's (AHT) purposes was to provide affordable housing. 1964 was the last time, the Town provided services for its seniors in regard to housing. Mr. Britton stated that it was time and added that the Town had a peculiar demographic with a high number of seniors. Like many communities, Hamilton was challenged to meet the 10% affordable threshold as part of its housing stock for seniors and families. The AHT had assembled a basket of suitable choices to be able to pursue. The Town of Hamilton was a pioneer in creating a host community agreement allowing to collaborate with Harborlight and Habitat for Humanity.

Peter Britton showed slides of Longmeadow, a proposed large scale affordable housing complex and the Winthrop School/COA/PSB site, a Smart Growth property (RR runs through town). A consultant was hired to see challenges of the Winthrop School site. Other slides included Gordon College which had indicated it was interested in working with the Town to provide a site for housing, 13 Essex St., which had been offered by its owners who were familiar with Harborlight, and Central Ave. which might be appropriate for Habitat for Humanity which dealt with one to two unit sites. According to Peter Britton, the AHT wanted to work hard to present the basket of choices to the Selectmen.

Board of Selectmen Report

The Moderator recognized Scott Maddern, Chairman of the Board of Selectmen.

Scott Maddern said: "Thank you Madam Moderator. Thank you for all attending and everyone who had a hand in organizing today's Special Town Meeting and our breakfast this morning. I'll give a brief update on a large number of topics

Communication: Regarding communications, we are pleased that residents are attending more Selectmen, Planning Board, Zoning Board, and Affordable Housing meetings. Planning Board meetings are now televised and we received a large amount of input about our trash program which helped us decide on the recent changes to weekly pickups. We have selected a vendor to help revise the town's website over the next six months so that you can find information more easily and engage virtually with new concepts and projects. There is still nothing like personal contact, so feel free to stop in at Town Hall, attend any of our many meetings or call any of the

hundreds of volunteers who serve on our multiple boards and committees. We are all neighbors and interested in your insights and opinions.

Financial Conditions: Regarding our financial condition, we had another excellent audit report for FY15 and hoping for more of the same for FY16. We also have the best bond rating possible which gets us the lowest borrowing costs available in the market. By all measure, out town is financially sound and we spend less in all departments except for schools than our neighboring towns. After consistent increases in our tax rates for many years through FY12, we have stabilized tax rates even though they are high. Today you will be asked to make some reductions that could lower our tax rate below \$17 per thousand of assessed value, approaching rates we haven't seen since FY10.

Housing: Under our new Senior Housing Bylaw, we added 12 new homes recently, more than double our average annual rate. Our total population hasn't changed much over the last decade and is expected to grow very modestly over the next few, but we have big changes within this population. Seniors are now almost a quarter of our population while school enrollment has declined more than 10%. We'll hear more today about affordable housing as we are highly susceptible to projects we may have little control over, sometimes referred to as 40B developments. We've reformed the Affordable Housing Trust, asked them to assess multiple locations around town, including Longmeadow and town-owned land, while reinforcing their working relationship with the Planning Board. We also performed a senior housing survey. I'm proud to announce that just this week we inked two Host Community Agreements, one with Habitat for Humanity who will build two homes on Asbury Street, and another with Harborlights Community Partners, a well-respected developer of affordable housing. These Host Community Agreements help align our local needs with quality developers for a long term relationship, a signal to other potential 40B developers that they need to work proactively with our town and neighborhoods. Local need is key, the law says that 40B developments need to align with our local needs, and for us the priority is rental units for seniors. While the State is impressed with our efforts, they measure progress in new affordable housing permits issued. We are not protected from unfriendly 40B projects until we have hundreds more affordable houses. I've lobbied extensively with the Commonwealth's Housing Department, with the great assistance of Representative Brad Hill, to ask them not to fund 40B developments that aren't in line with our local needs.

From our recent community survey, housing was a big issue. Cottage Housing was one of the top requests and that will be deferred by the Planning Board from today until next April, but they have drafts and the Selectmen are going to help keep the focus on new bylaws and have already provided a list of priorities to the Planning and Zoning Boards. Regarding the town-wide survey, a special thanks to the 1000 plus people who responded, twice the amount of our last survey, as well as Selectman Shawn Farrell and former Selectman Marc Johnson who lead the effort. Our boards and committees use the results of what you shared and the survey is accessible on our town's website.

Town Infrastructure: Regarding our town's infrastructure, the culvert and bridge replacement over the Miles River on Bridge Street is progressing well, we should have two lanes open again in a couple of weeks. Phase 1 and 2 of our water pipes replacement is progressing and we're repaving the streets that have new pipes. Those projects will replace our 60-70 years old, leaky

water pipes, ones that were only designed to last 30-40 years. We are assessing the water treatment plant, and that may need some upgrades as well.

I have another special thank you. We are searching for a new DPW Director and in the meantime Police Chief Russ Stevens has been leading that department and doing a fantastic job. If you are not aware, Chief Stevens is on the State Board of Police Chiefs, so not only does he and his department do a fabulous job for us locally, he's very well respected around the State. We rebuilt our pool because it was leaking so badly. We got a great deal on the pool, not only did it come in under budget, but our share was less than half of the total cost because we got Wenham to regionalize with us and we got State matches to our Community Preservation Act contribution. The pool open house was very well attended, everyone loving the layout, shallow/kiddie section and new bathhouse.

Our Town Manager did a superb job of maintaining our trash hauling services despite the vendor abruptly stopping pickups with almost no advanced notification. That company has since entered bankruptcy proceedings. We have a new trash vendor, providing weekly pickups for the same cost as bi-weekly but the costs are much high than the previous vendor, the reason for the increases in blue bag fees and large items. All the people who supported weekly pickups promised to continue to recycle, and we strongly encourage everyone to do so because each of our efforts reduces the overall cost of the waste program. We are simplifying the accounting and Annual Town Meeting approval of our waste program too, eliminating the Waste Enterprise Fund, an anachronism of the last decade and subject to constant complaints from our auditors and the State because we never collected enough in fees to balance our costs.

We added hundreds of acres of open space at Sagamore Hill, the largest area in Essex County overlooking beautiful Ipswich Bay by using our Community Preservations funds and matched by lots of private donations. Thanks to Essex County Greenbelt. We need to make some repairs on the Patton Homestead and we'll be asked to decide on that later this morning. And lastly on infrastructure, two possible cell towers are being designed on town-owned land, one behind the Public Safety Building and the other behind Town Hall. There is a private cell tower being considered off Asbury and Highland streets too. And lastly, you'll hear about potential Town Hall renovations later this morning.

HWRSD Agreement: Regarding our schools, we are justifiably proud of our district's high rankings both academically and athletically. Congratulations to the General high school girls cross country team which broke the national record for the most consecutive dual-meet wins with 134, and to Coach Steve Sawyer who encouraged our kids so effectively since 1979. As I mentioned earlier, our senior population is growing while our school enrollment has declined. The ratio of students is shifting to Wenham and the 3-year average enrollment determines the split of school costs. Last year Wenham had over a 9% increase in costs and is facing a budget deficit for next year. The School Committee is asking for a change in the allocation which we've discussed with FinCom, and have all agreed that a longer term of the average enrollment maybe 5 to 7 years is worth looking at, or even other more stable ways like the number of households in each town. We also asked that the ratio of representation on the School Committee, which now has 1 of 7 members from Hamilton be revised.

Administrative Item: For administrative matters, we have a number of policies: personnel; code of conduct; land acquisition; building use and the citizens' petition policy that was used recently for site plan review changes. A citizens' petition many years ago has led to today's decision for an Essex Street sidewalk.

Conclusion: And lastly, a reminder about voting. This year we begin using the Recreation Center, which adjoins the Library and will be closed for the day. The School District has been asking us for years to find alternate site. There will be signs at Winthrop School to redirect us to the Rec Center and special police details to help us too. The election is expected to have very heavy turnout so an easy alternative is to go to the Town Clerk's office and vote early, which you can do starting on October 24th. Thanks again for attending our special Town Meeting and to all our volunteers who help out with the many programs we have in Town. Thank you Madam Moderator."

ARTICLE 2016/10 1-1

The Moderator read: "To hear the reports of Town Officers and Selected Committee and to take action thereon or relative thereto. Reports will appear in the Town Report for calendar year 2016. Both the Board of Selectmen and the FinCom voted unanimously in favor for this motion."

Town Moderator recognized Scott Maddern.

Scott Maddern said: "I move that the reports of Town officers and committees be received and placed on file."

Seconded

Vote: Motion passes

SECTION 2: FINANCIAL ACTIONS

ARTICLE 2016/10-2-1 Prior Year Bills

The Moderator read: "To see if the Town will raise and appropriate or transfer from available funds a sum of money to pay utility, supplies, and services bills incurred in a prior fiscal year, or take any action thereon or relative thereto."

Scott Maddern stated that the Board of Selectmen voted in favor. David Wanger said the Finance and Advisory Committee voted in favor Moderator recognized David Wanger.

Motion moved by David Wanger who said: "I move that the Town raise and appropriate \$2,151.00 to pay bills for supplies and services incurred in FY '16."

Mr. Wanger invited the residents to attend FinCom meetings and said they currently had one vacancy. As to this particular warrant article, merchandise purchases or accounting services that were incurred in the last month of 2016 occurred but were not paid during that period.

Moderator said this article required a nine-tenths vote so it needed a unanimous vote by law. Vote: Motion carries. Nine-tenths requirement was met as declared by the Moderator

ARTICLE 2016/10 2-2 Classification/Compensation Table

The Moderator read: "To see if the Town will amend the Personnel By-law by adopting changes to the classification and compensation table, or take any action thereon or relative thereto. The Proposed Compensation/Classification Table is set forth in Appendix A to the Fall 2016 Warrant Book."

The Board of Selectmen voted 4-1 in favor.

The Finance and Advisory Committee voted unanimously in favor

Town Moderator recognized Bill Wilson.

Motion made by Bill Wilson who said: "I move that the Town amend the Personnel By-law by adopting the changes to the classification and compensation table for FY '17 as set forth in Appendix A to the Fall 2016 Warrant Book."

Seconded

Bill Wilson recognized Michael Lombardo

Michael Lombardo explained the wage and classification study which included internal inequity issues. The study was previously commissioned with negotiating contracts in process. The police union would keep its existing structure.

Vote: Motion carries.

ARTICLE 2016/10 2-3 FY'117 Budget Adjustments.

Town Moderator read: "To see if the Town will amend the Town operating budget for FY'17 by decreasing the appropriation for certain Town expenses, or take any action thereon or relative thereto."

The Selectmen recommended favorable action

The Finance and Advisory Committee recommended favorable action.

Town Moderator recognized David Wanger.

Motion made by David Wanger who said: "I move the Town amend the appropriations for certain Town expenses by the following actions:

Decrease General Town Government by \$20,000 to \$1,960,624

Decrease Public Safety by \$10,625 to \$2,638,803

Decrease Department of Public Works by \$8,300 to \$1,543,858

Decrease Recreation by \$1,100 to \$110,793.

If adopted it would result in a total appropriation for schools and all other Town expenses of for Fy'17 of \$28,391,204."

Seconded

Town Moderator asked David Wanger for a review of the topic to which he responded that the FinCom had asked the Town Manager to review fiscal FY'16 and it appeared that \$681,000 unexpended and unencumbered funds existed at the end of FY'16 with revenue coming at \$230,992 above expectations for a total excess of over \$900,000. Year-end amounts of \$200,000 might be acceptable but the \$900,000 was an unexpected amount. Fy'17 looked like there might be an excess due to underestimated revenue. The article determined reduced appropriations for itemized areas totaling about \$40,000 of revenue underestimated, about \$125,000, totaling a \$165,000 reduction to the tax rate of about .12 cents per thousand. The FinCom recommended reducing the tax rate using these funds for the last two quarters of FY'17. The rate would go

from 17.24 per thousand to 17.12. Mr. Wanger announced that the last article would further mitigate the tax rate. The FinCom thought there were lessons to be learned to tightening the Town's approach for appropriation and expected revenue would be in the future. Vote: Motion carries.

ARTICLE 2016/10 2-4 Appropriation for Waste/Recycling/Composting Contract
The Moderator read: "To see if the Town will raise and appropriate or transfer from available funds money to fund the remainder of the current fiscal year of the multi-year contract for collection and transportation of solid waste, recycling and composting, or take any action thereon or relative thereto."

The Moderator stated that both the Board of Selectmen and the Finance and Advisory Committee voted unanimous in favor of the motion.

Motion made by Jeff Hubbard who said: "I move that the Town appropriate and transfer from certified free cash \$180,000 to fund the remainder of the current fiscal year of the multi-year contract for collection and transportation of solid waste, recycling and composting." Seconded

Jeff Hubbard recognized Town Manager, Michael Lombardo.

Michael Lombardo described the situation when the previous trash vendor stopped collecting trash. Cosella was chosen after an RFP was put out to bid. The cost was \$180,000 more for the current fiscal year for recyclables tipping fee. Markets were upside down and became more expensive at \$5 to \$15 per ton. Mr. Lombardo said the Town was still negotiating as a five year contract would make the price go down. This Article was to fund the deficit as to what was budgeted in April.

Gretel Clark (823 Bay Road) offered an amendment to the motion to reinstitute a purchase bag requirement for residents who use the off week solid waste collection.

Motion made by Gretel Clark: "I move that the Town appropriate and transfer from certified free cash \$180,000 to fund the remainder of the current fiscal year of the multi-year contract for collection and transportation of solid waste, recycling and composting, conditioned upon reconstituting a purchased bag requirement for residents who use the off week solid waste collection."

Seconded

Town Moderator noted that the main motion is now that of Gretel Clark's.

Gretel Clark stated there were over 100 communities where residents paid for every ounce of their solid waste. Solid waste pick up had been every other week for the last few years in Hamilton. The new trash program chose to pick up every other week. Ms Clark said there was free coaching for recycling and asked everyone to compare Hamilton's waste and recycling with Wenham's noting that the Town of Hamilton's recycling was double and trash was half that of Wenham's. Ms Clark said it was only fair that those who did not take advantage of recycling should be the ones who pay for solid waste pick up.

Matt Nevins (10 Bittersweet Lane) stated that he was opposed because there was not enough

notice given by the Town to vote on something of this nature so everyone could express their view rather than a simple transfer of funds. Jeff Filapod (18 Gifford Road) thought it was confusing and messy and suggested purchasing stickers as opposed to bags. Roland Brook (466 Essex St. said the current vendor did not have adequate equipment and asked if there was additional cost to the Town.

Michael Lombardo said the prices were the same weekly and biweekly. Blue bag revenue went from \$32,000 to \$76,000 per year and the recycling rate was 52%. If residents don't recycle rigorously, there would be a greater cost. Michael Lombardo explained the \$180,000, a 40% increase was an additional cost. Scott Maddern thought it was impractical as there was no off week and the concept of paying more or less based on consumption like schools, was not appropriate as it didn't matter what you pay compared to what you use. Bill Wilson said the amendment was not the same intent as the original motion and did not have reasonable notice to the entire population.

Gretel Clark said unless you read facebook, you wouldn't know this was before the Town. Lynnette Fallon asked if the original schedule was part of a Town Meeting vote. Michael Lombardo responded that the Board of Selectmen had the authority to set policy to go to biweekly and assumed it was set by the Selectmen five years ago. Scott Maddern said Wenham had dramatically increased their recycling. The Moderator read the By-law (Section 2) which indicated that the Board of Selectmen had the authority to set refuse policy and her recollection was that Selectmen had implemented bi-weekly.

Bill Wilson said the article was to discuss appropriations not frequency. The Moderator said she had the discretion to change an article and noted Moderator leniency on the ask because she believed it was within the scope. Michael Lombardo compared the possibility of this being a binding vote versus advisory vote which would be clarified after Town Meeting as it may counter a standing By-law that allowed the Selectmen to make policy. Richard Lowe (938 Bay Road) wanted to know if the Town would still be constrained to one black barrel and Michael Lombardo responded that it would be the same constraint. Jerry Fallon (38 Union St.) said he placed black bags next to his barrel and they were picked up, to which Mr. Lombardo said he would correct.

The Moderator repeated the amended motion which was now the main motion: "That the Town appropriate and transfer from certified free cash \$180,000 to fund the remainder of the current fiscal year of the multi-year contract for collection and transportation of solid waste, recycling and composting, conditioned upon reconstituting a purchased bag requirement for residents who use the off week solid waste collection."

Vote: Motion did not pass

The Moderator announced that the main motion was on the table without any conditions.

Vote: Motion passes.

ARTICLE 2016/10 2-5 Community Preservation Fund Projects

The proposed projects are set for in Appendix B to the Fall 2016 Warrant Book. The Moderator said she would appoint former Town Moderator, Bruce Ramsey to act as moderator for Article 2-5

Interim Moderator read: "Please turn to Appendix B of the Fall 2015 Warrant Book. Since there are only two projects, we will act on the projects under separate motions. I therefore recognize Jay Butler for the first motion."

Historic Preservation - Town Hall Preservation Project.

Motion made by Jay Butler (78 Old Cart Road and Member of the CPC) who said: "I move that the Town appropriate \$45,000 from the Community Preservation Fund Balance (Unreserved) to fund the repair of the roof and dormers of the Patton Homestead, conditioned upon commencement of construction by June 30, 2017."

Seconded

Jay Butler reviewed the CPC funding and showed a chart that explained how the state matching funding worked. Mr. Butler explained that if the percentage paid by the Town went up, the match would also go up. Further charts showed where the buckets of money were spent and if project were approved at Town Meeting how much money would be left.

The Appointed Moderator reminded the Town that this vote was for the Town to appropriate \$45,000 from the Community Preservation Fund Balance (Unreserved) to fund the repair of the roof and dormers of the Patton Homestead, conditioned upon commencement of construction by June 30, 2017.

Vote: Motion passes

The Appointed Moderator returned the gavel to Town Moderator, Jennifer Scuteri.

Town Moderator read: "For the second motion under this article, I recognize Jay Butler."

Motion made by Jay Butler who said: "I move that the Town appropriate \$250,000 (\$185,000 from the Community Preservation Fund Balance (Unreserved) and \$65,000 from the Historic Preservation Reserves) to fund the historic preservation analysis and evaluation of existing condition of Town Hall, conditions upon completion of the analysis by June 30 2019." Seconded

Jay Butler said this project had been one of the larger and popular requests for the total ask of \$550,000, of which \$430,000 was eligible last time under CPC. The original approval by the CPC included Phase I was for the determination of conditions and analysis for \$250,000 and Phase II was an ask of \$180,000 with bonding for the rest to be requested later.

David Wanger said the FinCom recommended favorable action for this and the following article because there were environmental, workplace, and safety concerns. There would be CPC state matching for funding but the state match was not guaranteed, especially if cities like Boston join in which case it would reduce the money available which should be considered when comparing a 2% to 3% increase in the future.

Bob DeFelice (Cutler Road) asked if the \$250,000 was an estimate or fixed amount as it was a lot of money for an analysis to which Michael Lombardo said \$550,000 was needed to get to the bid documents. Once documents were developed for renovations and repairs were completed, the Town could decide if they wanted to support the costs of the project. An OPM would be needed even before design was started. Doug Trees (557 Bay Road) said the total cost would be 5-6M and he thought there should be a renovation committee as well as having the town agree to do a major project as a way to get residents involved. Mr. Trees said spending \$500,000 already commits the Town to the project. Virginia Cookson (318 Forest St.) and Alisa Royal (153 Highland St.) expressed that they were not in favor of the project.

John Serafini (601 Bay Road) reminded the Town that there was a model in this town that had a lot of talented volunteers to guide the process to consider the project. The model would get residents to study deficiencies for free to get to a point as to whether the Town would consider the project and then it could be defined what task was required for consultants before committing to \$550,000. Jeff Filipod (18 Gifford) suggested building a new town hall and fix up the existing as a historic monument. Michael Lombardo said that he and Patrick Reffett had both developed 100's of 1,000's of square feet of office space. Mr. Lombardo reminded residents that this was a new government of Town Manager and there would be public input and engagement of the Board of Selectmen and FinCom in public session. Mr. Lombardo explained the process and the issues that would require renovations and expansion of the space.

Jeanne Miranda (21 Garfield Ave.) said retrofitting didn't work and noted the historical value of the property versus daily use and Michael Madden (231 Highland St) said he was a project manager and was more than happy to act as project manager adding that there were resources in town and said there should not be an ask for money before the project was approved. Bob DeFelice (341 Cutler Rd.) spoke against the proposal recalling his work on the public safety building and the volunteers that had worked to save the town money. Vote: Motion fails

ARTICLE 2016/10 2-6 Town Hall Renovations Evaluation and Design

The Moderator asked the audience to read: "To see if the Town will appropriate a sum of money for the purpose of obtaining evaluations of the physical conditions and systems of the Hamilton Town Hall, and for the purpose of obtaining schematic designs, architectural and engineering designs, cost estimates, bid documents relating to the renovation, restoration, expansion, construction and reconstruction of the Hamilton Town Hall, including all costs for services incidental or related thereto, including, but not limited to, services of architects, engineers, and project managers, and further to determine whether this appropriation shall be raised by borrowing or otherwise; and if by borrowing to raise and appropriate a sum of money for bond issuance costs, or take any action thereon or relative thereto."

Scott Maddern stated that the Board of Selectmen requested to take no action. Seconded.

Vote: Approved to take no action

<u>ARTICLE 2016/10 2-7</u> Engineering Design for Essex Street Pedestrian and Bike Path The Moderator read: "To see if the Town will raise and appropriate or transfer from available funds a sum of money to pay for engineering design for the Essex Street Pedestrian and Bike Path or take other action thereon or relative thereto."

The Board of Selectmen recommend favorable action

The Finance and Advisory Committee recommended unfavorable action.

Town Moderator recognized Scott Maddern

Motion made by Scott Maddern who said: "I move that the Town raise and appropriate or transfer from available funds \$40,000 to pay for engineering design for the Essex Street Pedestrian and Bike Path."

Seconded

Scott Maddern recognized Michael Lombardo. Mr. Lombardo said there had been a review and the next step was to engineer the path with the sidewalk going from the Essex town line to the Wenham town line.

Bob Gray (Bay Road) recalled that at Town Meeting of May 12, 2012, the Town approved \$19,000 for a study and asked if this was a duplicate. Michael Lombardo said this was the next step for engineering specs and the path's construction costs would be \$800,000 to \$1M. Bob Gray said the total was \$19,000 with an additional \$40,000 for the study of a path.

David Wanger explained the FinCom's rationale, which was beyond fiscal issues as the Committee had a duty to look at any and all municipal recommendations. The FinCom did not believe the sidewalk would be used by bikers, only children in which there were safety concerns due to multiple crossing, which were required due to topography. Kids on bikes crossing Essex St was problematic. Regarding the fiscal concerns, the cost would not be just the \$19,000 and the \$40,000 or the \$800,000 to 1M impact along with other expenses. Maintenance such as snow removal, adding an employee at \$20,000 and need to purchase equipment. Mr. Wanger said it was a mistake to view the ask in a vacuum due to safety and convenience as this would apply to many streets without sidewalks which might be brought forward based on Essex St. The FinCom's role was to maintain level services and approach level funding. This might initiate inconsistent and similar asks.

Scott Maddern said it was a citizen's petition brought forward by Town citizens regarding safety issues as the road had dangerous curves and speeding. Rosemary Huett (24 Birch Road) and Hannah Fraley (Old Cart Road) both spoke in favor of the article. Bob DeFelice (Cutler Road) said he thought \$40,000 for an analysis was a lot as it was a sidewalk. Scott Maddern responded that serious engineering was required to create bid documents adding the sidewalk would be built in increments. Virginia Cookson (318 Forest St.) spoke against the article, while Robin David (Lois St.) recalled that a child had died on Essex St, but was reminded it was in the Sagamore/Moulton St. area. Jeff Hubbard said the Board of Selectmen were not unanimous and he had voted no after considering the \$800,000 to 1M cost.

Vote: Motion did not carry.

SECTION 3: PLANNING / ZONING ACTIONS

ARTICLE 2016/10 3-1 Zoning By-law Amendment (Phase I)

The Moderator read: "To see if the Town will amend the Zoning By-law by deleting the existing text and replacing it with the language set forth in Appendix C, or take any action thereon or relative thereto."

The proposed By-Law is set forth in Appendix C of the 2016 Fall Warrant book

The Board of Selectmen recommended favorable action.

The Finance and Advisory Committee recommended favorable action.

Seconded

The Moderator read: "You should all have a copy of Appendix C. There are copies available on the table in the lobby. There are some amendments to the Appendix which Chairman Melick will review in his discussion of the motion. An errata sheet identifying these changes was also available on the table in the lobby and you should all have picked up that as well." The Moderator recognized Jeff Melick.

Motion made by Jeff Melick who said: "I move that the Town amend the Zoning by-law by deleting the current language in its entirety and replacing it with the language set forth in Appendix C of the 2016 Fall Warrant Book, with the amendment as voted by the Planning Board, a copy of which will be provided to the Town Clerk for inclusion in the record of this Town Meeting, and leaving the Zoning Map unchanged."

The Moderator noted that this motion required a two-thirds vote.

Jeff Melick introduced Mark Brobowski and thanked Ann Gero and Patrick Reffett for their help in rewriting the By-laws. Mr. Melick said the Planning Board was not interested in making substantial changes in this Phase. Mark Brobowski said the task was a recodification to keep the By-law consistent with recent state changes, fix internal inconsistencies, and change definitions such as nursing homes being changed to long term care facilities. There were no boundary or dimensional changes according to Mr. Brobowski. Heather Keane (81 Old Cart Road) asked if signage had been considered and Mr. Melick said substantial changes would be due in Spring as Phase II. Lidia Szydlowska (Asbury St) spoke about the 200'setback for a cell tower to which Town Moderator responded that there were no dimensional changes under consideration. Mr. Melick said the changes were to make it easier for safety and Federal law.

Robin Sears (14 Bridge St.) inquired about substantive changes to which Jeff Melick responded that any substantive changes were relatively minor for the legal purposes. Virginia Cookson (318 Forest St.) asked about Section 11A and 11B which were deleted in the rewrite, and were regarding Land Subject to seasonal or periodic flooding, which were not addressed anywhere else. Mark Bobrowski said they were addressed in the Wetlands Act so were covered by Conscom. Rick Mitchell (36 Rock Maple) said the Board had been working on this for 1.5 years and urged the Town to pass the article.

Rosemary Kennedy (61 Rust St.) asked about the Table of Use Regulations and if the Business District including Willow and Railroad Avenue. She was concerned that two or more units on the second floor and above had not heard much discussion and there was no maximum. Patrick Reffett said it had been discussed at length and voted upon a few Town Meetings ago. The upper floor residential use was allowed but the HDC requested more specific language because multifamily was not an allowed use in the Business District. Virginia Cookson (318 Forest St.) responded to Section 1. 1-B and said she disagreed with Mr. Mark Bobrowski because the Wetland Regulations did not cover it. Rosemary Kennedy said Section 3 was a late entry and wondered how many units were allowed and the height defined. Ms Kennedy referred to the survey which indicated that people wanted to maintain a small village feel in the downtown. Mark Brobowski responded that mixed uses were part of Site Plan Review.

Vote: Motion carries. Two thirds requirement was met as declared by the Moderator

ARTICLE 2016/10 3-2 Cottage Housing Development Overlay District

The Moderator asked the community read the article: "To see if the Town will amend the Zoning By-law by adding a new section to provide for cottage housing development by adopting the "Cottage Housing Development Overlay District, set forth in Appendix D, or take any action thereon or relative thereto."

The proposed by law was set forth in Appendix D of the 2016 Fall Warrant Book Motion made by Jeff Melick who said: "I move that the Town take no action on this article." Seconded

Jeff Melick said the Planning Board wanted to bring the By-law to the Town at Spring Town Meeting.

Vote: Motion carries to take no action.

SECTION 4: TOWN BY-LAW AMENDMENTS

ARTICLE 2016/10 4-1 Regulation of Roosters

The Moderator asked the community to read: "To see if the Town will amend the Town Bylaws, Ch. XVIII, the Animal Control By-law, by adding a new Section 9 to read: Keeping of Roosters. It shall be unlawful to keep roosters on any property within the Town of Hamilton, except for commercial agriculture property pursuant to G.L. c 40A sec. 3 and the Town's Zoning By-law. Property owners in violation of this By-law are subject to a fine of \$50 per rooster first offense, \$75 per rooster second offense, \$100 per rooster third and further offenses, with each day constituting a separate offense, or take any action thereon or relative thereto."

Motion made by Scott Maddern who said: "I move that the Town take no action on this article." Seconded

Scott Maddern suggested neighbors talk to each other and the Agricultural Board was reviewing the rules and regulations with some teeth in it.

Bob DeFelice made motion and said: "I move the Town amend the Town By-laws, Ch. XVIII, the Animal Control By-law, by adding a new Section 9 to read: Keeping of Roosters. It shall be unlawful to keep roosters on any property within the Town of Hamilton except for commercial agriculture property pursuant to G.L. c 40A sec. 3 and the Town's Zoning By-law. Property

owners in violation of this By-law are subject to a fine of \$50 per rooster first offense, \$75 per rooster second offense, \$100 per rooster third and further offenses, with each day constituting a separate offense."

No second

Town Moderator said there was no second so the Town would return to the main motion. The main motion was to take no action.

Vote: Motion carries to take no action.

SECTION 5: OTHER APPROPRIATIONS AND ACTIONS

ARTICLE 2016/10 5-1 Sale of Town Land

The moderator read: "To see if the Town will authorize the Selectmen to sell, restrict or otherwise dispose of two parcels of land off of Central Avenue and Baker Avenue, identified as Lots 84 and 85 on Assessors Map 47, on such terms as may be acceptable to the Selectmen, or take any action thereon or relative thereto."

The Board of Selectmen recommended favorable action
The Finance and Advisory Committee recommended favorable action

Motion made by Bill Wilson who said: "I move that the Town authorize the Board of Selectmen to sell, restrict or otherwise dispose of two parcels of land off of Central Avenue and Baker Avenue, identified as Lots 84 and 85 on the Assessors Map 47, on such terms as may be acceptable to the Selectmen."

Seconded

Bill Wilson recognized Michael Lombardo.

Michael Lombardo said it was acquired town-owned land and the Town needed to dispose of it. There were no evaluation as to whether it was buildable, but when asked to put it out for public auction or bid, there was not enough value as indicated by an Auctioneer.

Amy Howell (29 Central Ave) described the neighborhood, built in the 1800's adding that the roadway widths were 17' and there was a safety issue in the neighborhood. Ms Howell noted that there was a 40B project where their park use to be located. Article 5-2 of 2005 requested the same land be disposed of for a 40B and it was rejected. Jeff Filopod (18 Gifford Road) agreed and added the 40B project was in disrepair and the police visited frequently. Alisa Roar (153 Highland St.) asked about a plan and potential developers.

Michael responded that several parcels were looked at for disposal. A RFP would be used to sell the parcel. Donna Brewer said restrictions could be put in place through the RFP process. Scott Maddern said no plans were in place but there was an effort to look at all town owned properties but this was not his favorite site for affordable housing. Mr. Taylor (Central Ave.), Chris Lelo (17 Central Ave.), and Anne Brady (12 Ricker Circle) spoke against the proposed article.

6—2016 Special Town Meeting October 22, 2016

Jeanne Moran (21 Garfield Ave) said the neighborhood was cottage style living and said: "I move the Town designate lots 84 and 85 Assessor's map 47 as conservation land in perpetuity." Town Moderator said there was a request to amend the main motion.

Seconded

Vote: Motion carries to amend the motion.

Motion made by Jeanne Moran who said: "that the Town designate lot 84 and 85 on Assessor's map 47 as conservation land in perpetuity."

Second

Virginia Cookson wanted a definition of conservation uses and if the custody and control would be by the Conscom. Town counsel responded that there was a definition in statue and by-laws that the care, custody, and control was under the conservation commission which would be used for passive recreation and open space

Vote: Motion passes and declared a two thirds vote by the Moderator

ARTICLE 2016/10 5-2 Grant of Easement for Utility Purposes

The moderator read: "To see if the Town will grant an easement on terms acceptable to the Board of Selectmen and such terms and conditions as are required by law, for an abutter to Town Hall to install an electrical line underground from a utility pole on town land to an abutter's property, or take any action thereon or relative thereto."

The Board of Selectmen recommended favor action
The Finance and Advisory Committee recommended favorable action.

Motion made by Jeff Hubbard who said: "I move that the Town grant an easement, on terms acceptable to the Board of Selectmen and such terms and conditions as are required by law, for an abutter to Town Hall to install an electrical line underground from a utility pole on town land to an abutter's property."

Seconded

The moderator added that this motion requires a two thirds vote.

Michael explained it was a utility easement to allow electrical access for the building next door.

Vote Motion carries. Two-thirds requirement was met as declared by the Moderator

SECTION 6: CLOSING FINANCIAL ACTIONS

ARTICLE 2016/10 6-1 Free Cash Application

The moderator read: "To see if the Town will reserve, appropriate and authorize the Assessors to use available certified free cash to reduce the tax rate for the Fiscal Year beginning July 1, 2016, or take any other action thereon or relative thereto."

The Board of Selection recommended favorable action.

The Finance and Advisory Committee recommended favorable action.

Motion made by David Wanger who said: "I move that the Town reserve, appropriate, and authorize the Assessors to use \$250,000 of available certified free cash to reduce the tax rate for the Fiscal Year beginning July 1, 2016."

Seconded

David Wanger explained they wanted to maintain a 2.5M reserve to maintain stability and confront the school budget which was 75% of the town budget. The motion would allow the tax rate for the remainder of FY'17 to be \$16.95

Vote: Motion carries.

ADJORNMENT

The Moderator announced that the Special Town Meeting was dissolved at 12:25pm.

A TRUE COPY: ATTEST:

Andrea J. Carlson
Town Clerk

FY17 Wage Grid

		Minimum							Maximum
Grade	****	Step 1	Step 2	Step 3	Step 4	Step 5	Step 6	Step 7	Step 8
	Hrly	\$9:36	\$9.55	\$9.74	\$9.93	\$10.13	\$10.33	\$10.54	\$10.75
	37.5/wk	\$351.14	\$358.13	\$365.25	\$372.38	\$379.88	\$387.38	\$395.25	\$403.13
1 1	52/yr	\$18,259.28	\$18,622.76	\$18,993.00	\$19,363.76	\$19,753.76	\$20,143.76	\$20,553.00	\$20,962.76
-	40/wk	\$374.54	\$382.00	\$389.60	\$397.20	\$405.20	\$413.20	\$421.60	\$430.00
	52/yr	\$19,476.08	\$19,864.00	\$20,259.20	\$20,654.40	\$21,070.40	\$21,486.40	\$21,923.20	\$22,360.00
	Hrly	\$10.30	\$10.51	\$10.71	\$10.92	\$11.14	\$11.36	\$11.59	\$11.83
_	37.5/wk	\$386.25	\$394.13	\$401.63	\$409.50	\$417.75	\$426.00	\$434.63	\$443.63
2	52/yr	\$20,085.00	\$20,494.76	\$20,884.76	\$21,294.00	\$21,723.00	\$22,152.00		\$23,068.76
	40/wk	\$412.00	\$420.40	\$428.40	\$436.80	\$445.60	\$454.40		\$473.20
	52/yr	\$21,424.00	\$21,860.80	\$22,276.80	\$22,713.60	\$23,171.20	\$23,628.80	\$24,107.20	\$24,606.40
•	Hrly	\$12.36	\$12.61	\$12.85	\$13.10	\$13.37	\$13.63	\$13.91	\$14.20
	37,5/wk	\$463.50	\$472.88	\$481.88	\$491.25	\$501.38	\$511.13	\$521.63	\$532.50
3	52/yr	\$24,102.00	\$24,589.76	\$25,057.76	\$25,545.00	\$26,071.76	\$26,578.76	\$27,124.76	\$27,690.00
_	40/wk	\$494.40	\$504.40	\$514.00	\$524.00	\$534.80	\$545.20		\$568.00
	52/yr	\$25,708.80	\$26,228.80	. \$26,728.00	\$27,248.00	\$27,809.60	\$28,350.40	\$28,932.80	\$29,536.00
	Hrly	\$15.45	\$15.76	\$16.06	\$16.38	\$16.71	\$17:04	\$17.39	\$17.75
	37.5/wk	\$579.38	\$591.00	\$602.25	\$614.25	\$626.63	\$639.00	\$652.13	\$665.63
4	52/yr	\$30,127.76	\$30,732.00	\$31,317.00	\$31,941.00	\$32,584.76	\$33,228.00	\$33,910.76	\$34,612.76
	40/wk	\$618.00	\$630.40	\$642.40	\$655.20		\$681.60		\$710.00
	52/yr	\$32,136.00	\$32,780.80	\$33,404.80	\$34,070.40	\$34,756.80	\$35,443.20	\$36,171.20	\$36,920.00
	Hrly	\$18.39	\$18.75	\$19.11	\$19.49	\$19.88	\$20.28	\$20.69	\$21.12
	37.5/wk	\$689.63	\$703.13	\$716.63	\$730.88	\$745.50	\$760.50	\$775.88	\$792.00
5	52/yr	\$35,860.76	\$36,562.76	\$37,264.76	\$38,005.76	\$38,766.00	\$39,546.00		\$41,184.00
l.	40/wk	\$735.60	\$750.00	\$764.40	\$779.60	\$795,20	\$811.20		\$844.80
	52/yr	\$38,251.20	\$39,000.00	\$39,748.80	\$40,539.20	\$41,350.40	\$42,182.40	\$43,035.20	\$43,929.60
	Hrly	\$21.61	\$22.03	\$22.45	\$22.90	\$23.36	\$23.83	\$24.31	\$24.82
	37.5/wk	\$810.38	\$826.13	\$841.88	\$858.75	\$876.00	\$893.63	\$911.63	\$930.75
6	52/yr	\$42,139.76	\$42,958.76	\$43,777.76	\$44,655.00	\$45,552.00	\$46,468.76		\$48,399.00
	40/wk	\$864.40	\$881.20	\$898.00	\$916.00	\$934.40	\$953,20		\$992.80
l	52/yr	\$44,948.80	\$45,822.40	\$46,696.00	\$47,632.00	\$48,588.80	\$49,566.40	\$50,564.80	\$51,625.60
	Hrly	\$22.37	\$22.80	\$23.24	\$23.70	\$24.18	\$24.66	\$25.16	
	37.5/wk	\$838.88	\$855.00	\$871.50	\$888.75	\$906.75	\$924.75	\$943.50	\$963.38
7	52/yr	\$43,621.76	\$44,460.00	\$45,318.00	\$46,215.00	\$47,151.00	\$48,087.00		\$50,095.76
'	40/wk	\$894.80	\$912.00	\$929.60	\$948.00	\$967.20	\$986.40		
	52/yr	\$46,529.60	\$47,424.00	\$48,339.20	\$49,296.00	\$50,294.40	\$51,292.80	\$52,332.80	\$53,435.20

6--2016 Special Town Meeting October 22, 2016

	Hrly	\$25.17	\$25.65	\$26.15	\$26.66	\$27.20	\$27.74	\$28.31	\$28,90
	37.5/wk	\$943.88	\$961.88	\$980.63	\$999.75	\$1,020.00	\$1,040.25	\$1,061.63	\$1,083.75
8	52/yr	\$49,081.76	\$50,017.76	\$50,992.76	\$51,987.00	\$53,040.00	\$54,093.00	\$55,204.76	\$56,355.00
	40/wk	\$1,006.80	\$1,026.00	\$1,046.00	\$1,066.40	\$1,088.00	\$1,109.60	\$1,132.40	\$1,156.00
	52/yr	\$52,353.60	\$53,352.00	\$54,392.00	\$55,452.80	\$56,576.00	\$57,699.20	\$58,884.80	\$60,112.00
	Hrly	\$26:43	\$26.93	\$27.46	* \$27/99 · · ·	\$28.56	\$29:13	\$29.73	\$30:35
_	37.5/₩k	\$991.13	\$1,009.88	\$1,029.75	\$1,049.63	\$1,071.00	\$1,092.38	\$1,114.88	\$1,138.13
9	52/yr	\$51,538.76	\$52,513.76	\$53,547.00	\$54,580.76	\$55,692.00	\$56,803.76	\$57,973.76	\$59,182.76
	40/wk	\$1,057.20	\$1,077.20	\$1,098.40	\$1,119.60	\$1,142.40	\$1,165.20	\$1,189.20	\$1,214.00
	52/yr	\$54,974.40	\$56,014.40	\$57,116.80	\$58,219.20	\$59,404.80	\$60,590.40	\$61,838.40	\$63,128.00
	Hrly	\$28.28	\$28.82	\$29.38	\$29,95	\$30.56	\$31.17	\$31.81	\$32.47
40	37.5/wk	\$1,060.50	\$1,080.75		\$1,123.13	\$1,146.00			\$1,217.63
10	52/yr	\$55,146.00	\$56,199.00		\$58,402.76				\$63,316.76
	40/wk	\$1,131.20	\$1,152.80		\$1,198.00				\$1,298.80
	52/yr	\$58,822.40	\$59,945.60	\$61,110.40	\$62,296.00	\$63,564.80	\$64,833.60	\$66,164.80	\$67,537.60
	Hrly	\$30.26	\$30.84		\$32.05	\$32!70	\$33.35	\$34.04	\$34.74
	37.5/wk	\$1,134.75	\$1,156.50		\$1,201.88		\$1,250.63		\$1,302.75
11	52/yr	\$59,007.00	\$60,138.00		\$62,497.76	\$63,765.00	\$65,032.76	\$66,378.00	\$67,743.00
1	40/wk	\$1,210.40	\$1,233.60		\$1,282.00		\$1,334.00		\$1,389.60
	52/yr	\$62,940.80	\$64,147.20	\$65,395.20	\$66,664.00	\$68,016.00	\$69,368.00	\$70,803.20	\$72,259.20
·	Hrly	\$31.62			\$33.49		\$34.85	\$35.57	\$36.30
	37.5/wk	\$1,185.75	\$1,208.63	\$1,231,88	\$1,255.88		\$1,306.88		\$1,361.25
12	52/уг	\$61,659.00	\$62,848.76		\$65,305.76		\$67,957.76		\$70,785.00
ļ	40/wk	\$1,264.80	\$1,289.20		\$1,339.60		\$1,394.00		\$1,452.00
	52/yr	\$65,769.60	\$67,038.40		\$69,659.20		\$72,488.00	\$73,985.60	\$75,504.00
	Hrly	\$34.15	\$34.81	\$35.48	\$36:17		\$37.64	\$38,42	\$39.20
10	37.5/wk	\$1,280.63	\$1,305.38		\$1,356.38	\$1,383.75	\$1,411.50		\$1,470.00
13	52/yr	\$66,592.76	\$67,879.76		\$70,531.76		\$73,398.00		\$76,440.00
İ	40/wk	\$1,366.00	\$1,392.40		\$1,446.80		\$1,505.60		\$1,568.00
	52/yr	\$71,032.00	\$72,404.80		\$75,233.60		\$78,291.20		\$81,536.00
	Hrly	\$34.49		\$35.83	\$3653			\$38.80	\$39.59
4.4	37.5/wk	\$1,293.38	\$1,318.50	\$1,343.63	\$1,369.88	\$1,397.63	\$1,425.75	\$1,455.00	\$1,484.63
14	52/yr	\$67,255.76	\$68,562.00		\$71,233,76		\$74,139.00		\$77,200.76
l	40/wk	\$1,379.60	\$1,406.40		\$1,461.20		\$1,520.80		\$1,583.60
	52/yr	\$71,739.20	\$73,132.80		\$75,982.40	\$77,521.60	\$79,081.60		\$82,347.20
	Hily	\$35.01	\$35.69	\$3637	\$37.08	\$37.83	\$38.59	\$39.38	\$40.18
1 1 1	37.5/wk	\$1,312.88	\$1,338.38		\$1,390.50	\$1,418.63	\$1,447.13	\$1,476.75	\$1,506.75
15	52/yr	\$68,269.76	\$69,595.76		\$72,306.00	\$73,768.76	\$75,250.76		\$78,351.00
	40/wk	\$1,400.40	\$1,427.60		\$1,483.20		\$1,543.60		\$1,607.20
	52/уг	\$72,820.80	\$74,235,20	\$75,649.60	\$77,126.40	\$78,686.40	\$80,267.20	\$81,910.40	\$83,574.40

6--2016 Special Town Meeting October 22, 2016

	Hrlv	\$36.17	\$36.87	\$37.57	\$38.30	\$39.08	\$39.86	\$40.68	\$41.51
	37.5/wk	\$1,356.38	\$1,382.63	\$1,408.88	\$1,436.25	\$1,465.50	\$1,494.75	\$1,525.50	\$1,556.63
16	52/yr	\$70,531.76	\$71,896.76	\$73,261.76	\$74,685.00	\$76,206.00	\$77,727.00	\$79,326.00	\$80,944.76
1 10	40/wk	\$1,446.80	\$1,474,80	\$1,502.80	\$1,532.00	\$1,563.20	\$1,594.40	\$1,627.20	\$1,660.40
	52/yr	\$75,233.60	\$76,689.60	\$78,145.60	\$79,664.00	\$81,286.40	\$82,908.80	\$84,614.40	\$86,340.80
	Hrly	\$37.26	\$37.98	\$38.70	\$39,45	\$40.25	\$41.06	\$41.90	\$42.76
	37.5/wk	\$1,397.25	\$1,424.25	\$1,451.25	\$1,479.38	\$1,509.38	\$1,539.75	\$1,571.25	\$1,603.50
17	52/уг	\$72,657.00	\$74,061.00	\$75,465.00	\$76,927.76	\$78,487.76	\$80,067.00	\$81,705.00	\$83,382.00
-,	40/wk	\$1,490.40	\$1,519.20	\$1,548.00	\$1,578.00	\$1,610.00	\$1,642.40	\$1,676.00	\$1,710.40
	52/yr	\$77,500.80	\$78,998.40	\$80,496.00	\$82,056.00	\$83,720.00	\$85,404.80	\$87,152.00	\$88,940.80
The same of the sa	Hrly	\$39.68	\$40.45	\$41.22	\$42.01	\$42.87	\$43.73	\$44.62	\$45.54
	37.5/wk	\$1,488.00	\$1,516.88	\$1,545.75	\$1,575.38	\$1,607.63	\$1,639.88	\$1,673.25	\$1,707.75
18	52/yr	\$77,376.00	\$78,877.76	\$80,379.00	\$81,919.76	\$83,596.76	\$85,273.76	\$87,009.00	\$88,803.00
~	40/wk	\$1,587.20	\$1,618.00	\$1,648.80	\$1,680.40	\$1,714.80	\$1,749.20	\$1,784.80	\$1,821.60
	52/yr	\$82,534.40	\$84,136.00	\$85,737.60	\$87,380.80	\$89,169.60	\$90,958.40	\$92,809.60	\$94,723.20
	Hrly	\$42.26	\$43.08	\$43.90	\$44.74	\$45.66	\$46.57	\$47.52	\$48.50
	37.5/wk	\$1,584.75	\$1,615.50	\$1,646.25	\$1,677.75	\$1,712.25	\$1,746.38	\$1,782.00	\$1,818.75
19	52/yr	\$82,407.00	\$84,006.00	\$85,605.00	\$87,243.00	\$89,037.00	\$90,811.76	\$92,664.00	\$94,575.00
	40/wk	\$1,690.40	\$1,723.20	\$1,756.00	\$1,789.60	\$1,826.40	\$1,862.80	\$1,900.80	\$1,940.00
	52/yr	\$87,900.80	\$89,606.40	\$91,312.00	\$93,059.20	\$94,972.80	\$96,865.60	\$98,841.60	\$100,880.00
	Hrly	\$44.58	\$45.45	\$46.31	\$47.20	\$48.17	\$49.13	\$50.13	\$51.17
	37.5/wk	\$1,671.75	\$1,704.38	\$1,736.63	\$1,770.00	\$1,806.38	\$1,842.38	\$1,879.88	\$1,918.88
20	52/yr	\$86,931.00	\$88,627.76	\$90,304.76	\$92,040.00	\$93,931.76	\$95,803.76	\$97,753.76	\$99,781.76
-	40/wk	\$1,783.20	\$1,818.00	\$1,852.40	\$1,888.00	\$1,926.80	\$1,965.20	\$2,005.20	\$2,046.80
	52/yr	\$92,726.40	\$94,536.00	\$96,324.80	\$98,176.00	\$100,193.60	\$102,190.40	\$104,270.40	\$106,433.60
	Hrly	\$46.36	\$47.27	\$48.16	\$49.09	\$50.10	\$51.10	\$52:14	\$53.22
	37.5/wk	\$1,738.50	\$1,772.63	\$1,806.00	\$1,840.88	\$1,878.75	\$1,916.25	\$1,955.25	\$1,995.75
21	52/yr	\$90,402.00	\$92,176.76	\$93,912.00	\$95,725.76	\$97,695.00	\$99,645.00	\$101,673.00	\$103,779.00
	40/wk	\$1,854.40	\$1,890.80	\$1,926.40	\$1,963.60			\$2,085.60	\$2,128.80
1	52/yr	\$96,428.80	\$98,321.60	\$100,172.80	\$102,107.20	\$104,208.00	\$106,288.00	\$108,451.20	\$110,697.60
	Hrly	\$50.53	\$51.52	\$52.49	\$53.51	\$54.61	\$55.70	\$56.83	\$58.01
	37.5/wk	\$1,894.88	\$1,932.00	\$1,968.38	\$2,006.63	\$2,047.88	\$2,088.75	\$2,131.13	\$2,175.38
22	52/yr	\$98,533.76	\$100,464.00	\$102,355.76	\$104,344.76	\$106,489.76	\$108,615.00	\$110,818.76	\$113,119.76
	40/wk	\$2,021.20	\$2,060.80	\$2,099.60	\$2,140.40	\$2,184.40	\$2,228.00	\$2,273.20	\$2,320.40
	52/yr	\$105,102.40	\$107,161.60	\$109,179.20	\$111,300.80	\$113,588.80	\$115,856.00	\$118,206.40	\$120,660.80

APPENDIX B

COMMUNITY PRESERVATION COMMITTEE BUDGET

1) Historic Preservation - Patton Homestead Restoration Project

To appropriate \$45,000 from the Community Preservation Fund Balance (Unreserved) to fund the repair of the roof and dormers of the Patton Homestead, conditioned upon commencement of construction by June 30, 2017.

The Community Preservation Committee voted five members in favor and one member in opposition of the appropriation.

2) Historic Preservation - Town Hall Preservation Project

To appropriate \$250,000 (\$185,000 from the Community Preservation Fund Balance (Unreserved) and \$65,000 from the Historic Preservation Reserves) to fund the historic preservation analysis and evaluation of existing condition of Town Hall, conditioned upon completion of the analysis by June 30, 2019. The Community Preservation Committee voted six members in favor of the appropriation.



TOWN OF HAMILTON ZONING BY-LAW

FIRST ADOPTED 1954 - INCLUDING AMENDMENTS UP TO OCTOBER 22, 2016

ADOPTED PURSUANT TO THE MASSACHUSETTS ZONING ACT M.G.L. CHAPTER 40A AS AMENDED

HAMILTON ZONING BY-LAW

TABLE OF CONTENTS

			<u>Page</u>			
SECT	ΓΙΟΝ 1.0	PURPOSE AND AUTHORITY	1			
1.1	PURPO	OSE	. 1			
1.2	AUTH	ORITY	1			
1.3	SCOPE	3	1			
1.4	APPLI	CABILITY	1			
	1.4.1 1.4.2	Applicability/Nonconformities Commencement of Construction or Operation	1 2			
1.5	AMEN	IDMENTS	2			
1.6	SEPAR	RABILITY	2			
1.7	DEFIN	IITIONS	2			
SECT	TION 2.0	ESTABLISHMENT OF DISTRICTS	2			
2.1	CLASS	SES OF DISTRICTS	2			
2.2	ZONIN	NG MAP	2			
2.3	OVER	LAY DISTRICTS (See also Section 9.0)	3			
2.4	BOUN	BOUNDARIES OF DISTRICTS				
	2.4.1 2.4.2 2.4.3 2.4.4 2.4.5	Center Lines Lot Lines Parallel Other Dispute	3 3 3 3 3			
2.5	SPLIT	LOTS	3			
	2.5.1 2.5.2	By Town Boundary By District Boundary	3			
SECT	TION 3.0	USE REGULATIONS	4			
3.1	PRINC	PRINCIPAL USES				
	3.1.1 3.1.2 3.1.3 3.1.4	Applicability of Use Regulations Table of Use Regulations Key Uses Subject to Other Regulations	4 4 4 7			

3.2	ACCI	ESSORY USES	7
	3.2.1	Accessory Uses in All Districts	7
	3.2.2	Accessory Uses and Structures in the Residence Districts	8
	3.2.3	Accessory Uses in the Business District	8
3.3	НОМ	E OCCUPATIONS	8
	3.3.1	General	. 8
	3.3.2	Conditions	8
3.4	CON	VERSION FOR TEMPORARY ADDITIONAL LIVING AREA	9
	3.4.1	Purpose	9
	3.4.2	Procedures	9
	3.4.3	Renewal	9
	3.4.4	Design and Dimensional Standards	10
	3.4.5	Other Approvals	10
	3.4.6	Special Restrictions	11
3.5	CON	VERSION TO TWO-FAMILY DWELLING	11
	3.5.1	Purpose	11
	3.5.2	Procedures	11
	3.5.3	Design and Dimensional Standards	11
	3.5.4	Other Approvals	12
3.6	ACCE	SSORY APARTMENT	12
	3.6.1	Purpose	12
	3.6.2	Procedures	12
	3.6.3	Design and Dimensional Standards	12
	3.6.4	Rolling Limit	13
	3.6.5	Other Approvals	13
	3.6.6	Special Restrictions	13
3.7	ACCE	SSORY APARTMENTS ON LARGE LOTS	13
	3.7.1	Purpose	13
	3.7.2	Procedures	13
	3.7.3	Design and Dimensional Standards	14
	3.7.4	Other Approvals	14
	3.7.5	Special Restrictions	14
SECTI	ON 4.0	DIMENSIONAL AND DENSITY REGULATIONS	14
4.1	GENE	RAL REGULATIONS	14
	4.1.1	Applicability	14
	4.1.2	Table of Dimensional Regulations	15
	4.1.3	Computation of Lot Area	15
	4.1.4	Irregular Lots	15
	4.1.5	Accessory Buildings	16

4.2	SPECL	SPECIAL REGULATIONS				
	4.2.1	Exemptions from Height Limits	16			
	4.2.2	Required Circle in Residential Districts	16			
	4.2.3	Frontage Exception for Larger Lots	16			
	4.2.4	Averaging of Setbacks	17			
	4.2.5	Double Frontage	17			
	4.2.6	Width	17			
	4.2.7	Minimum Lot Dimensions When Creating a New Street	17			
	4.2.7	Watercourse Yards	17			
		·	17			
	4.2.9	Appurtenant Open Space	17			
		Corner Clearance	17			
	4.2.11	Wind Turbine Setbacks (See also Section 7.4)	17			
4.3	CONTI	IGUOUS AREA OF LAND REQUIREMENT				
	4.3.1	General	17			
	4.3.2	Interpretation	18			
	4.3.3	Minimums	18			
	4.3.4	Proof	18			
		Location of House	18			
	4.3.6	Buildable Area	18			
SECT	TON 5.0	NONCONFORMING USES AND STRUCTURES	19			
5.1	APPLIC	CABILITY	19			
5.2	NONC	ONFORMING USES	19			
	5.2.1	Eligible Changes	19			
5.3	NONC	ONFORMING STRUCTURES	19			
	5.3.1	Eligible Changes	19			
	5.3.2	Variance Required	19			
	5.3.3	Nonconforming Single and Two Family Residential Structures	19			
	5.3.4	Eligible Changes	20			
5.4	ABAN	DONMENT OR NON-USE	20			
5.5	RECO	NSTRUCTION AFTER CATASTROPHE OR DEMOLITION	20			
	5.5.1	Procedures	20			
5.6	REVER	RSION TO CONFORMITY OR MORE RESTRICTIVE USE	21			
5.7	SUBST	CANDARD LOTS	21			
SECT	TON 6.0	GENERAL REGULATIONS	21			
6.1	OFF-S	TREET PARKING AND LOADING AREAS	21			
	6,1.1	Number of Required Parking Spaces	21			
	6.1.2	Location	21			
	6.1.3		22			
	614		22			

	6.1.5	Access	22
	6.1.6	Shared Parking	22
	6.1.7	Surface	22
	6.1.8	Lights	22
	6.1.9	Site Plan	22
	6.1.10	Special Permit	22
6.2	DRIVE	EWAYS	22
	6.2.1	Driveway Plan	22
	6.2.2	Standards	22
	6.2.3	Hydrant	23
	6.2.4	Connection to State Highway	23
	6.2.5	Connection with Access Street	23
	6.2.6	Common Driveway	23
6.3	SIGNS	}	24
	6.3.1	Permit Required	24
	6.3.2	Residence Districts	24
	6.3.3	Business District	24
	6.3.4	Standards	24
6.4	STANI	DARDS FOR NONRESIDENTIAL	
	BUILD	DINGS' CONSTRUCTION, USE AND OPERATION	25
	6.4.1.	General	25
SEC'	TION 7.0	SPECIAL REGULATIONS	25
7.1	ADUL	T ENTERTAINMENT USES	25
	7.1.1	Purpose	25
	7.1.2	Special Permit	26
	7.1.3	Location	26
	7.1.4	Standards	26
	7.1.5	Site Plan	26
	7.1.6	Imposition of Other Conditions	26
	7.1.7	Lapse of Permit	28
7.2	COMM TELEC	IUNICATION TOWERS (CT) AND COMMUNICATION ANTENNA FACILITIES	28
	7.2.1	Purpose	20
	7.2.1	Permitting	28
	7.2.3	Application	29
	7.2.3	Required Reports	29
	7.2.4	On Site Demonstration	29
	7.2.5	Requirements	30
	7.2.7		31
	7.2.7	Clearing	32
	7.2.8 7.2.9	Design Requirements	32
	7.2.9 7.2.10	Approval	33
		Construction Deadline	33
	7.2.11	Conditions of Completion and Use	33
	7.2.12 7.2.13	Lapse Abandonment	33
	//14	Abandonment	34

	7.2.14 7.2.15	Performance Guarantee Fees	34 34
	7.2.16	Waivers	34
	7.2.17		35
	7.2.18	Other Requirements	35
7.3	WIND	ENERGY FACILITIES	35
	7.3.1	Purpose	35
	7.3.2	Applicability	35
	7.3.3	Compliance with Laws and Regulations	35
	7.3.4	Building Permit	35
	7.3.5	Site Plan Review Required in the Commercial Overlay District	36
	7.3.6	Special Permit Required Outside the Commercial Overlay District	36
	7.3.7	Utility Scale Wind Energy Facilities Submittal Requirements	36
	7.3.8	Utility Scale Wind Turbine Standards	38
	7.3.9	Decommissioning	41
	7.3.10	Small Scale Wind Energy Facilities Submittal Requirements	41
	7.3.11	Small Scale Wind Turbine Standards	42
	7.3.12	Compliance with Other Regulations	43
	7.3.13	Utility Notification	43
	7.3.14	Special Permit Approval Criteria	43 44
	7.3.15	As-Built Plan	44
	7.3.16	Abandonment	44
	7.3.17	Utility Connections Modification	44
	7.3.18 7.3.19	Priority	44
	7.3.19	Thomy	
SECT	TION 8.0	SPECIAL RESIDENTIAL REGULATIONS	45
8. 1		SPACE AND FARMLAND PRESERVATION	45
	DEVE	LOPMENT	45
	8.1.1	Purpose	45
	8.12	Applicability	45
	8.1.3	Significance of Grant of Special Permit Future Subdivision	46
	8.1.4	Planning Board Regulations	46
	8.1.5 8.1.6	Permitted Uses	46
	8.1.7	Pre-Application Conference	46
	8.1.8	OSFPD Special Permit Application Procedures	47
	8.1.9	Sources of Data	47
	8.1.10	Allowed OSFPD Density	47
	8.1.11	OSFPD Concept Plan	47
	8.1.12	OSFPD Special Permit Design Process	48
	8.1.13	Common Open Space Requirement	49
	8.1.14	Ownership of the Common Open Space	50
	8.1.15	Dimensional Standards	50
	8.1.16	OSFPD Special Permit General Design Standards	50
	8.1.17	OSFPD Special Permit Site Specific Design Standards	51
	8.1.18	Review Procedures for OSFPD Special Permit	52
	8.1.19	Decision of the Planning Board for OSFPD Special Permit	53
	8.1.20	Effect of OSFPD Special Permit Approval	53
	8.1.21	Public Benefit Incentives	53
	8.1.22	Relationship Between the OSFPD Special Permit and OSFPD Definitive Subdivision Plan	54

	8.1.23		
		Cluster Design Plan Approval	54
	8.1.24	Cluster Design Plan Submittal Requirements	54
	8.1.25	Cluster Design Plan Procedures	55
	8.1.26	Cluster Design Plan Decision	55
8.2	SENIC	OR HOUSING	57
	8.2.1	Purpose	57
	8.2.2	Eligible Locations	57
	8.2.3	Administration	57
	8.2.4	Planning Board Rules and Regulations	57
	8.2.5	Age Restriction	57
	8.2.6	Limitations on Senior Housing Development	58
	8.2.7	Permitted Uses	58
	8.2.8	Developable Acres	59
	8.2.9	Base Senior Housing Density	59
	8.2.10	Incentives to Increase Base Senior Housing Density	59
	8.2.11	Minimum Distances Between Projects	61
	8.2.12	General Compliance	62
	8.2.13	General Requirements	63
	8.2.14	Exemptions	64
	8.2.15	Dimensional Standards	64
	8.2.16	Building and Design Standards	64
	8.2.17	Lighting	65
	8.2.18	Signs	65
	8.2.19	Site Access, Roadway Design and Management	65
	8.2.20	Stormwater Management and Erosion Control	65
	8.2.21	Parking	66
	8.2.22	Setbacks and Landscape Buffers	66
	8.2.23	Open Space and Natural Resources	66
	8.2.24	Use, Shape, Location of Common Open Space	67
	8.2.25	Senior Housing Special Permit Application Process	68
	8.2.26	Definitive Subdivision Plan	68
	8.2.27	Application Process	68
	8.2.28	Prohibition	68
	8.2.29	Conditions of Approval	68
	8.2.30	Employment of Outside Consultants	69
	8.2.31	Planning Board Findings	69
	8.2.32	Expansion	69
	8.2.33	Annual Reporting	69
8.3	INCLU	SIONARY HOUSING	70
	8.3.1	Purpose	70
	8.3.2	Applicability	70
	8.3.3	Mandatory Provision of Affordable Housing Units	70
	8.3.4	Methods of Providing Affordable Housing Units	70
	8.3.5	General Provisions	71
	8.3.6	Timing of Construction	72
	8.3.7	Certificate of Occupancy	72
	8.3.8	Preservation of Affordability; Restrictions on Resale	72

SECT	TION 9.0	SPECIAL DISTRICT REGULATIONS	72
9.1	GROUI	NDWATER PROTECTION OVERLAY DISTRICT	72
	9.1.1	Purpose	72
	9.1.2	Overlay District	73
	9.1.3	Location	73
	9.1.4	Dimensional Requirements	73
	9.1.5	Lot Partially in GPOD	73
	9.1.6	Boundary in Doubt	73
	9.1.7	Permitted Uses	73 74
	9.1.8	Prohibited Uses	75
	9.1.9	Uses and Activities Requiring a Special Permit	76
	9.1.10	Special Permit Application	76 76
	9.1.11	Procedures for Issuance of Special Permits	77
	9.1.12 9.1.13	Special Permit Criteria Enforcement	77
9.2	FLOOD	D PLAIN OVERLAY DISTRICT (FPOD)	77
, 			77
	9.2.1	Purpose	78
	9.2.2	FPOD Boundaries	78
	9.2.3	Base Flood Elevation and Floodway Data	78
	9.2.4	Use Regulations Other Requirements	79
	9.2.5	Other Requirements Permitted Uses	79
	9.2.6 9.2.7	Notification of Watercourse Alteration	80
	9.2.1	Notification of watercourse Afteration	
9.3	CONSE	ERVANCY DISTRICT	80
	9.3.1	Purpose	80
	9.3.2	Permitted Uses	81
	9.3.3	Special Permit	81
	9.3.4	Prohibited Uses	82
	9.3.5	Reversion	83
9.4	ESTAT	TE OVERLAY DISTRICT (EOD)	83
	9.4.1	Purpose	83
	9.4.2	Overview	83
	9.4.3	Applicability	83
	9.4.4	Access Requirements for the EOD	84 84
	9.4.5	Open Space Requirement	84
	9.4.6	Open Space Access	85
	9.4.7	Review Criteria	85
	9.4.8	Advisory Opinions	85
	9.4.9	Phasing Pro Application Conference	85
	9.4.10	Pre-Application Conference EOD Special Permit Design Process	85
	9.4.11 9.4.12	Certification	86
	9.4.12	Special Permit Application	86
	9.4.13	Criteria	87
	9,4.14	Modifications	87
	~, 1112		

	EOD	Part A – Estate House Adaptive Reuse	0.0
	9.4.16	S Part A	88
	9.4.17		88
		Part A – Dimensional Requirements	88
	9.4.19	Part A – Special Permit Application	88
	9.4.20	1	88
		Part B – New Floor Area Construction	88
	9.4.21		89
		Part B – Permitted Uses	89
		Part B – Methods of Achieving New Floor Area	89
	9.4.24	Part B – Dimensional Requirements	90
9.5	WILL	OW STREET OVERLAY DISTRICT (WSOD)	91
	9.5.1	Purpose	91
	9.5.2	Location	92
	9.5.3	Applicability	92
	9.5.4	Permitted Uses	92
	9.5.5	Administration	92
	9.5.6	Planning Board Rules and Regulations	92
	9.5.7	Pre-Application Conference	93
	9.5.8	Review Procedures for WSOD Special Permit	93
	9.5.9	General Requirements	93
	9.5.10	Q = 11-18-1 2 VIII III	94
	9.5.11	Criteria	96
	9.5.12		96
	9.5.13	Lapse	96
9.6	COMN	MERCIAL OVERLAY DISTRICT (COD)	96
	9.6.1	Purpose	96
	9.6.2	Overlay District	97
	9.6.3	Permitted Uses By Right in the COD	97
	9.6.4	Uses Permitted by Special Permit in the COD	97
	9.6.5	Administration	98
	9.6.6	Planning Board Rules and Regulations	98
	9.6.7	Pre-Application Conference	98
	9.6.8	Review Procedures for COD Special Permit	98
	9.6.9	Prohibition	99
	9.6.10	General Compliance	99
	9.6.11	General Requirements	99
	9.6.12	Building and Design Standards	100
	9.6.13	Signs	102
	9.6.14	Conditions of Approval	102
	9.6.15	Findings and Decision of the Planning Board for COD	102
		Special Permit	102
,	9.6.16	Lapse	103
SECT	ION 10.0	ADMINISTRATION AND ENFORCEMENT	103
10.1	ADMIN	NISTRATION	103
	10.1.1	Building Commissioner	103
	10.1.2	Building Permit Required	103
	10.1.3	Application for Permit	103
		••	103

	10.1.4 10.1.5	Approval of Applications Occupancy Permit	103 104
10.2	ENFOR	CEMENT AND PENALTIES	104
	10.2.1	Enforcement	104
	10.2.2	Appeal	104
	10.2.3	Violations and Penalty	104
10.3	ZONIN	G BOARD OF APPEALS	104
	10.3.1	Composition of Board	104
	10.3.2	Duties	104
	10.3.3	Powers	105
		Regulations; Fees	105
	10.3.5	Public Notice	105
10.4	PLANN	ING BOARD	105
	10.4.1	Establishment	105
	10.4.2	Powers	105
	10.4.3	Rules and Regulations; Fees	106
	10.4.4	Associate Members of the Planning Board	106
	10.4.5	Public Notice	106
10.5	SPECIA	AL PERMITS	106
	10.5.1	Special Permit Granting Authority	106
	10.5.2	Criteria	106
	10.5.3	Application	107
	10.5.4	Conditions	107
	10.5.5	Regulations	107
	10.5.6	Fees	107
	10.5.7	Lapse	107
10.6	SITE PI	LAN REVIEW	107
	10.6.1	Purpose	107
	10.6.2	Projects Requiring Approval	107
	10.6.3	Projects Requiring Abbreviated Site Plan Review	108
	10.6.4	Exemptions from Site Plan Review	108
	10.6.5	Site Plan Review; Procedures	109
	10.6.6	Site Plan Review; Hearing and Decision	109
	10.6.7	Abbreviated Site Plan Review; Procedures	110
	10.6.8	Abbreviated Site Plan Review; Hearing and Decision	110
	10.6.9	Site Plan Review; Contents	110
	10.6.10	Site Plan Review; Other Materials	111
		Abbreviated Site Plan Review; Plans and Submission Contents	112
		Site Plan Standards	112
		Final Action	112
		Implementation of Site Plans	113
	10.6.15	Appeal	113

10.5	arms n		
10.7		LAN REVIEW FOR RELIGIOUS AND	
	EDUC	ATIONAL USES AND CERTAIN CHILD CARE CENTERS	113
	10.7.1	Purpose	113
	10.7.2	Site Plan Review Required	113
	10.7.3	Scope of Site Plan Review	113
	10.7.4	Required Information	113
	10.7.5	Site Plan; Contents	114
	10.7.6	Decision	115
	10.7.7	Appeal	115
10.8	REQUE	EST FOR REASONABLE ACCOMMODATION	115
	10.8.1	Purpose	115
	10.8.2	Request	115
	10.8.3	Zoning Board of Appeals	116
	10.8.4	Information	116
	10.8.5	Zoning Board of Appeals Procedures	116
	10.8.6	Decision	117
	10.8.7	Appeal	117
	10.8.8	File	117
	10.8.9	Other Laws	117
	10.8.10	Effective Date	117
SECTIO	N 11.0	DEFINITIONS	117

HAMILTON ZONING BY-LAW

SECTION 1.0 PURPOSE AND AUTHORITY

- 1.1 PURPOSE. The purpose of the Hamilton Zoning By-law ("this By-law") is:
 - 1. To promote the health, safety, morals, convenience and general welfare of the inhabitants of Hamilton;
 - 2. To lessen the danger from fire and congestion, and from the hazards of flood water inundation, to protect and conserve the value of property;
 - 3. To preserve and increase the amenities of the Town;
 - 4. To conserve natural conditions, to promote the educational, cultural and economic welfare of the public through the preservation and protection of Buildings, sites, and districts of historic interest; and
 - 5. To improve and beautify the Town by encouraging the most appropriate uses of land within the Town in accordance with the General or Master Plan, and under the provisions of Chapter 40A of the Massachusetts General Laws, as amended, and Section 2A of 1975 Mass. Acts 808.
- **1.2 AUTHORITY.** This By-law is enacted in accordance with the provisions of the General Laws, Chapter 40A, and any and all amendments thereto, and by Article 89 of the Amendments to the Constitution of the Commonwealth of Massachusetts.
- 1.3 SCOPE. For these purposes, the construction, repair, alteration, reconstruction, height, number of stories, and size of Buildings and Structures, the size and width of Lots, the percentage of Lot area that may be occupied, the size of Yards, the density of population, and the location and use of Buildings, Structures, and land in the Town are regulated as hereinafter provided.
- 1.4 APPLICABILITY. All Buildings or Structures hereinafter erected, reconstructed, altered, enlarged, or moved, and the use of all premises in the Town, shall be in conformity with the provisions of the By-law. No Building, Structure or land shall be used for any purpose or in any manner other than is expressly permitted within the district in which such Building, Structure or land is located. Where the application of this By-law imposes greater restrictions than those imposed by any other regulations, permits, restrictions, easements, covenants, or agreements, the provisions of this By-law shall control.
- 1.4.1 Applicability/Nonconformities. Except as herein after provided, this By-law shall not apply to Buildings, Structures or uses lawfully in existence or lawfully begun, or to a building

permit or special permit issued before the first publication of notice of the public hearing on this By-law or any amendments thereto, but shall apply to any change or substantial extension of such use, to a building permit or special permit issued after the first notice or said public hearing, to any reconstruction, extension or structural change of such Building or Structure and to any alteration of a Building or Structure begun after the first notice of said public hearing to provide for its use in a substantially different purpose or for the same purpose in a substantially different manner or to a substantially greater extent except where alteration, reconstruction, extension or a structural change to a Single Family Dwelling or Two Family Dwelling does not increase the nonconforming nature of said Structure.

- **1.4.2** Commencement of Construction or Operation. Construction or operations under a building permit or special permit shall conform to any subsequent amendments to this By-law, unless the use or construction is commenced within a period of not more than six months after the issuance of the permit and in any case involving construction, unless such construction is continued through to completion as continuously and expeditiously as is reasonable.
- **1.5 AMENDMENTS.** This By-law may from time to time be changed by amendment, addition, or repeal by the Town Meeting in the manner provided in G.L. c. 40A, s.5, and any amendments thereto.
- **1.6 SEPARABILITY.** The invalidity of any section or provision of this By-law shall not invalidate any other section or provision herein.
- 1.7 **DEFINITIONS.** The first letters of each word in defined terms in this By-law are capitalized. The definitions are found in Section 11 of this By-law. In some cases, definitions are grouped according to the Section in which the term appears.

SECTION 2.0 ESTABLISHMENT OF DISTRICTS

2.1 CLASSES OF DISTRICTS. The Town of Hamilton is hereby divided into the following districts:

Residence District	R-1a	
Residence District	R-1b	
Residence-Agricultural District	RA	
Business District	В	
Conservancy District	C	

2.2 ZONING MAP. The Zoning Map dated December 28, 1964 filed with the Town Clerk, as amended by the Zoning Map - Conservancy District, dated February 1, 1971, and further amended May 7, 1974, May 14, 1979, May 13, 1985, November 14, 1999 and May 5, 2009 ("Zoning Map") is hereby made a part of this By-law.

2.3 OVERLAY DISTRICTS. The following overlay districts are also established, as set forth in Section 9.0, herein.

Groundwater Protection Overlay District	GPOD
Flood Plain Overlay District	FPOD
Estate Overlay District	EOD
Willow Street Overlay District	WSOD
Commercial Overlay District	COD

2.4 BOUNDARIES OF DISTRICTS.

- **2.4.1 Center Lines.** Where the boundary lines are shown upon said Zoning Map within the Street lines and private ways, the center lines of such ways shall be the boundary lines.
- **2.4.2 Lot Lines.** Where the boundary lines are shown approximately on the location of property or Lot lines, and the exact location of property, Lot or boundary lines is not indicated by means of dimensions shown in figures, then the property or Lot lines existing at the time of the adoption of this By-law shall be the boundary lines.
- **2.4.3 Parallel.** Boundary lines located outside of such Street lines and shown approximately parallel thereto shall be regarded as parallel to such Street lines, and dimensions shown in figures placed upon said Zoning Map between such boundary lines and Street side lines are the distances in feet of such boundary lines from such Street side lines, such distances being measured at right angles to such Street lines unless otherwise indicated.
- **2.4.4 Other.** In all cases which are not covered by other provisions of this Section, the location of boundary lines shall be determined by the distance in feet, if given, from other lines upon said Zoning Map, by the use of identifications as shown on the Zoning Map, or by the scale of the Zoning Map.
- **2.4.5 Dispute.** Whenever any dispute arises on district boundaries as to the exact location of a district boundary line, the location of such line shall be determined by the Building Commissioner, after consultation with the Planning Board.

2.5 SPLIT LOTS.

- **2.5.1** By Town Boundary. When a Lot is situated in part in the Town and in part in an adjacent municipality, the provisions of this By-law shall be applied to the portion of such Lot in the Town in the same manner as if the entire Lot were situated in the Town.
- 2.5.2 By District Boundary. Where a district boundary line between a residential and a business district divides any Lot existing at the time such line is adopted, the regulations for the less restricted portions of such Lots shall extend no more than thirty (30) feet into the more restricted portion, provided the Lot has Lot Frontage on a public way in the less restricted district.

SECTION 3.0 USE REGULATIONS

3.1 PRINCIPAL USES.

- **3.1.1** Applicability of Use Regulations. Except as provided by law or in this By-law, in each district no Building, Structure, or land shall be used or occupied except for the purposes permitted as set forth in the accompanying Table of Use Regulations. Any use not listed shall be construed to be prohibited.
- **3.1.2 Table of Use Regulations.** The Table of Use Regulations is hereby declared to be part of this By-law.
- **3.1.3 Key.** In the following Table of Use Regulations, the uses permitted by right in the district are designated by the letter "Y." Uses designated by the letter "N" are not permitted in the district. Uses that may be permitted by special permit in the district, in accordance with Section 10.5, are designated by identification of the Special Permit Granting Authority ("SPGA") with authority to permit the use in that district, which is either:

ZBA Zoning Board of Appeals PB Planning Board

BOS Board of Selectmen

TABLE OF USE REGULATIONS

A. <u>RESIDENTIAL</u>	<u>R-1a</u>	<u>R-1b</u>	RA	<u>B</u>
1. One Single Family Dwelling	Y	Y	Y	Y
2. Conversion of a Single Family Dwelling existing at the time of the adoption of the ordinance (1954) into a Two Family Dwelling (see Section 3.5)	ZBA	ZBA	ZBA	ZBA
3. Open Space and Farmland Preservation Development (see Section 8.1)	PB	PB	PB	N
4. Senior Housing (see Section 8.2)	PB	PB	PB	PB
5. Long Term Care Facility	ZBA	ZBA	ZBA	N
6. Garage with more than 4 motor vehicle spaces	ZBA	ZBA	ZBA	ZBA
7. Two or more dwelling units, second floor and above when part of a mixed-use building or development	N	N	N	Y

B. COMMUNITY FACILITIES	<u>R-1a</u>	<u>R-1b</u>	RA	<u>B</u>
1. Use of land or Structures for religious purposes		Y	Y	Y
2. Museums, libraries and parks, playgrounds, conservation areas, water supply areas and other land owned and operated for the public enjoyment or service by a public or semi-public agency	Y	Y	Y	Y
3. Use of land or Structures for educational purposes on land owned or leased by the Commonwealth of Massachusetts or any of its agencies, subdivisions or bodies politic or by a religious sect or denomination, or by a nonprofit educational corporation	Y	Y	Y	Y
4. Hospital, Medical Clinic, cemetery, and camps of educational and charitable institutions	ZBA	ZBA	ZBA	ZBA
5. Community or private club, not conducted for profit	ZBA	ZBA	ZBA	ZBA
6. Nonprofit civic or fraternal building	N	N	N	Y
7. Child Care Center or School Aged Child Care Program	Y	Y	Y	Y
8. Commercial Recreation, Outdoors	BOS	BOS	BOS	BOS
9. Temporary use for amusements and recreation	BOS	BOS	BOS	BOS
10. Municipal Buildings or facilities	ZBA	ZBA	ZBA	Y
11. Essential Services	N	N	N	Y
C. AGRICULTURAL	R-1a	R-1b	RA	В
1. On parcels five (5) acres or more, or two (2) Qualified Acres or more: agriculture, horticulture and floriculture	Y	Y	Y	Y
2. Gardens; riding stables; growing and storing of fruits, berries, vegetables, hay, fodder and ensilage; woodlots; forestry; and greenhouses	Y	Y	Y	Y
3. The raising or keeping of poultry, horses, or cows for other than the use of the occupants of the residence	ZBA	ZBA	ZBA	ZBA

D. COMMERCIAL	R-1a	R-1b	RA	В
1. Adult Entertainment Uses	N	N	N	ZBA
2. Motor Vehicle and/or Boat Sales and/or Leasing Shop	N	N	N	ZBA
3. Business or Professional Office; bank; financial institution	N	N	N.	Y
4. Communication Towers and Telecommunication Antenna Facilities (see Section 7.2)	PB	PB	PB	PB
5. Kennel	ZBA	ZBA	ZBA	ZBA
6. Funeral homes	N	N	N	ZBA
7. Garage with more than four automobile/truck spaces	ZBA	ZBA	ZBA	ZBA
8. Motor Vehicle Light Service Station	N	N	N	Y
9. Manufacturing of products sold on the premises at retail, where no more than five operators are employed in such manufacture.	N	N	N	ZBA
10. Rail or bus station or terminal	N	N	N	ZBA
11. Motor Vehicle Repair Shop	N	N	N	ZBA
12. General Service Establishment	N	N	N	Y
13. Personal Service Establishment	N	N	N	Y
14. Restaurant	N	N	N	Y
15. Restaurant, Fast Food	N	N	N	ZBA
16. Retail Store	N	N	N	Y
17. Veterinary Facility or Clinic	N	N	N	ZBA
18. Wind Energy Facility in COD, subject to site plan approval (see Section 7.3)	Y	Y	Y	Y
19. Wind Energy Facility (not in COD) (see Section 7.3)	ZBA	ZBA	ZBA	ZBA
20. Communications Tower and Telecommunication Antenna Facilities (see Section 7.2)	Y	Y	Y	Y
21. Mixed Use, subject to site plan approval	N	N	N	Y
22. Drive-In or Drive-Through Establishment	N	N	N	ZBA
23. For Profit Educational Facility	N	N	N	ZBA

E. ACCESSORY	R-1a	R-1b	RA	В
1. Up to three boarders in a Single Family Dwelling	Y	Y	Y	N
2. Conversion for Temporary Living (see Section 3.4)	ZBA	ZBA	ZBA	ZBA
3. Accessory Apartment (see Section 3.6)	ZBA	ZBA	ZBA	ZBA
4. Accessory Apartment on Large Lot (see Section 3.7)	ZBA	ZBA	ZBA	ZBA
5. Customary home occupation conducted in a Dwelling or Building accessory thereto by a person residing on the premises (see Section 3.3)	Y	Y	Y	Y
6. Accessory Uses or Building on the same Lot with and customarily incidental to a permitted main use on the same premises	Y	Y	Y	Y
7. Uses, whether or not on the same parcel as activities permitted as a matter of right, which are both: (a) accessory to activities permitted as a matter of right, and (b) necessary in connection with scientific research or scientific development or related production (see Section 3.2.1.2)	ZBA	ZBA	ZBA	ZBA
8. Parking area or garage for use of employees, customers or visitors	N	N	N	Y
9. Level Two Electric Charging Station	Y	Y	Y	Y
10. Large Family Child Care Home	ZBA	ZBA	ZBA	ZBA
11. Small Family Child Care Home	Y	Y	Y	Y
12. Adult Social Day Care	ZBA	ZBA	ZBA	ZBA

3.1.4 Uses Subject to Other Regulations. Uses permitted by right or by special permit shall be subject, in addition to use regulations, to all other provisions of this By-law.

3.2 ACCESSORY USES.

- **3.2.1 Accessory Uses in All Districts.** The following Accessory Uses are specifically permitted in all districts as of right or by special permit:
 - 1. Temporary Construction Trailers and Signs. The Building Commissioner may grant a temporary occupancy permit for temporary buildings, Signs and trailers during building construction where reasonably required for such construction. Such permit may be issued for an initial period of not more than one year. Permits may be renewed by the Building Commissioner for successive periods of not more than one (1) year each.
 - 2. Accessory Scientific Uses. The accessory scientific uses set forth in Section E.7 of the Table of Use Regulation may be permitted upon the issuance of a special permit by the

Zoning Board of Appeals, provided that the Zoning Board of Appeals finds that the proposed use does not substantially derogate from the public good.

- 3.2.2 Accessory Uses and Structures in the Residence Districts. The following provisions shall apply to Accessory Uses and Structures in the Residence Districts:
 - 1. Accessory Uses on the same Lot with and customarily incidental to a permitted main use on the same premises are allowed, including but not limited to: agricultural activities, private swimming pools, tennis courts, and Private Garages.
 - 2. Activities associated with agriculture, horticulture and floriculture, such as barns, private stables, and shelters, and other farm buildings are allowed as Accessory Uses and/or Structures on Lots not less than eighty thousand (80,000) square feet in area, and including Buildings for keeping animals and poultry, duly licensed by the Board of Health, for occupant's use only, provided that no such barns, private stables, and shelters, and other farm buildings shall be nearer than fifty (50) feet to any Lot line.
 - 3. Tool sheds, garden sheds, storage sheds, or other like Buildings are allowed as Accessory Uses subject to the dimensional requirements in Section 4.0.
 - 4. The following uses are prohibited as Accessory Uses: Kennels; contractor's yard for the storage of building materials or equipment; the storage or keeping of commercial landscaping equipment, materials, supplies, or piles; and commercial auto repair or service.
- **3.2.3** Accessory Uses in the Business District. In the Business District, any use permitted as a principal use is also permitted as an Accessory Use provided such use is customarily incidental to the main or principal Building or use of the land. Any use authorized as a principal use by special permit may also be authorized as an Accessory Use by special permit provided such use is customarily incidental to the main or principal Building or use of the land.

3.3 HOME OCCUPATIONS.

3.3.1 General. A customary home occupation may be conducted in a Dwelling or Building accessory thereto by a person residing on the premises.

3.3.2 Conditions.

- 1. Such use must be clearly incidental and secondary to the use of the premises for residential purposes;
- 2. Not more than two (2) persons other than residents of the premises shall be regularly employed thereon in connection with such use;
- 3. No offensive noise, vibration, smoke, dust, fumes, odors, heat, glare or unsightliness shall be produced;

- 4. There shall be no public display of goods or wares and there shall be no signs except as permitted in Section 6.3; and
- 5. There shall be no exterior storage of material or equipment (including the parking of more than two commercial vehicles) and no other exterior indication of such use or variation from the residential character of the premises.

3.4 CONVERSION FOR TEMPORARY ADDITIONAL LIVING AREA.

- **3.4.1 Purpose.** The purpose of this Section is to provide a way for families to create separate living quarters in their home to accommodate a temporary family, health, or security need by permitting construction of one temporary additional living area in a Single Family Dwelling, or an Accessory Building, subject to the following.
- **3.4.2 Procedures.** A conversion for a temporary additional living area under this Section shall be authorized only by special permit issued by the Board of Appeals under Section 10.5 of this By-law. The Board of Appeals may grant such a special permit provided the following conditions are met:
 - 1. The Board of Health approves all arrangements for sanitary waste, water supply, and drainage.
 - 2. The occupancy permit for the principal Dwelling Unit must have been issued at least two (2) years prior to the application for the special permit.
 - 3. One (1) of the Dwelling Units shall be occupied by the owner of the property, except for bona fide temporary absence.
 - 4. The temporary additional living area must be for use by the owner(s); or a Family member of the owner(s); or a caretaker or a health care provider to the occupant of one (1) of the Dwelling Units; or an Elderly Person; or a mentally or physically handicapped person.
 - 5. There is no other accessory Dwelling Unit on the Lot on which the proposed accessory Dwelling Unit is to be located.
 - 6. The applicant has submitted to the Board of Appeals a plot plan and scaled architectural drawings which comply with Section 3.4.4 below.
- **3.4.3 Renewal.** Renewal of the special permit shall take place every four (4) years from the date of issuance of the special permit and upon change of ownership or tenants. Renewal of the special permit will require an inspection by the Building Commissioner and a written certification by the owner and Building Commissioner, verifying that the terms of the special permit and this By-law are being met. The Building Commissioner shall verify that all means of

access and egress have been maintained, and that there has been no change in the occupancy, design, or dimensional standards under this Section.

- **3.4.4 Design and Dimensional Standards.** A plot plan and scaled architectural drawings of the existing Dwelling Unit, the proposed accessory Dwelling Unit, and any proposed addition shall be submitted, showing location of the Building on the Lot, proposed addition, location of septic system and parking, and compliance with this Section and the following items:
 - 1. The maximum Residential Gross Floor Area of the proposed accessory Dwelling Unit shall not exceed the lesser of one thousand (1,000) square feet or one third (1/3) of the Residential Gross Floor Area of the existing Dwelling Unit and the proposed accessory Dwelling Unit combined, as measured after conversion;
 - 2. Any extension to the Building shall not create more than a twenty-five (25) percent increase in the Residential Gross Floor Area of the existing structure.
 - 3. The maximum number of bedrooms shall be one (1).
 - 4. The maximum number of tenants shall be two (2).
 - 5. The Lot on which the proposed accessory Dwelling Unit is to be located shall contain at least ten thousand (10,000) square feet.
 - 6. The external appearance of the Building in which the accessory Dwelling Unit is located shall not be significantly altered from a Single Family Dwelling Unit.
 - 7. The kitchen facilities shall be of a type readily removable.
 - 8. Adequate provision shall be made for egress to the outside from the accessory Dwelling Unit; any external stairways shall be screened from view, buffered, or located out of sight from any Street.
 - 9. One off Street parking space shall be provided for the new Dwelling Unit, in addition to the required parking for the principal Dwelling Unit. Every effort shall be made to minimize the visibility of the additional parking space by location and screening.
- **3.4.5 Other Approvals.** The applicant shall, in addition to obtaining a special permit, also obtain a building permit, any necessary Conservation Commission approvals, and an occupancy permit prior to the occupancy of the temporary additional living area.

3.4.6 Special Restrictions.

- 1. The occupancy of the apartment shall be by the apartment tenant(s) for whom it was issued. Upon cessation of occupancy by such tenant(s), the permit shall lapse and be null and void; the Building Commissioner shall be notified of this event and the kitchen facilities shall be removed by the owner.
- 2. One (1) of the Dwelling Units shall continue to be occupied by the owner of the property, except for bona fide temporary absence.

3.5 CONVERSION TO TWO-FAMILY DWELLING.

- **3.5.1 Purpose.** The purpose of this Section is to provide a way to preserve large older homes in the Town by allowing the conversion of a Single Family Dwelling existing at the time of the first adoption of this By-law (1954) into a Two Family Dwelling, subject to the following.
- **3.5.2 Procedures.** Conversion of such a Single Family Dwelling into a Two Family Dwelling under this Section shall be authorized only by special permit issued by the Board of Appeals under Section 10.5 of this By-law. The Board of Appeals may grant such a special permit provided the following conditions are met:
 - 1. The Board of Health approves all arrangements for sanitary waste, water supply, and drainage.
 - 2. The applicant has submitted to the Board of Appeals a plot plan and scaled architectural drawings which comply with Section 3.5.3 below
- **3.5.3 Design and Dimensional Standards.** A plot plan and scaled architectural drawings of the existing Dwelling Unit and alterations shall be submitted, showing location of the Building on the Lot, proposed alterations, location of septic system and parking, and compliance with the following items:
 - 1. The Lot on which the proposed conversion is to be located shall contain at least twenty thousand (20,000) square feet, and the existing Dwelling Unit shall contain at least four thousand (4,000) square feet;
 - 2. The external appearance of the Building in which the Dwelling Units are located shall not be significantly altered from its previous single family character;
 - 3. Adequate provision has been made for egress to the outside from the additional Dwelling Unit. Any external stairways shall be screened from view, buffered, or located out of sight from any Street; and
 - 4. One off street parking space shall be provided for the new Dwelling Unit, in addition to the required parking for the principal Dwelling Unit. Every effort shall be made to minimize the visibility of the additional parking space by location and screening.

3.5.4 Other Approvals. The applicant shall, in addition to obtaining a special permit, also obtain a building permit, and any necessary Conservation Commission approvals, and an occupancy permit prior to occupancy of the proposed Dwelling Unit.

3.6 ACCESSORY APARTMENT.

- **3.6.1 Purpose.** The purpose of this Section is to provide for a variety of housing choices in Hamilton, without substantially altering the appearance and character of residential neighborhoods or the Town by permitting accessory apartments in Single Family Dwellings without a minimum Lot area requirement, subject to the following.
- **3.6.2 Procedures.** Such an accessory apartment shall be authorized only by special permit issued by the Board of Appeals under Section 10.5 of this By-law. The Board of Appeals may grant such a special permit for an accessory apartment provided the following conditions are met:
 - 1. The Board of Health approves all arrangements for sanitary waste, water supply, and drainage.
 - 2. The applicant has submitted to the Board of Appeals a plot plan and scaled architectural drawings which comply with Section 3.6.3 below.
 - 3. The applicant is an owner-occupant of the premises.
 - 4. The Single Family Dwelling on the Lot is at least ten (10) years old at the time of the accessory apartment special permit application.
- **3.6.3 Design and Dimensional Standards.** A plot plan and scaled architectural drawings of the existing Dwelling Unit, proposed accessory apartment, and proposed addition (if any) shall be submitted, showing location of the Building on the Lot, location of septic system, and parking, and compliance with the following conditions:
 - 1. The accessory apartment shall be located within a Single Family Dwelling or in an Accessory Structure on the same Lot, such as an attached garage or a detached garage or barn, and shall clearly be a subordinate part of the Building.
 - 2. The accessory apartment shall not exceed nine hundred (900) square feet of Residential Gross Floor Area.
 - 3. Not more than one accessory apartment shall be permitted on a Lot.
 - 4. The maximum number of bedrooms shall be one.

- 5. There shall be at least one additional off-street parking space to serve the accessory apartment.
- 6. To the maximum extent practical, the accessory apartment shall be designed so that the exterior appearance of the Building remains unchanged.
- 7. Any Building addition for accessory apartment space shall not exceed fifteen (15) percent of the Residential Gross Floor Area existing in the Single Family Dwelling or accessory structure, as applicable, at the time of the application.
- 8. Unless otherwise required by the State Building Code, any new exterior stairs needed to provide primary or secondary means of egress for the accessory Dwelling shall not be located on a side of the Building facing the Street.
- **3.6.4 Rolling Limit.** The Board of Appeals shall not issue more than ten (10) accessory apartment special permits under this Section in any twelve (12) month period.
- **3.6.5 Other Approvals.** The applicant shall, in addition to obtaining a special permit, also obtain a building permit, and any necessary Conservation Commission approvals, and an occupancy permit prior to the occupancy of the proposed accessory apartment.

3.6.6 Special Restrictions.

- 1. The apartment shall not be transferred into separate ownership from the principal Single Family Dwelling, including a condominium form of ownership or otherwise.
- 2. The applicant shall continue to be the owner of the premises, and shall remain an occupant of either the principal Single Family Dwelling or the accessory apartment, except for bona fide temporary absences.

3.7 ACCESSORY APARTMENTS ON LARGE LOTS.

- **3.7.1 Purpose.** The purpose of this Section is to provide for accessory apartments in a Single Family Dwelling or in an accessory building on Lots greater than ten (10) acres, subject to the following.
- **3.7.2 Procedures.** Such an accessory apartment shall be authorized only by special permit issued by the Board of Appeals under Section 10.5 of this By-law. The Board of Appeals may grant such a special permit provided the following conditions are met:
 - 1. One of the Dwelling Units is occupied by the owner of the property, except for bona fide temporary absence.
 - 2. The principal Dwelling Unit and the proposed accessory apartment are to be held in the same ownership.

- 3. The Board of Health approves all arrangements for sanitary waste, water supply, and drainage.
- 4. The applicant has submitted to the Board of Appeals a plot plan and scaled architectural drawings which comply with Section 3.7.3 below
- **3.7.3 Design and Dimensional Standards.** A plot plan and scaled architectural drawings of the existing Dwelling Unit and proposed addition shall be submitted, showing location of all Buildings on the Lot, the proposed addition, location of all septic systems and parking and compliance with the following items:
 - 1. The Lot on which the proposed conversion is to be located is at least ten (10) acres.
 - 2. Any external stairways shall be screened from view, buffered or located out of sight from any Street.
 - 3. One off street parking space shall be provided for the additional Dwelling Unit, in addition to the required parking for the principal Dwelling unit. Every effort shall be made to minimize the visibility of the additional parking space by location and screening.
- **3.7.4 Other Approvals.** The applicant shall, in addition to obtaining a special permit, also obtain a building permit, and any necessary Conservation Commission approvals, and an occupancy permit prior to the occupancy of the proposed accessory apartment.

3.7.5 Special Restrictions.

- 1. The apartment shall not be transferred into separate ownership from the principal Dwelling, including a condominium form of ownership or otherwise.
- 2. The lot upon which the principal Dwelling Unit and accessory apartment are located shall not be reduced in size to less than ten (10) acres.

SECTION 4.0 DIMENSIONAL AND DENSITY REGULATIONS

4.1 GENERAL REGULATIONS.

4.1.1 Applicability. The regulations for each district pertaining to minimum lot area, minimum Lot Frontage, minimum lot depth and width, minimum Front Yards, Rear Yards and Side Yards, maximum height of Buildings, maximum number of Stories, and maximum Building coverage shall be as specified in this Section and set forth in the Table of Dimensional Regulations, and subject to the further provisions of this Section.

4.1.2 Table of Dimensional Regulations. The Table of Dimensional Regulations is hereby declared to be a part of this By-law.

TABLE OF DIMENSIONAL REGULATIONS

	R-1a	R-1b	RA	В
Minimum Lot Area per Dwelling Unit (sq. ft.)	20,000	40,000	80,000	ZBA1
Minimum Lot Frontage (ft.)	125	175	175	ZBA2
Minimum Lot width and depth (ft.) (for Dwellings, see also Sections 4.2.2, 4.2.6 and 4.3)	100 at building	100 at building	100 at building	ZBA3
Maximum Building Height (ft.)	35	35	35	35
Maximum number of Stories	3	3	3	3
Maximum Building coverage of Lot (%)	25	25	25	754
Minimum Front Yard (ft.) (See also Section 4.2.4)	25/505	25/50s	25/505	25/505
Minimum Side Yard and Rear Yard (ft.)	15	15	15	ZBA6

- 1. For each Dwelling Unit, minimum lot area is twenty thousand (20,000) square feet or as determined by the SPGA. For other uses, to be determined during site plan review.
- 2. For a Building that contains a Dwelling Unit, 125 feet. For all other uses, to be determined during site plan review.
- 3. To be determined during site plan review.
- 4. Or as determined during site plan review.
- 5. Twenty-five (25) feet from the Street line, or fifty (50) feet from the Street center line, whichever is greater.
- 6. For a Dwelling Unit, 15 feet. For all other uses, to be determined during site plan review.
- **4.1.3 Computation of Lot Area.** In computing the area of any Lot, no part of a public or private way and no part of a pond or river shall be included. No portion of any Lot that is less than seventy-five (75) feet in width may be counted toward meeting the minimum Lot area required under this Section.
- **4.1.4 Irregular Lots.** In the event of an irregularly shaped Lot and a question as to the identification of the appropriate side Lot lines or front and rear lines for the purposes of computing the minimum Lot width and depth, the matter shall be decided by the Building Commissioner, after consultation with the Planning Board.

4.1.5 Accessory Buildings. Accessory Buildings, including detached garages, shall not be located closer than twenty-five (25) feet from the Street line, nor any closer to any side or rear Lot line than the height of such accessory Building above the ground or twenty (20) feet, whichever is greater; provided, however, that one storage shed not larger than eight (8) feet by twelve (12) feet and not greater than ten (10) feet in height may be located not less than five (5) feet from the rear and side Lot lines. No accessory Building shall be located closer than twenty (20) feet from any Dwelling or main Structure on a Lot unless both Structures conform with regulations relative to fire safety.

4.2 SPECIAL REGULATIONS.

- **4.2.1 Exemptions from Height Limits.** Limitation of height shall not apply to such Structures as churches, belfries, flagpoles, chimneys, radio and television antennae, silos, water tanks and similar non-habitable Structures. See Section 7.3 for height limitations applicable to Wind Energy Facilities.
- **4.2.2 Required Circle in Residential Districts.** To ensure that no residential Lot shall be laid out which is too irregular in shape, (except for those larger Lots with frontage exceptions), a Lot shall be such that a circle can be drawn on a plan between side Lot lines and tangential to the Lot Frontage, with such circle being entirely contained within the Lot's perimeter and having a diameter of ninety (90) percent of the required frontage: one-hundred twelve and a half (112.5) feet in the R-1a District; and one hundred fifty-seven and a half (157.5) feet in the R-1b and R-A Districts.
- **4.2.3 Frontage Exception for Larger Lots.** A Lot in an R-la, R-1b, or R-A District need not have the specified amount of Lot Frontage provided that:
 - 1. The area of the Lot exceeds by at least three (3) acres the minimum area required for such an R-la, R-1b, or R-A District or Groundwater Protection Overlay District.
 - 2. The Lot has a minimum continuous Street frontage of not less than fifty (50) feet and a width of not less than fifty (50) feet at any point between the Street and the site of the Dwelling.
 - 3. The minimum width of Lot measured at the shortest distance between side Lot lines taken through the Dwelling site on said Lot shall be a minimum of one hundred fifty (150) feet.
 - 4. There is not more than one other such Lot with frontage contiguous to the subject Lot.
 - 5. The Lot is not, in the opinion of the Planning Board so located as to block the future extension of a dead end Street.
 - 6. No such Lot on which a Dwelling is located shall be further subdivided, reduced in area, or changed in size or shape.

- **4.2.4** Averaging of Setbacks. No Building need be set back more than the average setback of the Buildings on either side, a vacant Lot being counted as though occupied by a Building set back fifty (50) feet from the Street center line.
- **4.2.5 Double Frontage.** In case of Lots abutting on more than one (1) Street, the full width of the Front Yard shall be provided from each Street.
- **4.2.6** Width. No residential Lot shall have a Lot width measured between side Lot lines of less than seventy-five (75) feet at any point in the buildable portion of said Lot. Every residential Lot laid out after the adoption of this amendment shall have a minimum depth of seventy-five (75) feet along eighty (80) percent of the minimum Lot Frontage.
- **4.2.7 Minimum Lot Dimensions When Creating a New Street.** No Lot shall be altered to create a new Street laid out under the Subdivision Control Law unless the center line of the new Street is at least one hundred twenty-five (125) feet from the side line of the Lot, and the Lot as altered maintains the minimum Lot Frontage on the existing Street.
- **4.2.8 Watercourse Yards.** There shall be a Yard or setback between any Building or Structure in any district and a watercourse, stream, swamp or floodway of a width to leave the area subject to flooding free of obstruction.
- **4.2.9** Appurtenant Open Space. No Yard or other open space required for a Building by this By-law shall, during the life of such Building, be occupied by or counted as open space for another Building.
- **4.2.10 Corner Clearance.** Within an area formed by the side lines of intersecting Streets and a line joining points on such lines fifteen (15) feet distant from their point of intersection, or in case of a rounded corner, from the point of intersection of their tangents, no Structure shall be erected and no foliage maintained between a height of three and one half (3.5) feet and a height of eight (8) feet above the plane through their curb grades.
- **4.2.11 Wind Turbine Setbacks.** Wind Turbine setbacks must comply with all the Yard requirements in the Table of Dimensional Regulations, and also with the regulations for Wind Energy Facilities, Small Scale Wind Energy Facilities, and Utility Scale Wind Energy Facilities set forth in Section 7.3, below.

4.3 CONTIGUOUS AREA OF LAND REQUIREMENT.

- **4.3.1 General.** Every Lot laid out for residential use shall contain a contiguous parcel of land for building which:
 - 1. Is not in the Conservancy District, as shown on the Zoning Map;
 - 2. Is not an area designated as having severe limitations for septic systems or house sites on the Hamilton Soil Survey Maps, U.S. Soil Conservation Service; and

- 3. Is not a wetland as defined in G.L., c. 131, s. 40.
- **4.3.2 Interpretation.** To facilitate determination of the existence of the required building area, the applicant shall designate that area on each plan submitted under this By-law to the Building Commissioner, together with accompanying evidence to demonstrate compliance. The Building Commissioner in interpreting this provision may seek advice of the Planning Board.
- **4.3.3 Minimums.** The contiguous area of land for building referred to in the preceding paragraph shall meet the following minimums, provided that the Groundwater Protection Overlay District minimum set forth in subsection 4 below shall override the minimum for the underlying zone.
 - 1. Ten thousand (10,000) square feet in the R-la District;
 - 2. Twenty (20,000) square feet in the R-1b District;
 - 3. Forty thousand (40,000) square feet in the R-A District;
 - 4. Eighty thousand (80,000) square feet in the Groundwater Protection Overlay District for Lots which are not Lots qualifying under Section 4.2.3 as reduced frontage lots;
 - 5. One half (0.5) the sum of the minimum Lot size plus three (3) acres for Lots qualifying as reduced frontage Lots; and
 - 6. One hundred five thousand three hundred forty (105,340) square feet in the Groundwater Protection Overlay District for Lots qualifying as reduced frontage Lots.
- **4.3.4 Proof.** If evidence certified by a Registered Professional Engineer, such as test pits, borings, vegetative analysis and other methods acceptable to the Building Commissioner is presented by the applicant to show that the proposed building area does not, in fact, include the characteristics prohibited above, or that the boundaries on a map used to enforce this Section are in error as to said area, the Building Commissioner may accept this alternative evidence as the basis for the determination of compliance with this Section.
- **4.3.5** Location of House. The proposed Dwelling site must fall within the area claimed as buildable land under this Section.
- **4.3.6 Buildable Area.** No portion of any Lot that is less than seventy-five (75) feet in width may be counted as buildable area.

SECTION 5.0 NONCONFORMING USES AND STRUCTURES

- **5.1 APPLICABILITY.** This By-law shall not apply to Structures or uses lawfully in existence or lawfully begun, or to a Building or special permit issued before the first publication of notice of the public hearing required by G.L. c. 40A, s. 5 at which this By-law, or any relevant part thereof, was adopted. Such prior, lawfully existing nonconforming uses and Structures may continue, provided that no modification of the use or Structure is accomplished, unless authorized hereunder.
- **5.2 NONCONFORMING USES.** The Zoning Board of Appeals may by Finding allow a change to a nonconforming use in accordance with this Section only if it determines that such change shall not be substantially more detrimental than the existing nonconforming use to the neighborhood.
- **5.2.1** Eligible Changes. The following types of changes to nonconforming uses may be considered by the Zoning Board of Appeals:
 - 1. Change or substantial extension of the use.
- 5.3 NONCONFORMING STRUCTURES. The Zoning Board of Appeals may by Finding allow a change to a nonconforming Structure in accordance with this Section only if it determines that such change shall not be substantially more detrimental than the existing nonconforming Structure to the neighborhood.
- **5.3.1 Eligible Changes.** The following types of changes to nonconforming Structures may be considered by the Zoning Board of Appeals:
 - 1. Reconstruction, extension or structural change; and
 - 2. Alteration to provide for a substantially different purpose or for the same purpose in a substantially different manner or to a substantially greater extent.
- **5.3.2 Variance Required.** Except as provided in Section 5.3.3 and 5.3.4, below, a variance from the Zoning Board of Appeals shall be required for:
 - 1. The change of a nonconforming Structure in such a manner as to increase an existing nonconformity or create a new nonconformity; and
 - 2. The extension of an exterior wall at or along the same nonconforming distance within a required Yard.
- **5.3.3 Nonconforming Single and Two Family Residential Structures.** Nonconforming Single and Two Family Dwelling Structures may be changed only upon a determination by the Building Commissioner that such proposed change does not increase the nonconforming nature of said Structure by more than one hundred (100) percent of the Residential Gross Floor Area.

- **5.3.4.** Eligible Changes. The following circumstances shall not be deemed to increase the nonconforming nature of Nonconforming Single and Two Family Dwelling Structures and the Building Commissioner may issue a building permit:
 - 1. The existing Structure is on a conforming Lot but is nonconforming because it encroaches on a setback and the proposed alteration or extension will not change the setback which fails to conform, and the entire structure meets all other requirements of this By-law including but not limited to height and Lot coverage.
 - 2. The existing Structure is nonconforming solely because it is located on a Lot which is nonconforming as to size and/or Lot Frontage as the result of a zoning change, and the existing Structure and proposed alteration or extension meets all other current requirements of the By-law including but not limited to setbacks, height and Lot coverage.
 - 3. The existing Structure is a residence in a district where residences are permitted, the Structure is situated on a nonconforming Lot or has nonconforming Yards, and the proposed alteration or extension is a reconstruction or repair which does not change the Lot size or Yards which fail to conform. For the purposes of this subsection only, the term "reconstruction" shall not include the voluntary demolition of such Structure and its rebuilding. See Section 5.5.

If the Building Commissioner determines that proposed change exceeds one (1) or more of the criteria set forth above, the Zoning Board of Appeals may, by Finding, allow such change where it determines that the proposed modification will not be substantially more detrimental than the existing nonconforming structure to the neighborhood.

- **5.4 ABANDONMENT OR NON-USE.** A nonconforming use or Structure which has been abandoned or not used for a period of two (2) years shall lose its protected status and be subject to all of the provisions of this By-law.
- **5.5 RECONSTRUCTION AFTER CATASTROPHE OR DEMOLITION.** Any nonconforming Structure may be reconstructed after a catastrophe or after voluntary demolition in accordance with the following provisions:

5.5.1 Procedures.

- 1. Reconstruction of said premises shall commence within two (2) years after such catastrophe or demolition, with the reconstruction completed and the Structure occupied within a reasonable time thereafter.
- 2. Building(s) reconstructed as of right shall be located on the same footprint as the original nonconforming Structure and shall be only as great in volume or area as the original nonconforming Structure.

- 3. In the event that the proposed reconstruction would (a) cause the Structure to exceed the volume or area of the original nonconforming Structure, or (b) cause the Structure to be located other than on the original footprint, a special permit shall be required from the Zoning Board of Appeals. In the case of voluntary demolition, the special permit shall be obtained prior to such demolition.
- 5.6 REVERSION TO CONFORMITY OR MORE RESTRICTIVE USE. No nonconforming use shall, if changed to a conforming use or to a more restrictive use, revert to a nonconforming or less restrictive use.
- 5.7 SUBSTANDARD LOTS. When a prior lawful nonconforming Structure is located on a Lot which does not meet current dimensional requirements, such Lot shall not be changed, unless the change does not result in exacerbation of an existing nonconformity or a new nonconformity.

SECTION 6.0 GENERAL REGULATIONS

6.1 OFF-STREET PARKING AND LOADING AREAS.

6.1.1 Number of Required Parking Spaces. Off-street parking spaces and loading areas shall be required in at least the ratio specified below for the following uses of land and Buildings:

USE	REQUIRED SPACES
Dwellings	One (1) for each Dwelling Unit
Places of public assembly not including churches:	One (1) for each three seats therein
Schools	One (1) for each classroom therein, plus one (1) for each two (2) employees or staff positions other than teachers; and where an auditorium is provided, one (1) for each three (3) seats therein
Other service establishments and retail businesses	One (1) for each three hundred (300) square feet or fraction thereof of Business Gross Floor Area, excluding basement storage area
Other uses requiring off-street parking and loading space	Space in accordance with anticipated needs as determined by Board of Appeals with the advice of the Planning Board

6.1.2 Location. Required off-street parking and loading spaces shall be located on the same Lot as the Building or use they are intended to serve, or in the case of parking spaces, on other premises within two hundred (200) feet of such Lot.

- **6.1.3** No Reduction. Required off-street parking and loading spaces shall not hereafter be reduced, nor any loading space counted as or substituted for a parking space.
- **6.1.4** Size. Required off-street parking spaces shall each be nine (9) feet by eighteen (18) feet for each vehicle suitable for parking together with areas for turning and exclusive of necessary drives and other access ways from the street to the parking area; provided, however, that a Driveway or garage may be considered as the required parking space for a Single Family Dwelling.
- **6.1.5** Access. Required off-street parking and loading spaces shall all have adequate vehicular access to a Street.
- **6.1.6 Shared Parking.** No part of an off-street parking or loading space required for any Building or use shall be included as a part of an off-street parking area required for another Building or use unless a determination is made by the Board of Appeals to the effect that the period of usage of such structure or uses will not be simultaneous.
- **6.1.7 Surface.** All off-street parking areas and loading areas, other than those provided for Dwellings but including drives and other access ways, shall be treated with bituminous or other surfacing material, and shall be provided where necessary with appropriate bumper and wheel guards.
- **6.1.8 Lights.** Illumination shall be so arranged as to deflect the light away from adjoining Lots and abutting Streets; and screening shall be provided where required by this By-law.
- **6.1.9** Site Plan. Parking areas and loading areas in the Business District shall be shown on a site plan, as provided in Section 10.6.
- **6.1.10** Special Permit. The Planning Board may, by special permit, waive any provision of this Section, provided that such waiver shall not result in substantial detriment.

6.2 DRIVEWAYS.

- **6.2.1 Driveway Plan.** Driveway access to a Dwelling or other Building site shall be indicated on a plan submitted to the Building Commissioner at the time of a building permit application. In determining compliance with the following standards, the Building Commissioner shall confer with the Department of Public Works.
- **6.2.2 Standards.** Driveways shall provide both access and turnaround for vehicles including moving vans, ambulance, fire and police vehicles without substantial hardship, financial or otherwise in construction. Reconstruction and/or resurfacing of an existing Driveway, which will result in a significant change of grade in the Driveway shall require approval of the Department of Public Works. All Driveways shall have:
 - 1. Width of at least ten (10) feet but shall be cleared to a width of at least twelve (12) feet;

- 2. A centerline radius of at least sixty (60) feet in the driveway curves;
- 3. Maximum grade of ten (10) percent;
- 4. Flare of five (5) feet at intersection of Driveway with public way; and
- 5. A leveling-off section.
- **6.2.3 Hydrant.** A hydrant shall be required if a proposed Dwelling or other Building is greater than one thousand (1000) feet as measured along the proposed Driveway from an existing or proposed fire hydrant. In such a case, a town water line and hydrant acceptable to the Department of Public Works shall be installed. Such hydrant shall be not more than one thousand (1000) feet along the Driveway from the Dwelling as measured along the proposed Driveway.
- **6.2.4 Connection to State Highway.** Plans showing special requirements for Driveways connecting with state-maintained ways are available at District or State Offices of the Massachusetts Department of Public Works.
- **6.2.5 Connection with Access Street.** The connection of any access way with a Town Street must be approved by the Department of Public Works in consultation with the Police Department for safety issues. In connection with said approval, the owner shall release the Town from all damages resulting from the flow of water from such way. Said approval shall be in the form of a signed curb cut permit.
- **6.2.6 Common Driveway.** Except by special permit granted by the Planning Board, no more than two lots may share an access Driveway. The Town may require two (2) or more Lots to share a common Driveway when, in the opinion of the Police Department, it is deemed necessary for safety purposes. No building permit shall be granted for any Lot served by a common Driveway until there has been compliance with the following:
 - 1. Curb Cut Permit. A curb cut permit has been obtained pursuant to Section 6.2.5 above;
 - 2. Location. The common Driveway shall lie entirely within the Lots to which it provides access, and shall comply with all Driveway regulations in this Section.
 - 3. *Easement*. An easement providing for maintenance and snow removal and running with the land in perpetuity has been executed by the owner(s) of the Lots sharing the Driveway and recorded at the Registry of Deeds, and evidence thereof is submitted to the Building Commissioner.
 - 4. *Frontage*. The common Driveway is not being used to satisfy zoning frontage requirements.

6.3 SIGNS.

- **6.3.1 Permit Required.** All signs require building permit approval.
- **6.3.2 Residence Districts.** In every residential district, signs will be permitted as follows:
 - 1. One (1) sign pertaining to the lease, sale or use of a Lot or Building on which such sign is placed not exceeding a total area of six (6) square feet.
 - 2. One (1) sign for identification of professional and home occupations or the occupant, not exceeding a total area of three (3) square feet.
 - 3. For directional purposes, one (1) sign not exceeding a total area of three (3) square feet.
 - 4. Subject to the grant of a special permit by the Planning Board for specified periods not exceeding one (1) year, other and larger signs.
- **6.3.3 Business District.** In the Business District, signs will be permitted as follows:
 - 1. Advertising goods and services available on the premises not exceeding one (1) square foot for every linear foot of store frontage and in no case exceeding a total area of thirty six (36) square feet per lot.
 - 2. For identification of the business, company or agency on a wall or parapet of a main Building not exceeding thirty (30) square feet for each separate business in a Business District.
 - 3. One (1) sign pertaining to the lease or sale of a lot or Building on which such sign is placed not exceeding a total area of six (6) square feet.
 - 4. For the purpose of identifying a business development or shopping center, one (1) free-standing sign with a total of thirty-six (36) square feet of area for each street on which the business development fronts.

6.3.4 Standards.

- 1. Projecting Sign Clearance. A clear space of not less than twelve (12) feet shall be provided below all parts of such signs.
- 2. *Prohibited*. No sign in any district shall be flashing, internally illuminated, animated or illuminated by other than white light, nor project beyond any Street line or above any Building.

6.4 STANDARDS FOR NONRESIDENTIAL BUILDINGS' CONSTRUCTION, USE AND OPERATION.

- **6.4.1 General.** Nonresidential Buildings and uses permitted in the Table of Uses shall conform to the following minimum standards for construction, use and operation as evidenced by detailed plans submitted to the Building Commissioner for review and certified as to compliance by the architects and/or engineers responsible for such plans. In the event of any reasonable doubt by the Building Commissioner as to compliance with the following minimum standards, he or she shall refer the building permit in such cases to the Zoning Board of Appeals for resolution.
 - 1. Waste Disposal and Water Service. Water service, and waste and refuse disposal methods shall comply with pertinent health regulations and shall be in accordance with the approved site plan.
 - 2. Storage. Equipment, including company owned or operated vehicles, shall not be stored between the Street line and the front line of structures on the subject Lot, or if there be no Structure within fifty (50) feet of the Street line, and in no case to be visible from the Street.
 - 3. Screening. Screening by fences, walls and/or evergreen planting, in accordance with an approved site plan, shall be provided, erected and maintained to shield the business uses of land and Buildings from any adjoining residential property.

SECTION 7.0 SPECIAL REGULATIONS

7.1 ADULT ENTERTAINMENT USES.

- 7.1.1 Purpose. It has been documented in numerous towns and cities throughout the Commonwealth of Massachusetts and elsewhere in the United States that Adult Entertainment Uses are distinguishable from other business uses and that Adult Entertainment Uses degrade the quality of life in the areas of a community where they are located. Studies have shown secondary impacts such as increased levels of crime and blight resulting from the clustering and concentration of Adult Entertainment Uses. Late night noise and traffic also increase due to the late hours of operation of many of these establishments. This Section is adopted pursuant to G.L. c. 40A, s. 9A with the purpose and intent of regulating and limiting the location of Adult Entertainment Uses (as defined herein) so as to prevent the secondary effects associated with these establishments and to protect the health, safety, and general welfare of the present and future inhabitants of the Town of Hamilton.
 - 1. The provisions of this Section have neither the purpose nor effect of imposing a limitation or restriction on the content of any communicative matter or materials, including sexually oriented matter or materials.
 - 2. Similarly, it is not the intent or effect of this Section to restrict or deny access by adults to sexually oriented matter or materials protected by the Constitution of the United

States or of the Commonwealth of Massachusetts, nor restrict nor deny rights that distributors or exhibitors of such matter or materials may have to sell, distribute, or exhibit such matter or materials. Neither is it the intent or effect of this subsection to legalize the distribution of obscene matter or materials.

- **7.1.2 Special Permit.** No Adult Entertainment Use shall be established or operated in the Town unless a special permit has been granted by the Zoning Board of Appeals. An application that meets the standards in Section 7.1.4 shall be granted.
- 7.1.3 Location. No such use shall be located within the following designated areas. All distances specified below shall be measured by a straight line from the nearest property line of the premises on which the adult entertainment establishment is to be located to the nearest boundary line of a residential zoning district, or to the nearest property line of any residential use, public or private school, church or other religious facility, public park, or recreational area, group day care center, family day care center, library, nursing home, hospital or retail, restaurant, or cabaret use whose principal business falls under G.L. c. 138, s. 12, or any other Adult Entertainment Use as the case may be:
 - 1. Two hundred (200) feet from the nearest boundary of any residential zoning district or from the nearest property line of any residential use;
 - 2. Five hundred (500) feet from the nearest property line of any public or private school;
 - 3. Two hundred (200) feet from the nearest property line of any church or other religious facility;
 - 4. Five hundred (500) feet from the nearest property line of any public park or recreation area and any principal or accessory private recreational facility use, or area where youths commonly gather;
 - 5. Five hundred (500) feet from the nearest property line of any group day care center, family day care center, library, nursing home and hospital;
 - 6. One thousand (1,000) feet from the nearest property line of any other Adult Entertainment Use;
 - 7. Two hundred (200) feet from the nearest property line of any retail, restaurant, or cabaret use whose principal business falls under G.L. c. 138, s. 12; and

An Adult Entertainment Use may not be allowed within a building containing other retail, consumer or residential uses, or within a shopping center, shopping plaza, or mall.

- 7.1.4 Standards. An Adult Entertainment Use shall comply with the following standards:
 - 1. Parking. The Adult Entertainment Use shall comply with all off-street parking requirements contained in Section 6.1.

- 2. Signs. No Adult Entertainment Use may have a freestanding accessory sign or a sign in excess of a height of fifteen (15) feet. Further, no sign shall rotate, be illuminated or contain reflective or fluorescent elements which will sparkle in sunlight. Signs will comply with all requirements contained in Section 6.3.
- 3. Flashing Lights. No Adult Entertainment Use may have any flashing lights visible from outside the establishment.
- 4. *Displays*. No pictures, publications, videotapes, movies, covers or other advertising items shall be displayed in the windows of or on the building of any adult bookstore, adult cabaret, adult motion picture theater, adult paraphernalia store or adult video store.
- 5. *Inside Views*. All building openings, entries, and windows for any adult entertainment use shall be located, covered, or screened in such a manner as to prevent a view into the interior of an Adult Entertainment Use Building from any area open to the general public.
- 6. Noise. The Building in which the Adult Entertainment Use will be located shall be designed so that noise from any proposed entertainment is not audible (0 decibels) outside.
- 7. Dimensional and Density Regulations. All standards and conditions set forth in Section 4.0 shall be applicable to the granting of a special permit under this Section.
- 8. *Prohibition.* No special permit may be granted hereunder to any person convicted of violating the provisions of G.L. c. 119, s. 63, or G.L. c. 272, s. 28.
- 7.1.5 Site Plan. An Adult Entertainment Use is subject to site plan approval pursuant to Section 10.6. A site plan shall be submitted by the applicant in order that the SPGA may determine that the above standards have been met. The site plan shall meet all the requirements of Section 10.6, and shall show the distances between the proposed Adult Entertainment Use and any residential zoning district, public or private school, church, or other religious facility, public park or recreation area, group day care center, Child Care Center, Family Child Care Center (Small or Large), library, Long Term Care Facility, and hospital, retail, restaurant, or cabaret use whose principal business falls under G.L. c. 138, s. 12, and any other Adult Entertainment Use.
- 7.1.6 Imposition of Other Conditions. The SPGA may impose in addition to any applicable conditions specified herein, such additional conditions as it finds reasonably appropriate to safeguard the neighborhood or otherwise serve the purposes of this Section, including but not limited to the following:
 - 1. Front Yards, Side Yards, or Rear Yards greater than the minimum required by Section 4.0;
 - 2. Screening buffers or planting strips, fences or walls;

- 3. Limitations upon the size, method and time of operations;
- 4. Time duration of the permit;
- 5. Regulation of number and location of driveways or other traffic features; and
- 6. Off-street parking.
- 7.1.7 Lapse of Permit. Any special permit granted hereunder for an Adult Entertainment Use shall lapse: (a) after one (1) year from its date of issuance, including such time required to pursue or await the determination of an appeal from the grant thereof, (b) if a substantial use thereof has not sooner commenced except for good cause, or (c) in the case of a permit for construction, if construction has not begun by such date except for good cause.

7.2 COMMUNICATION TOWERS (CT) AND TELECOMMUNICATION ANTENNA FACILITIES (TAF).

- **7.2.1 Purpose.** The purpose of this Section is to establish general guidelines for the siting of a communication towers ("CT") and telecommunication antenna facility ("TAF"). The goals of this Section are:
 - 1. To minimize the adverse visual impacts of CTs and TAFs;
 - 2. To avoid damage to adjacent properties;
 - 3. To lessen impact on surrounding properties;
 - 4. To lessen impact on traffic;
 - 5. To encourage the location of towers on municipal land;
 - 6. To minimize the number of towers throughout the community;
 - 7. To require the co-location of new and existing tower and facility sites;
 - 8. To encourage users of towers and facilities to locate them, to the extent possible, in areas where the adverse impact on the community is minimal; and
 - 9. To make available all CT and TAF locations to local municipal agencies.
- **7.2.2 Permitting.** No CT or TAF may be erected without first obtaining a special permit from the Planning Board.
- **7.2.3 Application.** An application for a special permit under this Section shall be filed with the SPGA in compliance with Planning Board Rules and Regulations Governing Special Permits, and applicable provisions of this By-law. A site plan shall be prepared by a Registered

Professional Engineer licensed by the Commonwealth of Massachusetts and shall include the following minimum requirements, in addition to the site plan requirements in Section 10.6:

- 1. A locus map will be prepared by a Registered Professional Land Surveyor and shall show all abutters, streets, bodies of water, landscape features, historic sites, habitats for endangered species, and rights of way within two hundred (200) feet of the facility, and all structures within five hundred (500) feet of the facility;
- 2. Tower and/or facility location, including guy wires, if any, and tower height;
- 3. Topography;
- 4. Fencing and landscaping;
- 5. Access and parking;
- 6. Lighting;
- 7. Areas to be cleared of vegetation and trees; and
- 8. Site boundaries.
- **7.2.4 Required Reports.** Reports prepared by one or more qualified professionals shall be submitted with a site plan application that:
 - 1. Describe the tower, the facility, and the technical, economic and other reasons for the tower and facility design, and the need for the tower at the proposed location;
 - 2. Demonstrate that the tower and facility comply with all applicable standards of the Federal and State governments, including, but not limited to, radio frequency emissions, air navigation safety, and environmental impact. The applicant should also demonstrate that the facility and/or tower should also demonstrate compliance with applicable industry standards for structural integrity, such as EIA/TIA Standard 222 in its most current revision;
 - 3. If a tower or other new antenna support structure is proposed, describe the capacity of the structure to accommodate additional communications services including the number and type of communications facilities that it can accommodate and the basis for the calculation of capacity;
 - 4. Demonstrate that the tower and site comply with this Section;
 - 5. Describe the role the proposed facility will play in the applicant's development of its communications network, including demonstrating the need for the proposed height and location by making a technical showing. Such a technical showing will include showing coverage from existing and other proposed facilities in the region, coverage anticipated

from the proposed facility, and coverage from the proposed facility at lower antenna elevations (when relevant);

- 6. Describe how the applicant's communications needs are expected to evolve over a five (5) to ten (10) year period, with particular emphasis on how it will affect the need for the proposed facility and for additional facilities in the Town;
- 7. Describe other feasible sites, including existing sites, if any. Demonstrate with a technical evaluation showing that they do not have significantly less impact on the community than the proposed facility and/or demonstrate that the coverage obtained from one or more alternative installations is significantly poorer than that from the proposed facility. The Planning Board may require data for comparing the alternatives with the proposed facility, including coverage maps and/or visibility maps of the proposed facility and the alternatives;
- 8. Describe the appearance, location, and anticipated use of any generators that support the functions of the tower; and
- 9. Applicant shall provide evidence that the proposed facility or tower will comply with or is exempt from applicable regulations administered by the Federal Aviation Administration (FAA), Federal Communications Commission (FCC), Massachusetts Aeronautics Commission, the Massachusetts Department of Public Health, and other relevant regulatory agencies. Such evidence may be in the form of official agency documentation, documentation prepared by a qualified professional, or other forms of written evidence satisfactory to the Planning Board. The applicant shall send a subsequently received agency statement, if any, to the Planning Board, or the Planning Board may require verification of agency determinations by the Building Commissioner prior to issuance of a building permit.
- 7.2.5 On Site Demonstration. When a tower or other new antenna support structure is proposed, this subsection shall apply. Between the date of public notice of the public hearing and the hearing itself, a balloon shall be put in place at the height of the proposed tower or support structure, for not less than three (3) days. The balloon shall be floated at a height equal to the top of the proposed tower or support structure. The applicant may propose alternative means of complying with this requirement. The applicant shall also prepare for the public hearing photosimulations of the tower on that site from key locations as determined by the Planning Board. The Planning Board may require that a visibility map be created by a qualified professional to indicate where and to what degree the proposed structure and/or antennas will be visible from public ways or land. Such map may be used to compare the visibility of the proposed facility with maps of the visibility of existing facilities, of alternative heights of the proposed facility, or of feasible alternative facilities.
- **7.2.6 Requirements.** The tower and its appurtenances shall be located in accordance with all applicable federal and state regulations in effect at the time of construction and further, the operation shall comply with all such regulations of these agencies during the entire period of operation. In addition, the tower and its facilities shall be located within the Town as follows:

- 1. New Towers. New towers shall be considered only upon a determination by the Planning Board that one (1) or more existing structures or approved towers cannot physically accommodate, or cannot provide substantially the same service as, the wireless communication facilities on the proposed tower.
- 2. Co-Location. Towers shall be designed and made available for co-location of as many service providers possible, based on the structure height and appearance in the context of the tower sites. To the extent feasible for their sites, communications accessory Buildings shall have the appearance of a single Building, or otherwise be designed to provide an orderly appearance compatible with the community.
- 3. *Setback*. The setback from property lines shall be no less than a distance at least equal to the height of the tower. Whenever possible, setbacks behind existing trees and/or Buildings is encouraged to provide visual masking of the tower and facility.
- 4. Wetlands. Setback from designated wetlands and water bodies shall be at least one hundred and fifty (150) feet.
- 5. Existing Structures. Distance from all existing structures shall be at least two hundred (200) feet, excluding existing structures on the lot.
- 6. Fencing. Fencing shall be provided to control access to the base of the tower, and such fencing shall be compatible with the scenic character of the Town and shall not be of barbed wire or razor wire.
- 7. Access. Access shall be provided to a site by a roadway that respects the natural terrain, does not appear as a scar on the landscape, and is approved by the Planning Board and the Fire Chief to assure emergency access at all times.
- 8. *Design*. Consideration shall be given to design that minimizes erosion, construction on unstable soils, and steep slopes.
- 9. Height. The applicant shall demonstrate to the satisfaction of the Planning Board that the location of the CT and/or TAF is necessary and that the size and height are the minimum necessary for the purpose. The Planning Board shall give consideration to the possible need for additional height to accommodate future co-locators. It may require that a structure be built to, or be designed to be extended to, a future height limit.
- 10. Signs. There shall be no signs, except for announcement signs, applicable warning signs, no trespassing signs and a required sign giving a phone number where the owner can be reached on a twenty-four (24) hour basis. All signs shall conform to the sign requirements of Section 6.3, and shall be subject to conditions of the Site Plan Review and any special permit.

- 11. *Interconnections*. To the extent feasible all cabled network interconnections and utilities from and to the communications site shall be installed underground.
- 12. Visual Impact. The CT and/or TAF shall minimize adverse visual effect on the environment and the residents of Hamilton. The Planning Board may impose reasonable conditions to ensure this result, including, but not limited to: painting, lighting standards, screening, and camouflage.
- **7.2.7 Clearing.** Clearing shall be performed in a manner that will maximize preservation of natural beauty and conservation of natural resources, which will minimize marring, and scarring of the landscape or silting of streams.
 - 1. The time and method of clearing rights-of-way should take into account soil stability, the protection of natural vegetation, the protection of adjacent resources, such as the protection of natural habitat for wildlife, and appropriate measures for the prevention of silt deposition in water courses.
 - 2. Clearing of natural vegetation should be limited to that material which poses a hazard to the CT or TAF.
 - 3. The use of "brush blades" instead of dirt blades on bulldozers is recommended in clearing operations where such use will preserve the cover crop of grass, low growing brush, or other vegetation.
 - 4. Areas should be cleared only when necessary to the operation, maintenance, and construction of the tower.

7.2.8 Design Requirements.

- 1. Visual impacts of the CT and/or TAF shall be minimized.
- 2. Concealed or camouflaged facilities shall be provided to the maximum extent technically practicable. To the maximum extent practicable, towers and antenna facilities, if not camouflaged, shall be designed to provide a clean architectural appearance and to minimize any visually cluttered appearance.
- 3. The applicant shall demonstrate that the proposed CT and/or TAF is the minimum height necessary to accommodate transmitters and receivers.
- 4. Tower height shall be limited to one hundred ten (110) feet and shall be designed to accommodate the use for which the application is made. However, applications for CTs in excess of one hundred and ten (110) feet may be approved if accompanied by a detailed technical information showing as to the necessity of the additional height (colocation, reduced tower density in community, etc.).
- 5. All CTs greater than one hundred ten (110) feet in height shall be monopole in type.

- 6. Night lighting of towers shall be prohibited unless required by the Federal Aviation Administration (FAA). Lighting shall be limited to that needed for emergencies and/or as required by the FAA. To the fullest extent practicable, facilities and towers shall be designed to avoid FAA required marking and/or lighting requirements.
- 7. Siting shall be such that the view of the tower from other areas of Town shall be as minimal as possible. The Planning Board will pay particular attention to the visibility of the proposed structure from public ways, from historic districts, from popular scenic areas, and from natural views from residential neighborhoods.
- 8. Access roads shall be designed to visually conceal all CTs.
- 9. Shared use of towers and facilities is to be encouraged. When sharing a tower is technically not practical, replacement of the tower or the addition of a new tower may be considered. Towers shall be separated on a site so that each tower will not significantly impact the other.
- 10. The tower and/or facility shall be designed to accommodate the maximum number of users technologically practical.
- **7.2.9** Approval. A special permit shall be granted by the Planning Board in accordance with the provisions of this Section and Section 10.5.2. Any extension of the tower or antenna support structure, addition of antennas not previously permitted, replacement of antennas with substantially different configuration or appearance, or construction of new or replacement towers or antenna arrays shall be subject to an amendment to the special permit, following the same procedure as for an original grant of a special permit.
- **7.2.10 Construction Deadline.** Construction under a special permit shall be completed within eighteen (18) months of the issuance of the special permit unless otherwise provided for in the special permit, or in a subsequent amendment of the special permit approved by the Planning Board.
- 7.2.11 Conditions of Completion and Use. The date of completion shall be considered the date that the Building Commissioner issues the certificate of occupancy, or the first day of beneficial use, whichever comes first.
- **7.2.12 Lapse.** The special permit shall lapse after one (1) year from the date of issuance if substantial use or construction as granted by the permit has not commenced, except for good cause. Time required to pursue determination of an appeal shall not be considered part of the one (1) year limitation.
- **7.2.13 Abandonment.** Any tower or facility that is not operated for a continuous period of twelve (12) months occurring any time after the date of completion shall be considered abandoned. The owner of such tower and facility shall remove the same within ninety (90) days after the abandonment period.

- 7.2.14 Performance Guarantee. Insurance in a reasonable amount determined and approved by the SPGA after consultation at the expense of the applicant with one (1) or more insurance companies shall be in force to cover any liabilities that could arise out of the construction, operation and maintenance of the wireless facility and/or tower. Annual proof of said insurance shall be filed with the Town Clerk.
 - 1. An initial bond shall be posted to cover construction costs, in an amount approved by the Planning Board. The Planning Board may require posting of a bond to cover the cost of annual maintenance and repairs for the access road, site, and towers.
 - 2. Annual certification demonstrating continuing compliance with the standards of the Federal Communications Commission and Massachusetts Department of Public Health for radio frequency emissions regulations, and FAA for air navigation safety regulations, and other regulating agencies, shall be filed with the Building Commissioner by the special permit holder.
 - 3. Failure to post an approved bond and/or provide proof of insurance and/or obtain annual certification shall be grounds for enforcement.
- 7.2.15 Fees. The Planning Board may retain experts to review an application and the costs shall be borne by the applicant, pursuant to the provisions of G.L. c. 44, s. 53G. In addition, a special fee for approved applications will be assessed by the Town as determined by the Board of Selectmen. The intent of this fee is to offset the costs to the Town to process the application, to assess compliance with the by-laws, and to conduct any other business related to the construction and operation of towers within the Town. This fee will be assessed upon approval of the special permit. Failure to pay the fee in full will render the special permit subject to revocation and non-renewal.
- **7.2.16 Waivers.** The Planning Board may waive strict adherence with this Section if it finds that the safety and well being of the public will not be adversely affected by such waiver. For each waiver granted, the Planning Board shall make a written record indicating that the proposed tower and/or facility meets the purposes of Section 7.2.1.
 - 1. All requests for waivers shall be made in writing on a separate sheet (or sheets) of paper and be attached to the special permit application and be presented at the time of the initial application.
 - 2. Requests for waivers shall indicate the section number and the reason the applicant needs the waiver along with any documentation to support the request.
 - 3. The Planning Board shall grant requests for waivers only upon a majority vote. Each request shall be voted on separately. The applicant shall have the right to withdraw the request at any time prior to the actual vote. Once a request for a waiver is withdrawn it may not be presented again for a period of one (1) year.

- 4. Requests for more than three (3) waivers will be an indication to the Planning Board that:
 - a. The site is inadequate for the proposed use; or
 - b. The site plan is incomplete.
- 7.2.17 Term of Special Permit. The special permit granted under this Section shall expire within five (5) years of the date of issuance of the special permit. Renewal of the special permit shall follow the process and rules of Abbreviated Site Plan Review procedure as found in Section 10.6 of this By-law. In this case, for CT and TAF, the Planning Board shall administer the Abbreviated Site Plan Review. Each renewal shall be good for five (5) years unless specified otherwise by the Planning Board.
- 7.2.18 Other Requirements. Where the requirements of this Section 7.2 differ from or conflict with other requirements of this By-law, the requirements established herein shall prevail.

7.3 WIND ENERGY FACILITIES.

- **7.3.1 Purpose.** The purpose of this Section is to provide standards for the placement, design, construction, operation, monitoring, modification and removal of Wind Energy Facilities that address public safety, minimize impacts on scenic, natural and historic resources and to provide adequate financial assurance for the eventual decommissioning of such facilities.
- **7.3.2** Applicability. This Section applies to all Wind Energy Facilities proposed to be constructed after the effective date of this Section. This Section also applies to physical modifications to existing Wind Energy Facilities that materially alter the type, configuration, or size of such facilities or related equipment.
- **7.3.3 Compliance with Laws and Regulations.** The construction and operation of all Wind Energy Facilities shall be consistent with all applicable local, state and federal requirements, including but not limited to all applicable safety, construction, environmental, electrical, communications and aviation requirements.
- **7.3.4 Building Permit.** No Wind Energy Facility shall be erected, constructed, installed or modified as provided in this Section without first obtaining a building permit. The application for a building permit for a Wind Energy Facility must be accompanied by the fee required for a building permit as established by the Building Department.
- 7.3.5 Site Plan Review Required in the Commercial Overlay District. If the Wind Energy Facility is located or is to be located in the Commercial Overlay District, it shall not be erected, constructed, installed or modified without first undergoing site plan review by the Planning Board using the process in Section 10.6. A special permit is not required.
- **7.3.6** Special Permit Required Outside the Commercial Overlay District. If the Wind Energy Facility is located or is to be located outside the Commercial Overlay District, it shall not

be erected, constructed, installed or modified without first obtaining a special permit from the Zoning Board of Appeals. A separate site plan review by the Planning Board is not required.

- 7.3.7 Utility Scale Wind Energy Facilities Submittal Requirements. For both site plan review and the special permit process, the applicant shall submit to the appropriate Board a plan and supporting data, which shall bear the seal and signature of a registered professional engineer licensed to practice in Massachusetts. Submissions shall include a plan showing the information set forth below, except that in connection with a special permit application, the five hundred (500) foot distance noted below for inclusion of details may be reduced to three hundred (300) feet.
 - 1. Property lines of the applicant and abutters within 500 feet;
 - 2. Proposed location of the Wind Turbine;
 - 3. Location and uses (residence, garage, shed, etc.) of the proposed and abutters' Buildings within five hundred (500) feet including distance from the proposed wind turbine to each of the Buildings;
 - 4. Overhead transmission and distribution lines, and any radio or telecommunications towers within five hundred (500) feet of the Wind Turbine;
 - 5. Drainage or utility easements crossing within five hundred (500) feet of proposed power or control lines to or from the Wind Turbine;
 - 6. Location of guy wire anchors, if any;
 - 7. Location of all ways, Driveways or Streets, public or private, temporary or permanent within five hundred (500) feet of the proposed Wind Turbine;
 - 8. Existing areas of tree cover, including average height of trees, on the site parcel and any adjacent parcels within a distance measured from the Wind Turbine foundation, of one and one fifth (1.2) times the height of the Wind Turbine;
 - 9. Proposed changes to the landscape of the site, grading, vegetation clearing and planting, exterior lighting (other than FAA lights), screening vegetation or structures;
 - 10. A dimensional representation of the Wind Turbine mounted on its support structure, including foundation dimensions (both surface and depth), foundation materials, method of attaching tower to foundations, Wind Turbine dimensions including, tower height and rotor diameter, and clearance distances of blades to ground and nearest structure to be prepared by a Professional Engineer licensed to practice in the Commonwealth of Massachusetts;
 - 11. Statement as to whether the proposed Wind Turbine is a tested production model, or an experimental, one-of-a kind or prototype design including manufacturer and model;

- 12. Wind Turbine design data including manufacturer's specifications and installation/operation instructions; certification by the manufacturer or a registered engineer that tower design is sufficient to withstand wind load requirements for structures as established by the Massachusetts Building Code;
- 13. Site-specific wind speed data including the monthly mean wind-speed for a period no less than six months;
- 14. An operation and maintenance plan for the Wind Turbine and associated buildings, vegetation and roadways;
- 15. One or three line electrical diagram detailing Wind Turbine, associated components, and electrical interconnection methods, with all National Electrical Code compliant disconnects and over-current devices;
- 16. A location map consisting of a copy of a portion of the most recent USGS Quadrangle Map at a scale of one (1) inch to twenty-five thousand (25,000) square feet (1:25,000) showing the proposed facility site, including turbine site, and the area within at least two miles from the facility;
- 17. Documentation of actual or prospective access and control of the project site;
- 18. Proof of liability insurance;
- 19. Certification of height approval from the Federal Aviation Administration;
- 20. Statement that evidences the Wind Energy Facility's conformance with the provisions of the Department of Environmental Protection's Division of Air Quality Noise Regulations at 310 CMR 7.10; and
- 21. Description of financial surety in the form of either an escrow account, bond or otherwise, to cover the cost of removal in the event the Town must remove the facility and remediate the landscape, in an amount and form determined to be reasonable by the Planning Board, but in no event to exceed more than one hundred twenty-five (125) percent of the cost of removal and compliance with the additional requirements set forth herein, as determined by the applicant. Such surety will not be required for municipal or state-owned facilities. The applicant shall submit a fully inclusive estimate of the costs associated with removal, prepared by a qualified engineer. The amount shall include a mechanism for calculating increased removal costs due to inflation.

7.3.8 Utility Scale Wind Turbine Standards.

1. Setback. A Wind Turbine for a Utility Scale Wind Energy Facility may not be sited within: (i) a distance equal to the height of the Wind Turbine from Buildings, critical infrastructure, or private or public ways that are not part of the Wind Energy Facility; (ii)

three times the height of the Wind Turbine from the nearest existing residential or commercial Structure; or (iii) one and one-half (1.5) times the height of the Wind Turbine from the nearest property line. The applicable permit granting authority may reduce the minimum setback distance as appropriate based on site-specific considerations, or written consent of the affected abutter(s), if the project satisfies all other criteria for the granting of a building permit under the provisions of this Section.

- 2. Height. The height of Wind Turbine shall not exceed three hundred twenty-five (325) feet.
- 3. *Minimum Blade Height*. Minimum blade elevation shall be not less than fifteen (15) feet above the ground at the lowest point of blade arc.
- 4. *Sound*. The operation of the Wind Energy Facilities shall conform to the provisions of the Department of Environmental Protection's Division of Air Quality Noise Regulations at 310 CMR 7.10.
- 5. Signs. Wind Turbines shall not be used for displaying any advertising except for reasonable identification of the manufacturer or operator of the wind energy facility. Signs on Wind Energy Facilities shall comply with Section 6.3. The following signs are required:
 - a. At least one (1) sign shall be posted near ground level on the tower structure warning of high voltage and any danger;
 - b. Signage necessary to identify the owner, including a twenty-four (24) hour emergency contact phone number; and
 - c. Educational signs providing information about the facility and the benefits of renewable energy.
- 6. *Posted Information*. In addition, the following information shall be posted on a label on the generator or alternator of the Wind Turbine, and on the Wind Turbine control panel:
 - a. Maximum power output of system and wind speed at which it is achieved;
 - b. Nominal voltages and maximum current;
 - c. Manufacturer's name and address, model number and serial number;
 - d. Normal and emergency shutdown procedures; and
 - e. Maximum wind speed the Wind Turbine, in automatic unattended operation, can sustain without damage to structural components, or loss of ability to function normally.

- 7. Safety. The design of the proposed Wind Turbine shall be such that:
 - a. In the event of loss of utility power, the Wind Turbine shall not back feed a dead power line;
 - b. In the event of high wind speeds, the Wind Turbine shall brake or feather below survival wind speed;
 - c. In the event of blade imbalance, the Wind Turbine shall be able to support added blade weight of at least ten (10) percent at the tip of any blade; and
 - d. The applicant shall provide a copy of the project summary, electrical schematic and site plan to the police and fire departments. Upon request the applicant shall cooperate with local emergency services in developing an emergency response plan. All means of disconnecting the Wind Energy Facility shall be clearly marked. The applicant or facility owner shall identify a responsible person for public inquiries or complaints throughout the life of the project and that responsible person's contact information shall be provided to the Town Manager, Police and Fire Departments;
- 8. *Flicker*. Wind Energy Facilities shall be sited in a manner that minimizes showing or flicker impacts. The applicant has the burden of proving that this effect does not have significant adverse impact on neighboring or adjacent uses.
- 9. Wind Capacity. The Wind Turbine, inclusive of its supporting structure, shall be designed to withstand a wind speed of at least one hundred twenty (120) miles per hour.
- 10. Access. To prevent unauthorized climbing, tower access shall be restricted by either:
 - a. A secured fence and locked gate, both at least six (6) feet high, constructed around the perimeter of the base of the supporting structure, provided that such barrier is not required for any Wind Turbine erected on dwelling or other structure which provides no opportunity for climbing for at least six (6) feet; or
 - b. Removal of climbing apparatus on the support structure to at least ten (10) feet above the ground; or
 - c. Anti-climbing shrouds over the bottom portion of the structure.
- 11. Electrical Equipment. Electrical equipment shall be locked.
- 12. Building Code. Tower construction shall conform with the Massachusetts State Building Code as applicable.

- 13. Guy Wires. If the tower is to be supported by guy wires, fencing must be provided to prevent grazing animals from rubbing against the wires, as uneven tension on wires can make tower unstable.
- 14. *Electromagnetic Interference*. Electromagnetic interference with radio frequency communication, traceable to the operation or location of the Wind Turbine, shall be limited in accordance with all applicable sections of the specifications of the Federal Communications Commission.
- 15. Appearance; Color and Finish. Color and appearance shall comply with the safety requirements of the FAA.
- 16. Lighting. Wind Turbines shall be lighted only if required by the FAA. Lighting of other parts of Wind Energy Facilities, such as appurtenant Structures, shall be limited to that required for safety and operational purposes, and shall be reasonably shielded from abutting properties. Except as required by the FAA, lighting of Wind Energy Facilities shall be directed downward and shall incorporate full cut-off fixtures to reduce light pollution.
- 17. *Utility Connections*. Reasonable efforts, as determined by the Planning Board, shall be made to place all utility connections from Wind Energy Facilities underground, depending on appropriate soil conditions, shape and topography of the site and any requirements of the utility provider. Electrical transformers for utility interconnections may be above ground if required by the utility provider.
- 18. Appurtenant Structures. All appurtenant structures to Wind Energy Facilities shall be subject to reasonable regulations concerning the bulk and height of structures, lot area, setbacks, open space, parking and building coverage requirements. All such appurtenant structures, including but not limited to, equipment shelters, storage facilities, transformers, and substations, shall be architecturally compatible with each other and shall be contained within the turbine tower whenever technically and economically feasible. Whenever reasonable, structures should be shaded from view by vegetation and/or located in an underground vault and joined or clustered to avoid adverse visual impacts.
- 19. Land Clearing, Soil Erosion and Habitat Impacts. Clearing of natural vegetation shall be limited to that which is necessary for the construction operation and maintenance of the wind energy facility or otherwise prescribed by applicable laws, regulations and bylaws.
- 20. Monitoring and Maintenance. The applicant shall maintain the wind energy facility in good condition. Maintenance shall include, but not be limited to, painting, structural repairs, and integrity of security measures. Site access shall be maintained to a level acceptable to the local Fire Chief and local emergency services. The facility owner shall be responsible for the cost of maintaining Wind Energy Facilities and any access road(s).

7.3. Decommissioning. Decommissioning shall consist of:

- 1. Physical removal of all Wind Turbines, structures, equipment, security barriers and transmission lines from the site;
- 2. Disposal of all solid and hazardous waste in accordance with local, state, and federal waste disposal regulations;
- 3. Stabilization or re-vegetation of the site as necessary to minimize erosion. The Planning Board may allow the owner to leave landscaping or designated below-grade foundations in order to minimize erosion and disruption of vegetation;
- 4. Decommissioning shall occur after one (1) year of discontinued use without written consent from the permit granting authority. If the applicant fails to remove the facility in accordance with the requirements of this Section within one hundred fifty (150) days of abandonment or proposed date given by the applicant, the Town may enter the property and physically remove the facility at the applicant's cost through the financial surety set by the Town and applicant during site plan review or special permit process, as applicable.
- **7.3.1** Small Scale Wind Energy Facilities Submittal Requirements. The applicant shall submit a plan and supporting data, which shall bear the seal and signature of a registered professional engineer licensed to practice in Massachusetts. Submissions shall include a plan showing:
 - 1. Property lines of the applicant and abutters within three hundred (300) feet;
 - 2. Proposed location of the Wind Turbine;
 - 3. Location and uses (residence, garage, shed, etc.) of the proposed and abutters' buildings within three hundred (300) feet including distance from the proposed Wind Turbine to each of the buildings;
 - 4. A dimensional representation of the Wind Turbine mounted on its support structure, including foundation dimensions (both surface and depth), foundation materials, method of attaching tower to foundations, Wind Turbine dimensions including, tower height and rotor diameter, and clearance distances of blades to ground and nearest structure to be prepared by a Professional Engineer licensed to practice in the Commonwealth of Massachusetts;
 - 5. Small Scale Wind Energy Facility and Wind Turbines must be approved by a small wind certification program recognized by the American Wind Energy Association;

- 6. The Wind Turbine's design data including manufacturer's specifications and installation/operation instructions; and
- 7. Certification by the manufacturer or a registered engineer that tower design is sufficient to withstand wind load requirements for structures as established by the Massachusetts State Building Code.

7.3.1 Small Scale Wind Turbine Standards.

- 1. Setback. A Wind Turbine may not be sited within one and one-half times the height of the Wind Turbine from the nearest property line. A Wind Turbine may not be sited within a distance equal to the height of the Wind Turbine from critical infrastructure, or private or public ways.
- 2. Reduction. Setback distances of the Wind Turbine may be reduced by the SPGA based on site specific criteria and if the project is consistent with special permit granting criteria and only after review of substantial evidence, including but not limited to detailed engineering report or product engineering celification, which demonstrate that safety concerns have been minimized and that setbacks have been complied with to the maximum extent practicable. Setbacks for other than the Wind Turbine shall comply with this By-law.
- 3. *Property Boundaries*. No part of Small Scale Wind Energy Facilities support Structures, including guy wire anchors, may extend closer to the property boundaries than the standard structure setbacks for the zoning district in which the land is located.
- 4. *Height*. The height of a Wind Turbine shall be no greater than one hundred fifty (150) feet. The height of a Wind Turbine shall be measured from natural grade to the tip of the rotor blade at its highest point, or blade-tip height.
- 5. *Noise*. The operation of Small Scale Wind Energy Facilities shall conform to the provisions of the Department of Environmental Protection's, Division of Air Quality Noise Regulations at 310 CMR 7.10.
- 6. Restricted Access. To prevent unauthorized climbing, tower access shall be restricted by either:
 - a. A secured fence and locked gate, both at least six (6) feet high, constructed around the perimeter of the base of the supporting structure, provided that such barrier is not required for any Wind Turbine erected on a Dwelling or other Structure which provides no opportunity for climbing for at least six (6) feet; or
 - b. Removal of climbing apparatus on the support structure to at least ten (10) feet above the ground; or
 - c. Anti-climbing shrouds over the bottom portion of the structure.

- **7.3.12 Compliance with Other Regulations.** All applicants granted permits for Utility Scale and Small Scale Wind Energy Facilities pursuant to Section 7.3 of the By-law, shall comply with the following:
 - 1. Regulations of the Federal Communications Commission (FCC);
 - 2. Massachusetts State Building Code;
 - 3. Regulations of the FAA;
 - 4. National Electric Code; and
 - 5. Regulations of the Federal Energy Regulatory Commission (FERC).
- **7.3.13 Utility Notification.** All Wind Energy Facilities approved under Section 7.3 of the Bylaw shall be installed until evidence has been given that the utility company has been informed of the customer's intent to install an interconnected customer-owned generator and an interconnection agreement has been approved by the utility. Off-grid systems shall be exempt from this requirement.
- **7.3.14** Special Permit Approval Criteria. In addition to the criteria set forth in Section 10.5.2, a special permit may be granted for either a Small Scale or a Utility Scale Wind Facility as long as it is located outside of the COD pursuant to Section 7.3 of the By-law and shall meet the following standards:
 - 1. The specific site is an appropriate location for such use including but not limited to consideration of sound, flicker and visual impact;
 - 2. The use will not pose a significant adverse impact to the health and safety of the neighborhood;
 - 3. There will be no serious hazard to persons or vehicles from the use; and
 - 4. Adequate and appropriate infrastructure will be provided for the proper operation of the small scale wind energy conversion system.
- 7.3.15 As-Built Plan. No facilities approved under Section 7.3 of the By-law shall commence operation until an "As-Built" plan and a certificate of compliance have been submitted to the Planning Board and the Building Commissioner represents that the facility has been constructed substantially in compliance with the site plan approved by the Planning Board or the special permit issued by the Zoning Board of Appeals, as applicable. Said certificate and plan shall be signed and stamped by a registered professional engineer licensed in the Commonwealth of Massachusetts.

- **7.3.16 Abandonment.** Wind Energy Facilities shall be considered to be abandoned if they are inoperable for a period of one year, or considered hazardous by the Building Commissioner. Once Wind Energy Facilities are designated as abandoned or hazardous, the owner shall immediately physically remove the installation, which shall include, but not be limited to:
 - 1. Removal of the facility, any equipment shelters and security barriers from the subject property;
 - 2. Proper disposal of the waste materials from the site in accordance with local and state solid waste disposal regulations;
 - 3. Restoration of the subject property to its natural condition, except that any landscaping and grading may remain in the post-removal.
- 7A Wind Energy Facility which has reached the end of its useful life or has been abandoned shall be removed. The owner/operator shall physically remove the facility no more than one hundred fifty (150) days after the date of discontinued operations. The applicant shall notify the permit granting authority by certified mail of the proposed date of discontinued operations and plans for removal. Decommissioning shall comply with Section 7.3.11.
- **7.3.18 Utility Connections.** Reasonable efforts, as determined by the applicable Board, shall be made to place all utility connections from Wind Energy Facilities underground as depending on appropriate soil conditions, shape and topography of the site and any requirements of the utility provider.
- **7.3.19 Modification.** All material modifications to a Wind Energy Facility made after issuance of the required building permit shall require approval by the applicable Board.
- **20**All substantial modifications to Wind Energy Facilities made after issuance of the required building permit shall require site plan approval by the Planning Board or issuance of a special permit by the Zoning Board of Appeals, as applicable.
- **7.3. Priority.** The provisions set forth in this Section shall take priority over all other Sections of this By-law when considering applications related to the construction, operation, and/or repair of land-based Wind Energy Facilities.

SECTION 8.0 SPECIAL RESIDENTIAL REGULATIONS

- 8.1 OPEN SPACE AND FARMLAND PRESERVATION DEVELOPMENT.
- **8.1.1** Purpose. The purpose of Open Space and Farmland Preservation Development (OSFPD) is to:

- 1. Protect permanently open space, agricultural and forestry land, historical and archeological resources, existing and proposed trails, wildlife habitat and corridors, and other natural resources including wetlands and water bodies, in a manner consistent with the goals of the Hamilton Master Plan;
- 2. Encourage creative, environmentally sensitive design as the preferred form of residential development;
- 3. Encourage a more efficient form of development that consumes less open land and respects existing topography and natural features better than a conventional or grid subdivision;
- 4. Provide a variety of housing choices, particularly for older persons;
- 5. Reduce the anticipated negative fiscal impact on the Town associated with conventional residential development; and
- 6. Provide a development option to large-parcel landowners who wish to see a continuation of their stewardship of the land.
- **8.1.2** Applicability. In the R-la, R-lb, or R-A Districts, an applicant may obtain a special permit from the Planning Board for an OSFPD. An OSFPD is encouraged for developments which involve ten (10) or more acres or five (5) or more Dwelling Units but also is available for smaller developments. An OSFPD special permit allows the applicant a greater number of Lots or Dwelling Units on the site than allowed under a conventional subdivision in return for the applicant providing at least 50% of the total land area as permanently protected common open space as well as other benefits, all on terms and conditions further described in this Section 8.
- **8.1.3** Significance of Grant of Special Permit. Approval of an OSFPD special permit is a preliminary approval only. An applicant wishing to proceed with an OSFPD must then proceed with a definitive subdivision plan and/or a Cluster Design Plan, as described below. Approval of an OSFPD special permit does not indicate approval of the Board of Health, Conservation Commission, or any other entity from which the development requires separate permits or approvals. At any time during the OSFPD approval process the applicant maintains the option of withdrawing from the OSFPD process and proceeding with a conventional subdivision plan.
- **8.1.4 Future Subdivision.** The common open space and all Lots in an OSFPD may not be further subdivided, and a notation to this effect shall be placed on the plan of record, which shall be recorded at the Registry of Deeds.
- **8.1.5 Planning Board Regulations.** The Planning Board shall adopt and from time to time may amend OSFPD Rules and Regulations to implement this Section. Such regulations shall include but will not be limited to submission requirements; plan requirements, such as size, form, number and contents; development standards; site standards; and standards for Building

placement and design. Such rules and regulations are authorized under G.L. c. 40A, s. 9, and shall be adopted after vote by the Planning Board.

- 8.1.6 Permitted Uses. An OSFPD may include the following uses:
 - 1. Single Family Dwellings.
 - 2. Townhouse Dwellings, not to exceed four Dwelling Units per Building.
 - 3. Multi-family Dwellings, not to exceed six Dwelling Units per Building.
 - 4. Open space and conservation areas.
 - 5. Passive recreation, including trails for walking, hiking, cross country skiing, and horseback riding, and areas for other low-impact activities such as picnicking and wildlife observation.
 - 6. Agricultural, equestrian, and horticultural uses.
 - 7. Accessory recreational uses, such as a tennis court or playground in the developed area.
- 8.1.7 Pre-Application Conference. The applicant shall request and attend a pre-application review at a regular business meeting of the Planning Board. The Planning Board shall invite the Board of Health, Building Commissioner, Conservation Commission and Open Space Committee, Department of Public Works, Fire Chief, Police Chief, Historic District Commission, Office on Disability, and Zoning Board of Appeals to attend. The primary purpose of the pre-application conference is to identify the site's natural or historically important features, riding or walking trails on the site and abutting parcels, public safety, traffic or infrastructure issues, and areas the Town prefers to see preserved as open space or for agricultural or equestrian uses. The secondary purpose of a pre-application review is to minimize the applicant's cost of engineering and other technical experts, and to commence discussions with the Planning Board at the earliest possible stage of development. Meetings may be held by mutual agreement of the Planning Board and the applicant. At the pre-application review, the applicant may outline the proposed OSFPD, seek preliminary feedback from the Planning Board, and set a timetable for submittal of a formal application. At the expense of the applicant, the Planning Board may engage technical experts to review the informal plans of the applicant and to facilitate submittal of a formal application for an OSFPD special permit.
- **8.1.8 OSFPD Special Permit Application Procedures.** An application for a special permit for an OSFPD shall include a Yield Analysis and an OSFPD Concept Plan. The Planning Board may engage technical experts, at the applicants' expense, as reasonably necessary in connection with its review of the applicant's proposed analysis or plan(s).
- **8.1.9 Sources of Data.** Both the Yield Analysis and the OSFPD Concept Plan may be prepared from existing data, such as deed information, USGS topographical maps, FEMA floodplain

maps, assessor's maps, photographs, soil maps, or Department of Environmental Protection (DEP) Wetlands Conservancy Program maps. The locations of wetlands, streams and forest limits or locations will not be verified during the Concept Plan special permit process until the applicant has delineated the same in the field and has obtained from the Hamilton Conservation Commission a positive determination verifying delineation, or the applicant has been issued an Order of Resource Area Delineation under the procedures outlined in G.L. c. 131 s. 40 and Chapter 17 of the Town of Hamilton By-laws. It is not necessary to verify these constraints for Concept Plan submittals, but the applicant must be aware that these locations should be as accurate as possible in order to avoid significant changes to the Concept Plan in subsequent applications for approval of an OSFPD definitive subdivision plan or Cluster Design Plan.

- **8.1.10** Allowed OSFPD Density. The Allowed OSFPD Density is the maximum number of Lots (or Dwelling Units) in an OSFPD which shall be one and one fifth (1.2) times the Base Maximum Density, as may be further increased by the Planning Board pursuant to Section 8.1.21 based on additional public benefits being provided by the OSFPD. Computations shall be rounded to the nearest whole number.
- **8.1.11 OSFPD Concept Plan.** The OSFPD Concept Plan shall be prepared by a Registered Professional Landscape Architect, Civil Engineer, or Architect, or by a multi-disciplinary team of which one (1) member must be a Registered Professional Landscape Architect, and shall include the information listed below. The OSFPD Concept Plan shall be produced through the five-step OSFPD special permit design process and shall incorporate the common open space requirement, dimensional standards, and OSFPD special permit design standards all as set forth in this Section.
 - 1. Title Block including the name of the Owner of record, name of applicant, address of the property, and the Assessors' Map and Lot Number; name of the company preparing the plan, address and phone number, signature and stamp of professional(s) preparing the plan, date of plan, scale;
 - 2. The location of the proposed development;
 - 3. The size of the site in acres;
 - 4. An existing conditions inventory and description of conservation areas identified during the OSFPD special permit design process pursuant to this Section;
 - 5. The total number and approximate locations of the proposed Buildings, Dwelling Units and/or Lots, and the approximate size of each in square feet;
 - 6. The acreage and proposed use(s) of permanently protected common open space;
 - 7. A statement on the disposition or manner of ownership of the proposed open space;

- 8. The areas or approximate delineation of Lots that will be used as building areas, and the areas or approximate delineation of Lots that are to remain as permanently protected common open space;
- 9. A general description of how drainage will be handled, including a soils statement (soil conservation survey is acceptable) and the general area of the site to be used for stormwater management facilities; and
- 10. Sufficient detail of proposed built and natural features as described in this Section to enable the Planning Board to make the required determinations.

8.1.12 OSFRD Special Permit Design Process.

- 1. Identification of Conservation Areas. The first step in the design process requires identification of conservation areas on the site, including wetlands, areas located in the Conservancy District, riverfront areas, and floodplains regulated by state, federal, or local law; unprotected natural landscape features such as steep slopes, mature woodlands, prime farmland, meadows, wildlife habitats and wildlife corridors or connections thereto; cultural features such as historic and archeological sites and scenic views; and recreational features such as established trails used for horseback riding, walking and cross-country skiing. To the maximum extent feasible, conservation areas shall include areas identified by the Planning Board during the pre-application conference.
- 2. Identification and Delineation of the Proposed Development Area. The second step in the design process is to define and delineate the area that will contain buildings, roadways, other site improvements and amenities for residents of the development. To the maximum extent feasible, the proposed development area shall consist of land outside the identified conservation areas.
- 3. Location of Dwelling Units. The third step in the design process is to identify and delineate the approximate location of Dwelling Units in the proposed development area. The number of Dwelling Units should conform to the Allowed OSFPD Density calculated pursuant to Sections 8.1.10 and 8.1.21. The location of Dwelling Units should account for proximity to common open space and other amenities, including community buildings for use by residents of the development. Toward this end, the number of Dwelling Units with direct access to the amenities of the development should be maximized.
- 4. Roads and Trails. The fourth step in the design process is to identify and delineate the approximate location of roads and trails. Roads shall be aligned to access the Dwelling Units. The layout of new trails should anticipate internal and external connections to existing and/or potential future roads, trails and sidewalks.
- 5. Lotting. If applicable, the final step in the design process is to identify the approximate location of Lot lines.

- **8.1.13 Common Open Space Requirement.** To qualify for an OSFPD special permit, a development must provide at least fifty (50) percent of the total land area as permanently protected, usable, common open space that is functional for purposes intended by this Section. The common open space shall have no Structures, parking, private Yards, patios, or gardens that are restricted for the exclusive or principal use by residents of individual Dwelling Units. The following standards apply to the common open space in an OSFPD:
 - 1. Use, Shape, Location of Common Open Space. To the maximum extent feasible, the common open space shall be undisturbed, unaltered and left in its natural condition or existing condition. It shall be appropriate in size, shape, dimension, location, and character to assure its use as a conservation area, or where appropriate, a recreational area, and serve as a visual and natural amenity for the development and the Town;
 - 2. *Contiguous*. The common open space shall be Contiguous Open Space and linked as a unit, with links at least seventy-five (75) feet wide, unless waived by the Planning Board;
 - 3. Common open space shall be functional for wildlife habitat, passive recreation, resource preservation, agriculture, or equestrian uses;
 - 4. The location(s) and configuration of the common open space shall be subject to approval by the Planning Board;
 - 5. Each parcel of common open space shall offer adequate access to residents of the OSFPD;
 - 6. Land used for common or shared septic systems may not be counted toward the minimum common open space requirement unless authorized by the Planning Board;
 - 7. Not more than fifty (50) percent of the common open space in an OSFPD shall consist of areas subject to the Wetlands Protection Act, G.L. c. 131, s. 40, for reasons other than being subject to flooding, or land in the Conservancy District.
 - 8. Existing utility easements may not be counted as common open space; and
 - 9. Up to five (5) percent of the minimum required open space may be used for gravel roadways, pavement or structures accessory to the dedicated use or uses of the common open space. Principal or accessory structures and access roads essential to an agricultural use are exempt from this requirement, except for indoor/covered riding rings.
- 8.1.14 Ownership of the Common Open Space. Any common open space within an OSFPD shall be conveyed to the Town and accepted for park or open space use, and/or shall be conveyed to a nonprofit organization the principal purpose of which is the conservation of open space, and/or shall be conveyed to a corporation or trust owned or to be owned by the owners of Lots or residential units within the plan, as provided by G.L. c. 40A, s. 9. In any case where the common open space is not conveyed to the Town, a restriction enforceable by the Town or the

Conservation Commission shall be recorded providing that such land shall be kept in an open or natural state and not be built for residential use or developed for accessory uses such as parking or roadway.

- **8.1.15 Dimensional Standards.** To maintain harmony with surrounding areas, existing dimensional requirements of this By-law shall apply to any lot located along the perimeter of an OSFPD or abutting any existing residential development. In addition, any OSFPD lot that relies on an existing public way for frontage shall conform to the applicable Lot Frontage and Front Yard setback requirements. For interior lots in an OSFPD, the Planning Board may waive the requirements for minimum lot area, setbacks, building coverage, frontage, number of Dwelling Units per Lot, shared driveways or irregular Lot shape that would normally apply in the zoning district in order to maximize the amount of open space, reduce site disturbance and protect significant farmland or scenic landscapes. The following additional requirements apply within the OSFPD:
 - 1. The minimum distance between clusters of any Townhouse Dwellings shall be fifty (50) feet unless waived by the Planning Board to further the purposes of this Section.
 - 2. At least fifty (50) percent of the required Yard setbacks shall be maintained on interior lots in the OSFPD unless a reduction is authorized by the Planning Board to accommodate "zero lot line" design.
- **8.1.16 OSFPD Special Permit General Design Standards.** The following general design standards shall apply to all OSFPDs and shall govern the development and design process:
 - 1. The landscape shall be preserved in its natural state, insofar as practicable, by minimizing tree and soil removal. The grade changes shall be in keeping with the general appearance of the neighboring developed areas. The orientation of individual building sites shall be such as to maintain maximum natural topography and cover. Topography, tree cover, and natural drainage ways shall be treated as fixed determinants of road and lot configuration rather than as malleable elements that can be changed to follow a preferred development scheme;
 - 2. Streets shall be designed and located in such a manner as to maintain and preserve natural topography, significant landmarks and trees, to minimize cut and fill, and to preserve and enhance views and vistas on or off the subject property;
 - 3. Building designs shall relate harmoniously to the terrain and the use, scale, and architecture of existing buildings in the vicinity that have functional or visual relationship to the proposed buildings;
 - 4. All open space shall be designed to add to the visual amenities of the area by maximizing its visibility for persons passing the site or overlooking it from nearby properties; and

- 5. The removal or disruption of historic, traditional or significant uses, structures, or architectural elements shall be minimized insofar as practicable, whether these exist on the site or any adjacent properties.
- **8.1.17 OSFPD Special Permit Site Specific Design Standards.** The following site specific design standards shall apply to all OSFPDs and shall govern the development and design process:
 - 1. *Mix of Housing Types*. The OSFPD may consist of a combination of Single Family Dwelling, Two Family Dwelling, and other Multifamily Dwelling. Two-bedroom Dwelling Units are encouraged to be designated as restricted to Elderly Persons;
 - 2. Parking. Each Dwelling Unit shall be served by two (2) off-street parking spaces. Parking spaces in front of garages may count in the computation. All parking areas with greater than four spaces shall be screened from view from the road. Residential structures should be oriented toward the street serving the premises and not the required parking area;
 - 3. Buffer Areas. A vegetated buffer area of twenty-five (25) feet minimum shall be provided at the following locations:
 - a. Perimeter of the property where it abuts residentially zoned and occupied properties;
 - b. Certain resource areas on or adjacent to the tract such as rock outcrops, ledge, agricultural or recreational fields, and land held for conservation purposes;
 - c. A landscape plan will be required for buffers to ensure screening from abutting properties. Additions to existing plantings may be required by the Planning Board. The Planning Board may waive the buffer requirement when it determines that a small buffer (or no buffer) will suffice to accomplish the objective set forth herein; and
 - d. Driveways necessary for access and egress to and from the tract may cross such buffer areas.
 - 4. Stormwater Runoff and Drainage. The peak rate of stormwater runoff and drainage design shall comply with the DEP Stormwater Management Policy. All structural surface stormwater management facilities shall be accompanied by a conceptual screening and landscape plan. The Planning Board shall encourage low impact development practices such as the use of "soft" (non-structural) natural stormwater management techniques (such as open swales) and other drainage techniques that do not create impervious surfaces and that enable infiltration where appropriate. Water conservation measures, including but not limited to the use of rainwater retention systems, such as rain barrels and cisterns for water irrigation purposes, are also strongly encouraged;

- 5. Common/Shared Driveway. A common or shared Driveway may serve a maximum of three Dwelling Units unless otherwise approved by the Planning Board, with input from public safety officials; and
- 6. On-site Pedestrian and Bicycle Circulation. Walkways and bicycle paths shall be provided to link residences with parking areas, recreation facilities (including parkland and open space) and adjacent land uses where appropriate.

8.1.18 Review Procedures for OSFPD Special Permit.

- 1. The applicant shall furnish a copy of the application to the Town Clerk. The applicant shall furnish sufficient copies of the special permit application to the Planning Board to distribute for review to the Board of Health, Board of Selectmen, Building Commissioner, Conservation Commission and Open Space Committee, Department of Public Works, Fire Chief, Police Chief, Historic District Commission, Office on Disability, and Zoning Board of Appeals.
- 2. Reports from the above-named boards and officials shall be submitted to the Planning Board within thirty-five (35) days of receipt by the reviewing party. In the event that the public hearing by the Planning Board is held prior to the expiration of the thirty-five (35) day period, the Planning Board shall continue the public hearing to permit the formal submission of reports and recommendations within that 35 day period. Public hearing procedures shall comply with the requirements of G.L. c. 40A, s. 9.
- 3. Planning Board members shall conduct a site visit with the applicant prior to or during the public hearing.
- 4. An OSFPD special permit may be issued only following a public hearing held within sixty-five (65) days after filing an application with the Planning Board.
- 5. Once an OSFPD special permit application has been submitted, no tree removal, no utility installation, no ditching, grading or construction of roads, no grading of land or lots, no excavation, except for purposes of soil testing, no dredging or filling and no construction of buildings or structures shall be done on any part of the site until the application (a) has been reviewed and decided as provided by these regulations, or (b) has been withdrawn and the applicant has obtained other permits for such work.
- **8.1.19** Decision of the Planning Board for OSFPD Special Permit. The Planning Board shall take one of the following actions within ninety (90) days following the date of the public hearing unless extended by written agreement between the Planning Board and the applicant in accordance with G.L. c. 40A, s. 9.
 - 1. The Planning Board may grant an OSFPD special permit with any conditions, safeguards, and limitations, considering each of the following criteria in addition to those in Section 10.5.2:

- a. The degree to which the conceptual design and layout of the proposed OSFPD is superior to a conventional development in preserving open space for conservation and recreation, preserving natural features of the land, achieving more efficient provision of streets, utilities and other public services, and providing a high degree of design quality;
- b. The degree to which the OSFPD promotes permanent preservation of open space, agricultural land, forestry land, historical and archeological resources, existing and proposed trails, wildlife habitat and corridors, and other natural resources including wetlands and water bodies;
- c. The degree to which the OSFPD achieves sustainable design through a more efficient form of development that consumes less open land and conforms to existing topography and natural features better than a conventional subdivision; and
- d. The degree to which the OSFPD furthers the goals and policies of the Master Plan and the purposes of this Section.
- 2. The Planning Board may deny an OSFPD special permit upon finding that the application does not comply with the provisions of this Section.
- **8.1.20** Effect of OSFPD Special Permit Approval. Approval of an OSFPD special permit under this Section shall not be considered approval of any construction. This approval is a preliminary approval, intended to give guidance to the applicant for the development of an OSFPD definitive subdivision plan or a Cluster Design Plan, and to determine whether the applicant's submittal meets the objectives of this Section. Approval of an OSFPD special permit does not indicate approval of the Board of Health, Conservation Commission, or any other entity from which the development requires separate permits or approvals.
- **8.1.21 Public Benefit Incentives.** In approving an OSFPD special permit, the Planning Board may authorize an increase in the number of Dwelling Units beyond 1.2 times the Base Maximum Density. Such increase may be approved only for an OSFPD that provides one (1) or both of the following public benefits:
 - 1. Additional Common Open Space. For each additional ten (10) percent of the site (over and above the required fifty (50) percent) set aside as common open space, including preserving and providing public access to existing walking or riding trails or for providing connections to existing trails on abutting parcels, an increase of ten (10) percent of the Base Maximum Density may be approved; provided, however, that the total number of additional units shall not exceed twenty-five (25) of the Base Maximum Density. Computations shall be rounded to the nearest whole number.
 - 2. Housing for Older Persons. For every one (1) Dwelling Unit restricted to occupancy by Elderly Persons, the Planning Board may approve an increase of one (1) additional

Dwelling Unit; provided, however, that the total number of additional Dwelling Units approved shall not exceed twenty-five (25) percent of the Base Maximum Density. Computations shall be rounded to the nearest whole number.

3. Limited Public Access to Common Open Space. An increase of ten (10) percent of the Base Maximum Density may be approved by the Planning Board for limited public access to the common open space for passive recreation.

In no event shall the Planning Board approve an increase of more than fifty (50) percent of the Base Maximum Density.

- **8.1.22** Relationship Between the OSFPD Special Permit and OSFPD Definitive Subdivision Plan. An OSFPD that involves a subdivision of land may be submitted to the Planning Board for approval under the Subdivision Control Law following the issuance of an OSFPD special permit. Planning Board approval of an OSFPD special permit shall neither oblige the Planning Board to approve any related definitive plan nor substitute for such approval. The OFSPD definitive subdivision plan shall substantially comply with the OSFPD Concept Plan.
- **8.1.23** Relationship Between the OSFPD Special Permit and Cluster Design Plan Approval. The issuance of an OSFPD special permit allows the applicant to submit a Cluster Design Plan to the Planning Board for review and approval. A Cluster Design Plan shall be considered neither a subdivision plan under the Subdivision Control Law nor a site plan subject to the provisions of Section 10.6. No Cluster Design Plan may be filed unless an OSFPD special permit has been approved and has not lapsed.
- **8.1.24** Cluster Design Plan Submittal Requirements. The applicant shall submit an application for Cluster Design Plan approval to the Planning Board, with a copy filed with the Town Clerk. The size, form, number and contents of the Cluster Design Plan application and plan requirements shall be set forth in the Planning Board's OSFPD Rules and Regulations.
 - 1. The Cluster Design Plan shall be designed to conform to the approved OSFPD special permit.
 - 2. Cluster Design Plan application filing fees shall be governed and set by the Planning Board in OSFPD Rules and Regulations, and shall be assessed to the owner and/or the applicant. Separate technical review fees as provided for in state statute may be assessed for engineering, architectural or other review by one or more consultants selected by the Planning Board.
- **8.1.25** Cluster Design Plan Procedures. Cluster Design Plan approval may be issued by the Planning Board only following a public hearing held within sixty-five (65) days after filing an application with the Planning Board.
 - 1. The applicant shall furnish sufficient copies of a complete Cluster Design Plan application for the Planning Board to distribute to the Board of Health, Board of Selectmen, Building Commissioner, Conservation Commission and Open Space

Committee, Department of Public Works, Fire Chief, Police Chief, Historic District Commission, Office on Disability, and Zoning Board of Appeals.

- 2. Notice of the time, place and subject matter of the public hearing shall be in accordance with the provisions of G.L. c. 40A, s. 11. Legal notice mailed to abutters shall be sent by the Planning Board by certified mail at the applicant's expense. Reports from other boards and officials shall be submitted to the Planning Board within thirty (30) days of receipt by the reviewing party. In the event that the public hearing by the Planning Board is held prior to the expiration of the thirty (30) day period, the Planning Board shall continue the public hearing to permit the formal submission of reports and recommendations within that thirty (30) day period.
- 3. The Planning Board shall determine whether the Cluster Design Plan complies with the requirements of Section 8 within sixty-five (65) days of the public hearing. If no action is taken within sixty-five (65) days, the application shall be deemed approved as submitted.
- 4. Within fifteen (15) days of its vote on the Cluster Design Plan, a copy of the Cluster Design Plan decision shall be filed with the Town Clerk and the Building Commissioner, with a copy being sent by certified mail or hand delivered to the applicant. Any interested party aggrieved by the Cluster Design Plan decision may file an appeal under the provisions of G.L. c. 40A, s. 17.
- 5. The Building Commissioner shall not approve any building permit application subject to these provisions without receipt of Planning Board approval and expiration of the appeal period, as certified by the Town Clerk.
- **8.1.26** Cluster Design Plan Decision. The Planning Board shall take one of the following actions within sixty-five (65) days following the date of the public hearing:
 - 1. The Planning Board shall approve a Cluster Design Plan only upon a determination that the plan substantially complies with the OSFPD Special Permit and satisfactorily addresses all of the following criteria:
 - a. Adequate access to each structure for public safety equipment and personnel;
 - b. Adequate utility service and drainage, consistent where appropriate with the Hamilton Subdivision Regulations in effect at the time of the submission of the site plan;
 - c. Adequate capacity on impacted streets to accommodate the proposed project, based on reports received from the Police Department and the Department of Public Works or technical review consultants retained by the Planning Board;
 - d. Adequate measures to reduce the volume of cut or fill, soil erosion, and visual intrusion of parking areas viewed from public ways or abutting properties;

- e. Protection of pedestrian and vehicular safety within the site and egressing from it;
- f. Compliance with conceptual landscape, building design, and placement of Buildings;
- g. Consistency with the Planning Board's OSFPD Rules and Regulations; and
- h. Compliance with all the other requirements of this By-law.
- 2. The Planning Board may determine that the Cluster Design Plan does not to comply substantially with the OSFPD special permit based on any of the following conditions:
 - a. An increase in the number of Buildings or Dwelling Units;
 - b. A significant decrease in the open space acreage;
 - c. A significant change in the general development pattern which adversely affects natural landscape features and open space preservation;
 - d. Significant changes to the storm water management facilities; and/or,
 - e. Significant changes in the wastewater management systems.
- 3. The Planning Board may conditionally approve a Cluster Design Plan that does not substantially comply with the OSFPD special permit. However, such conditional approval must identify where the plan does not substantially comply with the special permit for the OSFPD Concept Plan and shall require that the special permit be amended to be in compliance with the significant changes identified by the Planning Board.
- 4. The Planning Board shall also require that the applicant file an application to amend the OSFPD special permit within a specified time period. The public hearing on the application to amend the OSFPD special permit shall be limited to the significant changes identified by the Planning Board in their conditional approval of the Cluster Design Plan. These are the only considerations that the Planning Board may take into account in deciding whether to amend the OSFPD special permit for the Concept Plan.

8.2 SENIOR HOUSING.

- **8.2.1 Purpose.** The purpose of this Section is:
 - 1. To permit the development of moderately priced, affordable, and market rate housing for Elderly Persons, by allowing for a greater variety of Building types at a higher density than would normally be allowed;

- 2. To allow greater flexibility in land use planning in order to improve site layouts, protect natural features and environmental values and utilize land in harmony with neighboring properties;
- 3. To encourage the implementation of "Smart Growth" techniques to reduce land consumption and sprawl, provide for open space preservation, expand housing options, and encourage re-use of existing Structures;
- 4. To allow a type of development which has positive fiscal benefits to the Town; and
- 5. To protect Hamilton's rural New England character by permitting development of residential housing in clusters and villages, in a manner which is in harmony with Hamilton's historic development patterns and is less demanding on its natural resources.
- **8.2.2 Eligible Locations.** A special permit may be granted for Senior Housing on any Lot(s) in the R- la, R-1b, R- A, or B Districts that meet the requirements of this Section and other applicable provisions of this By-law. Parcels in existence as of May 5, 2008, may not be assembled to create a larger parcel for Senior Housing development.
- **8.2.3** Administration. The Planning Board shall be the SPGA for this Section. An applicant may file an application with the Planning Board in conformance with this Section and the Planning Board Rules and Regulations.
- **8.2.4 Planning Board Rules and Regulations.** The Planning Board shall adopt and from time to time may amend Senior Housing Special Permit Rules and Regulations to implement this Section. Such regulations shall include but will not be limited to submission requirements, fees, plan requirements (such as size, form, number and contents), development standards, site standards, and standards for Building placement and design. Such rules and regulations are required and authorized under G.L. c. 40A, s. 9, and shall be adopted after vote by the Planning Board.
- **8.2.5** Age Restriction. Each Dwelling Unit in a Senior Housing development shall be owned and occupied by at least one (1) Elderly Person. All other residents must be over the age of eighteen (18) years of age.
- **8.2.6** Limitations on Senior Housing Development. The following shall apply:
 - 1. A maximum of one hundred (100) Senior Housing Dwelling Units may be permitted Town-wide under this Section.
 - 2. A maximum of fifty (50) Dwelling Units per project shall be allowed.
 - 3. The maximum number of Dwelling Units permitted in any year shall be fifty (50). For the purpose of this Section, a "year" shall mean a calendar year. Any partial year shall be pro-rated.

- 4. Dwelling Units shall be counted in the calendar year in which the special permit is granted, not the year in which the application was filed.
- 5. If fewer than the maximum number of Dwelling Units is permitted in a year, the balance shall be carried forward to the next year. However, the maximum number of Dwelling Units permitted per year shall remain at fifty (50).
- 6. In the event that substantial use or construction under a special permit issued has not commenced within two (2) years of approval and no extensions have been granted, the number of Dwelling Units shall be added back into the number of unpermitted Dwelling Units, and shall be available to be permitted under this Section.
- 8.2.7 Permitted Uses. A Senior Housing development may include the following uses:
 - 1. Single Family Dwellings;
 - 2. Two Family Dwellings;
 - 3. Semi-attached Dwellings;
 - 4. Three Family Dwellings;
 - 5. Four Family Dwellings:
 - 6. Townhouse Dwellings;
 - 7. Multi-family Dwellings;
 - 8. Open space and conservation areas;
 - 9. Passive recreation:
 - 10. Agricultural, equestrian, and horticultural uses:
 - 11. Accessory Uses, such as a tennis court, pool, community building, and other amenities, in the developed area, for the benefit of the development;
 - 12. Recreational fields; and
 - 13. In the Business District, a mixture of residential and business uses.
- **8.2.8 Developable Acres.** "Developable Acres" is defined as the area of the tract of land, exclusive of the area of land lying in the Conservancy District and the area in wetlands subject to G.L. c. 131, s. 40, as mapped by the applicant's consultant and approved by the Hamilton Conservation Commission. In addition, land subject to a preexisting Conservation Restriction,

except for land area in a defined "building envelope", shall be excluded from the total acreage. This net land area shall constitute "Developable Acres".

- **8.2.9** Base Senior Housing Density. The Base Density per Developable Acre for a Senior Housing special permit shall be: four (4) Dwelling Units in the Business District; three (3) Dwelling Units in the RI-a District, two (2) Dwelling Units in the RI-b District, and one (1) Dwelling Unit in the RA District.
- **8.2.10** Incentives to Increase Base Senior Housing Density. See the following matrix for incentives which may be utilized to increase the Base Senior Housing Density. The final number shall be rounded to the nearest whole number.

	A	В	C	D	E	F	G	Н	I	J
			Incentives to Increase Base Senior Housing Density for:				enior			
	Zoning District	Base Senior Housing Density (BSHD) per Dev. Acre	Mix of Smaller Units (50% units under 1300 sq. ft.)	On or Off Site Construction of Inclusionary Housing Obligation (Affordable Housing)	Smart Growth, per Comm. of Mass. DHCD Criteria (PB Rules & Regs)	Additional 25% of Property as Open Space	Voluntary Cap on Appreciation of Units	Subtotal of Base Senior Housing Density Incentives (C+D+E+ F+G)	Incremental Density per Developable Acre Rewardable by PB for Trails, Fields, Environmental, energy, water conservation, building green, & other public benefits	Maximum Density per Dev. Acre
1	В	4	25 %	25%	25%	N/A	25%	100%	1.0	6.00
2	R-1a	3	25 %	25%	25%	25%	25%	125%	1.0	4.00
3	R-1b	2	25 %	25%	25%	25%	25%	125%	1.0	4.00
4	RA	1	25 %	25%	25%	50%	25%	150%	2.0	4.00

NOTES ON HOW TO READ THE MATRIX:

The Incentives to Increase Base Senior Housing Density may be selected and utilized by an applicant to increase the Base Senior Housing Density shown in Column B. The percentage shown in Columns C, D, E, F, G, and Column I would, if selected, increase the number of Dwelling Units per Developable Acre in Column B, up to the amount shown in Column J, Maximum Density per Developable Acre.

EXPLANATION OF COLUMNS:

Column A: The existing Zoning Districts in Hamilton.

Column B: The Base Senior Housing Density (BSHD) per Developable Acre. This is the base acreage that one may apply for in a Senior Housing special permit, without utilizing any of the Incentives in Columns C, D, E, F, G, and/or I.

- **Column C:** Encourages the construction of smaller-sized Dwelling Units. For meeting this requirement, an applicant earns twenty-five (25) percent more Dwelling Units than what is shown in Column B.
- Column D: Encourages construction of affordable Dwelling Units. Section 8.3, Inclusionary Housing, requires an affordable housing component for projects of ten (10) or more Dwelling Units. This column requires that the housing be constructed on-or off-site, rather than meeting Inclusionary Housing requirements by other authorized methods. For meeting this requirement, an applicant earns twenty-five (25) percent more Dwelling Units than what is shown in Column B.
- **Column E:** Encourages the utilization of "Smart Growth" or "Sustainable Development Principles", some of which are listed in part in Section 8.2.1. See Planning Board Rules and Regulations for Smart Growth Principles issued by Commonwealth of Massachusetts, Office for Commonwealth Development. For meeting this requirement, an applicant earns twenty-five (25) percent more Dwelling Units than what is shown in Column B.
- **Column F:** Encourages the preservation of open space above what is required by this Section. For meeting this requirement, an applicant earns twenty-five (25) percent more Dwelling Units than what is shown in Column B.
- Column G: Encourages resale prices which will become more affordable for subsequent purchasers of the Senior Housing Dwelling Units due to a voluntary cap on the allowable rate of appreciation on sales in future years. A deed restriction (to be included in association documents) imposed by the applicant, places a cap on appreciation of the resale price of Dwelling Units. For meeting this requirement, an applicant earns twenty-five (25) percent more Dwelling Units than what is shown in Column B.
- **Column H:** Represents a sub-total of the incentives in Columns C G.
- Column I: Encourages other initiatives which provide a public benefit to the Town such as, but not limited to, preservation and enhancement of trails; creation of playing fields; environmental, energy, water conservation, green building design, and other public benefits. For one (1) or a combination of these initiatives, an applicant may earn one (1) additional Dwelling Unit per Developable Acre in the Business, R1-a, and Rl-b Districts, and up to two (2) additional Dwelling Units per Developable Acre in the RA District.
- **Column J:** Indicates the maximum density per Developable Acre which one may earn by using a variety of the incentives allowed. Incentives shall be allowed at the discretion of the Planning Board. The final number shall be rounded to the nearest whole number.
- **8.2.11 Minimum Distance Between Projects.** In order to mitigate the impact of multiple developments on surrounding neighborhoods, a proposed project must be located at least the distance established in the Minimum Distance Between Projects Table below from any other existing or approved Senior Housing project, any existing or approved project including multi-unit residential buildings under G.L. c. 40B, or any existing or approved project including

multi-unit residential buildings under any other provision or By-law. Excluded from this calculation and this Section are any existing multi-family projects approved prior to January 1, 2003, and apartments under Sections 3.4, 3.6 or 3.7.

The minimum distance between projects shall be calculated using the following Table:

Combined Total Dwelling Units in Proposed Project and in Existing or Approved Projects	Minimum Distance between Projects			
Up to and including 50 total Dwelling Units	0.33 mile			
Exceeds 50 total Dwelling Units	0.66 mile			
Exceeds 75 total Dwelling Units	1.0 mile			

- 1. The distance between projects shall be measured from the closest property boundary. Parcels in existence on May 5, 2008 may not be subdivided to avoid application of this provision. A project shall be deemed to be existing or approved if it has been constructed or if a special or comprehensive permit, or a site plan approval, has been granted; or if a special or overlay zoning district which allows multi-family housing has been approved.
- 2. The determination of the required separation between projects shall be made at the time the special permit is issued under this By-law, in order to assess the most up to date existing project status.
- 3. If a Chapter 40B housing proposal is filed and approved while an application for Senior Housing special permit has been filed and is under review by the Planning Board, that Senior Housing project shall be exempt from the distance requirement of this Section.
- 4. Senior Housing proposals in the Business District shall be exempt from the provisions of this Section and developments wholly within the Business District shall not be considered as existing or approved projects for purposes of establishing required project separation in any other district.
- **8.2.12** General Compliance. A proposal must comply with all other applicable Town By-laws, and the applicable rules, regulations, and requirements of all departments, boards, and commissions, including the special sensitivities of the Historic District and the Groundwater Protection Overlay District.
 - 1. Extension or Alteration of Pre-Existing Nonconforming Use. To the extent that such a finding is required, that approval shall be granted before applying for a Senior Housing special permit.

- 2. *Inclusionary Housing*. There shall be compliance with Section 8.3, Inclusionary Housing. The Town of Hamilton Local Preference Policy shall apply to affordable Dwelling Units created.
- 3. Local Preference. The applicant is encouraged to provide a local preference program for fair market rate Dwelling Units, for an initial limited duration, for the purchase of Dwellings Units by those eligible under the Local Preference Policy.

8.2.13 General Requirements. The following requirements shall apply.

- 1. Low Impact Development. The use of low-impact development techniques is required, where applicable. The applicant shall employ meaningful low impact techniques which will result in less impervious area, direction of roof runoff toward rain gardens and swales, and plantings indigenous to the area. The use of recycled or recaptured rainwater is encouraged. A Low Impact Development Handbook and other references are available from the Planning Board Office.
- 2. *Minimizing Disturbance*. The applicant is encouraged to maintain as much of the site as possible in its natural state. The applicant is urged to incorporate horticultural and landscape design that reduces the need for supplemental irrigation and chemical fertilization, e.g. minimizing lawn area.
- 3. Location of Wastewater Treatment. All wastewater shall be treated and discharged on-site. Off-site wastewater treatment is not allowed. An exception to this is for parcels located in the Business District, where wastewater treatment may be located off-site.
- 4. Paths and Trails. All existing paths and trails shall be connected, preserved, and increased when possible.
- 5. Conversion to Apartments. Dwelling Units constructed by special permit under this Section shall not be eligible for subsequent conversion to add apartments under Sections 3.4, 3.5, 3.6, or 3.7.
- 6. Reuse of Existing Buildings. Existing Buildings and Dwelling Units may be reused and rehabilitated for the purposes of this Section provided all standards and requirements are met.
- 7. Homeowners Association and/or Condominium Documents. Prior to submitting an application to the Building Commissioner and prior to construction of infrastructure, the applicant shall submit to the Planning Board and receive approval of all homeowner's association and/or condominium documents. The Planning Board or legal counsel shall review and approve the documents to ensure that the intent of this Section and the relevant sections of the decision of the Planning Board are incorporated into the documents.

- **8.2.14 Exemptions.** A proposal for a Senior Housing special permit shall be exempt from Section 10.6, Site Plan Review.
- **8.2.15** Dimensional Standards. The following dimensional and other standards shall supersede all dimensional standards in the underlying zoning district and shall be applied to any Senior Housing development:

TABLE OF DIMENSIONAL REQUIREMENTS

	A	В	C	D	E	F	G	Н	I
	Zoning District	Under- lying Zoning District Require- ment (s.f.)	Minimum Parcel Size (sq.ft.) to be eligible for develop- ment	Minimum Parcel Frontage Required (feet) to be eligible for develop- ment	Front Setback (feet) from perimeter property line	Side & Rear Setbacks (feet) from perimeter property line	Vegetated Buffer (feet) around perimeter of parcel	Minimum Open Space Required of total acreage	Maximum Lot Coverage of entire parcel
1	Bus. District		none	site specific	0	01	0^1	0	site specific
2	R-1a	20,000	80,000	125	25	25 ²	20	0	25%
3	R-1b	40,000	80,000	175	25	25 ²	20	15%	25%
4	RA	80,000	80,000	175	50	25 ²	20	25%	25%

- Note 1. The Planning Board shall require a setback and a vegetated buffer where the parcel abuts a residential zone.
- Note 2. Minimum of twenty-five (25) feet or the height of the Building, whichever is greater.
- **8.2.16** Building and Design Standards. In order to achieve a development that reflects the residential character of the neighborhood in which it is located, the following minimum building and design standards shall be applied to all Structures constructed in a Senior Housing development:
 - 1. Dwelling Units shall not contain more than two (2) bedrooms.
 - 2. Buildings shall be designed to be consistent with the residential character of the Town and shall be complementary in exterior design with each other and, where applicable, with the existing neighborhood in which the development is located.
 - 3. A mixture of small, medium, and large Dwelling Units is encouraged, in order to create a range of selling price levels, and to address the needs of various members of the population.

- 4. A minimum of twenty-five (25) foot separation between Buildings shall be provided and shall be landscaped.
- 5. Accessory structures shall comply with all setback requirements and shall be designed with architectural detailing of similar nature to the principal Buildings located thereon.
- 6. Other Building and Design Standards in applicable Planning Board Rules and Regulations shall apply.
- **8.2.17** Lighting. All lighting fixtures shall be integrated into the architectural style of the development. All exterior structural and site lighting (not including access road lighting) shall be localized and of full cut-off design, with light which shall be retained on site and shall not create a nuisance to abutting properties and streets. All exterior light sources shall be appropriately shielded from off-premises viewing. Access road lighting shall be designed to prevent nuisance to abutting properties by reason of light and glare.
- **8.2.18** Signs. The need for an identifying sign is very site dependent. Accordingly, the Planning Board shall determine if a sign is needed based on the size, location, and visibility of a project. If the Planning Board determines that a sign is needed, one (1) sign to identify the property, no greater than three (3) square feet in size, and mounted on posts with total height (including the sign) of not more than four (4) feet above the ground, shall be allowed at the intersection of the project's access road or driveway with the abutting public way. Signage may be illuminated with targeted lighting that minimizes light spill-over, but it shall not be backlit or internally illuminated.
- **8.2.19** Site Access, Roadway Design and Management. All Structures located within a Senior Housing development shall be accessed by: (a) an existing Driveway or an existing private way, subject to Planning Board determination as to adequacy of the access, and/or (b) by a new Driveway or a new private way. If access to a proposed development is derived from a private way, then the applicant shall submit a legal opinion, acceptable to Town Counsel, establishing the right of access from the private way to the development as proposed.
 - 1. All access roadways and associated infrastructure improvements shall be considered private, under the ownership of the property owner and/or a homeowner's association established pursuant to G.L. c. 183A, and shall not be owned or maintained by the Town of Hamilton.
 - 2. The Planning Board shall pay special attention to the suitability of the location of the site access road relative to abutting properties when evaluating the special permit application.
- **8.2.20 Stormwater Management and Erosion Control.** The peak rate of storm water runoff from a Senior Housing development shall comply with Department of Environmental Protection Stormwater Management Policy, and the provisions of General By-law XXIX, Stormwater Management By-law. General soil erosion of the proposed development site shall be minimized by integrating the development into the existing terrain and by reasonably retaining natural

grades and soil cover. During grading and construction of all improvements, including all Structures and infrastructure improvements, erosion of soil shall be minimized using best management practices.

- **8.2.21 Parking.** All Senior Housing developments shall provide for adequate off-street parking which will protect the health, safety and welfare of the residents and guests. A minimum of two (2) off-street parking spaces shall be provided per Dwelling Unit. Each garage unit shall be counted as a parking space.
 - 1. Visitor parking spaces shall be provided at a rate of one (1) per five (5) required parking spaces, and shall be clearly marked.
 - 2. A minimum of two (2) parking spaces shall be provided for postal delivery accessory structures, if provided.
 - 3. In all parking areas, the use of low impact development techniques is required, as appropriate, in order to reduce impervious area and runoff.
 - 4. No parking spaces or parking lots shall be located within the required minimum Yard setback requirements for principal Structures from access roads and property lines except that the parking of vehicles shall be allowed within Driveway areas providing access to a principal Structure, as long as the Driveway is designed at a proper width for parking.
 - 5. No parking areas, parking lots or access drives shall be located within the minimum twenty five (25) foot separation area between Structures.
- **8.2.22** Setbacks and Landscape Buffers. Building setbacks shall remain vegetated and undisturbed to the extent possible in order to maintain the existing natural features. The Planning Board may require designated "No Cut Zones" in order to minimize disturbance and impacts to abutting properties. Buffering from the Street and abutting properties in order to minimize visibility is desirable. (see Dimensional Requirements in Table at 8.2.15). A landscaped buffer shall be provided along the perimeter of the property. For said landscaped buffer, natural vegetation shall be supplemented with the planting of evergreen trees and shrubs. A Landscape Plan shall be required.
- **8.2.23 Open Space and Natural Resources** (see Dimensional Requirements in Table at 8.2.15). A Senior Housing development which includes an open space component must provide that land area as permanently protected, usable, common open space that is functional for purposes intended by this Section. The common open space shall have no Structures, parking, private Yards, patios, or gardens that are restricted for the exclusive or principal use by residents of individual Dwelling Units. The following standards apply to the common open space in a Senior Housing development:
 - 1. Ownership of the Common Open Space. Any common open space within a Senior Housing development shall be: (1) conveyed to the Town and accepted for park or open space use; and/or (2) shall be conveyed to a non-profit organization, the principal purpose

of which is the conservation of open space; and/or (3) shall be conveyed to a corporation or trust owned or to be owned by the owners of Lots or residential units within the plan, as provided by G.L. c. 40A, s. 9. In any case where the common open space is not conveyed to the Town, a perpetual restriction enforceable by the Town or the Conservation Commission shall be recorded providing that such land shall be kept in an open or natural state and not be built for residential use or developed for accessory uses such as parking or roadway.

- **8.2.24** Use, Shape, Location of Common Open Space. To the maximum extent feasible, the common open space shall be undisturbed, unaltered and left in its natural condition or existing condition. It shall be appropriate in size, shape, dimension, location, and character to assure its use as a conservation area, or where appropriate, a recreational area, and serve as a visual and natural amenity for the development and the Town.
 - 1. The common open space shall be Contiguous Open Space, to the maximum extent possible.
 - 2. Common open space shall be functional for wildlife habitat, passive recreation, resource preservation, agriculture, or equestrian uses.
 - 3. The location(s) and configuration of the common open space shall be subject to approval by the Planning Board.
 - 4. Each parcel of common open space shall be accessible to residents of the Senior Housing development.
 - 5. Land used for common or shared septic systems and required reserve area may not be counted toward the minimum common open space requirement unless authorized by the Planning Board.
 - 6. Not more than fifty (50) percent of the common open space in a Senior Housing development shall consist of areas subject to the Wetlands Protection Act, G.L. c. 131, s. 40, for reasons other than being subject to flooding, or the Conservancy District as defined by Section 9.3.
 - 7. Existing utility easements may not be counted as common open space.
 - 8. Up to five (5) percent of the minimum required open space may be used for gravel roadways, pavement or structures accessory to the dedicated use or uses of the common open space. Principal or accessory structures and access roads essential to an agricultural use are exempt from this requirement, except for indoor/covered riding rings.

8.2.25 Senior Housing Special Permit Application Process.

- 1. For applications which do not contain an open space element, the application may be filed with the Planning Board in conformity with applicable Planning Board Rules and Regulations and this Section.
- 2. If an application contains an open space element, then the requirements of Section 8.1, Open Space and Farmland Preservation Development, Pre-Application Conference and OSFPD Special Permit Design Process, shall apply. Following the conclusion of this procedure, an application may be filed as set forth below.
- **8.2.26 Definitive Subdivision Plan.** In either case, if a Definitive Subdivision is proposed, a Definitive Subdivision Plan shall be filed in conformity with G.L. c. 41, s. 8lK-81GG, and the Town of Hamilton Subdivision Regulations, along with an application for a Senior Housing special permit. To the extent possible, hearings will be held concurrently.

If a Definitive Subdivision is not proposed, an application shall be filed which includes a Senior Housing Design Plan, which shall conform to Planning Board regulations. A Senior Housing Design Plan shall be considered neither a subdivision plan under the Subdivision Control Law nor a site plan subject to the provisions of Section 10.6. Such plan shall comply with applicable Stormwater Management regulations.

- 8.2.27 Application Process. The applicant shall furnish a copy of the application to the Town Clerk, which shall be time stamped as the official submittal. The applicant shall furnish sufficient copies of the special permit application to the Planning Board to distribute for review to the Board of Health, Board of Selectmen, Building Commissioner, Conservation Commission and Open Space Committee, Department of Public Works, Fire Chief, Police Chief, Historic District Commission, Office on Disability, and Zoning Board of Appeals. The size, quantity, form, and content of plans shall comply with Planning Board Rules and Regulations. Reports from the above-named boards and officials are not mandatory, and if officials wish to comment, reports shall be submitted to the Planning Board within thirty (30) days of receipt by the reviewing party. In the event that the public hearing by the Planning Board is held prior to the expiration of the thirty (30) day period, the Planning Board shall continue the public hearing to permit the formal submission of reports and recommendations within that thirty (30) day period. Planning Board members shall conduct a site visit with the applicant prior to or during the public hearing.
- **8.2.28 Prohibition.** Once a Senior Housing special permit application has been submitted, the following activities are prohibited on any part of the site until the application has been reviewed and decided as provided by this Section: tree removal, utility installation, ditching, grading or construction of roads, grading of land or lots, excavation, except for purposes of soil testing, dredging or filling, and construction of buildings or structures.
- **8.2.29 Conditions of Approval.** The Planning Board may impose conditions, safeguards, requirements, and other standards as part of its approval.

- **8.2.30** Employment of Outside Consultants. The Planning Board may employ outside consultants, at the applicant's expense, under the terms of G.L. c. 44, s. 53G, and Planning Board Rules and Regulations Governing Special Permits, to assist in its permit decision, including but not limited to plan review, drainage and stormwater analysis, to determine conformance with this Section and other requirements, and for construction, inspection, etc.
- **8.2.31 Planning Board Findings.** In addition to the criteria set forth in Section 10.5.2, the Planning Board must make written findings on the following standards for the proposed use, buildings and structures for a Senior Housing development. The proposed Senior Housing development must:
 - 1. Be compatible with adjacent land uses and with the character of the neighborhood in which it is located;
 - 2. Mitigate impact to abutting land and natural resources by reason of air or water pollution, noise, dust, vibration, or stormwater runoff;
 - 3. Provide safe and convenient access to the site from existing or proposed roads, and to proposed structures thereon, with particular reference to pedestrian and vehicular safety, traffic flow and control, and access in case of fire or emergency;
 - 4. Provide for adequate capacity for public services, facilities, and utilities to service the proposed development such as water pressure and sewer capacity;
 - 5. Provide for visual and noise buffering of the development to minimize impact to abutting properties;
 - 6. Provide for the perpetual preservation and maintenance of open space, trails, and recreation areas; and
 - 7. Demonstrate compliance with the intent of Section 8.1 Open Space and Farmland Preservation Development, Special Permit Design Process, in order to encourage cluster development.
- **8.2.32** Expansion. Once any Senior Housing development has been permitted under this Section, further expansion shall not be permitted, and no subdivision of the property or change in property lines shall be allowed. A notation to this effect shall be written on the plan.
- **8.2.33** Annual Reporting. The organization of homeowners established for the management of the development, or if none, the owners individually, shall annually file a written report with the Building Commissioner listing the residents of each occupied Dwelling Unit. The format for the annual report shall be obtained from the Building Commissioner. Said annual report shall include the names and ages of the owners and each person residing in each Dwelling Unit as of January 1st of each year, and any other information necessary to ensure compliance with and enforce any required conditions of special permit. The annual report shall be filed with the Building Commissioner on January 15th of each year.

8.3 INCLUSIONARY HOUSING.

- **8.3.1 Purpose.** The purpose of the Inclusionary Housing By-law is to:
 - 1. Produce high-quality Affordable Housing Units to Low or Moderate Income Households;
 - 2. Encourage more housing choices in Hamilton;
 - 3. Promote geographic distribution of Affordable Housing Units throughout the Town and avoid over-concentration; and
 - 4. Assist the Town in creating units eligible for the Chapter 40B Subsidized Housing Inventory through means other than a comprehensive permit.
- **8.3.2** Applicability. This Section applies to all developments involving the creation of ten (10) or more Dwelling Units or ten (10) or more Lots for residential use.

Developments may not be segmented to avoid compliance with this Section. Divisions of land that would cumulatively result in an increase by ten (10) or more residential Lots or Dwelling Units above the number existing on any parcel or any contiguous parcels in common ownership in the twenty-four (24) months prior to any application for development under this By-law or the subdivision control law are subject to this Section. For purposes of this Section, a division of land shall mean any division of land subject to G.L. c. 41, s. 81K-81GG.

- **8.3.3 Mandatory Provision of Affordable Housing Units.** In any development subject to this Section, the tenth Dwelling Unit and every seventh unit thereafter shall be an Affordable Housing Unit. Nothing in this Section shall preclude a developer from providing more Affordable Housing Units than are required hereunder.
- **8.3.4 Methods of Providing Affordable Housing Units.** The Planning Board may approve one (1) or more of the following methods, or any combination thereof, for the provision of Affordable Housing Units:
 - 1. The Affordable Housing Units may be constructed or rehabilitated on the locus of the development.
 - 2. The Affordable Housing Units may be constructed or rehabilitated on a locus different from that of the development. The Planning Board may allow a developer of non-rental Dwelling Units to develop, construct or otherwise provide Affordable Housing Units reasonably equivalent to those required by this Section in an off-site location in the Town of Hamilton. All requirements that apply to on-site provision of Affordable Housing Units shall apply to provision of off-site Affordable Housing Units. In addition, the location of the off-site Affordable Housing Units shall be approved by the Planning Board as an integral element of the development review and approval process.

- 3. A donation of land may be made in lieu of providing Affordable Housing Units. An applicant may offer, and the Planning Board may accept, subject to approval of the Board of Selectmen, donations of land in fee simple, on-or off-site, that the Planning Board determines are suitable for the construction of an equivalent number of Affordable Housing Units. Land donated for this purpose shall be subject to a restriction assuring its use for affordable housing. Prior to accepting land as satisfaction of the requirements of this Section, the Planning Board may require the applicant to submit an appraisal or other data relevant to the determination of suitability for an equivalent number of Affordable Housing Units.
- 4. An equivalent fee in lieu of units may be made. The Planning Board may allow a developer of non-rental Dwelling Units to make a cash payment to the Town through its Affordable Housing Trust Fund for each Affordable Housing Unit required herein. The cash payment per unit shall be in accordance with the following formula: two times an amount equal to eighty (80) percent of area median income for a Family of four (4) in the metropolitan or non-metropolitan area that includes the Town of Hamilton, as determined annually by the U.S. Department of Housing and Urban Development.
- **8.3.5 General Provisions.** The Planning Board shall be charged with administering this Section and shall promulgate Inclusionary Housing Rules and Regulations, including but not limited to submission requirements and procedures, application and review fees, minimum requirements for a marketing plan, and documentation required by the Town to qualify the Affordable Housing Units for listing on the Chapter 40B Subsidized Housing Inventory.
 - 1. Affordable Housing Units shall be dispersed throughout the Building(s) in a development and shall be comparable to market housing units in terms of location, quality and character, room size, bedroom distribution, and external appearance.
 - 2. The selection of qualified purchasers or qualified renters shall be carried out under a marketing plan approved by the Planning Board and shall comply with Local Initiative Program guidelines. The marketing plan must describe how the applicant will accommodate local preference requirements, if any, established by the Board of Selectmen.
 - 3. Developers may sell Affordable Housing Units to the Town of Hamilton, the Hamilton Housing Authority, or to any non-profit housing development organization that serves the Town of Hamilton, in order that such entity may carry out the steps needed to market the Affordable Housing Units and manage the choice of buyers.
 - 4. Developers shall be responsible for preparing applications and other documentation required by the Department of Housing And Community Development (DHCD) to assure that the Affordable Housing Units are eligible for listing on the Chapter 40B Subsidized Housing Inventory.

8.3.6 Timing of Construction. Unless a different schedule is approved by the Planning Board, Affordable Housing Units shall be provided in proportion to the development of market-rate units, but in no event shall the construction of Affordable Housing Units, the payment of fees in lieu of constructing Affordable Housing Units, or the provision of off-site Affordable Housing Units be delayed beyond the schedule below. Fractions shall be rounded to the nearest whole number.

% Building Permits Issued for Market Rate Units	% Affordable Units (Building Permits, Fees, Off-Site Units, or Land, as
II. 4- 200/	Applicable)
Up to 29%	None required
30%	At least 10%
50%	At least 30%
70%	At least 50%
85%	At least 70%
90%	100%

- **8.3.7 Certificate of Occupancy.** A Certificate of Occupancy for an Affordable Housing Unit shall not be issued until the applicant submits evidence to the Building Commissioner that an Affordable Housing Restriction or a regulatory agreement for the project has been approved by the Planning Board.
- **8.3.8 Preservation of Affordability; Restrictions on Resale.** An Affordable Housing Unit created in accordance with this Section shall be subject to an Affordable Housing Restriction or regulatory agreement that contains limitations on use, resale and rents. The Affordable Housing Restriction or regulatory agreement shall meet the requirements of the Town and the Local Initiative Program, and shall be in force for the maximum period allowed by law.
 - 1. The affordable housing restriction or regulatory agreement shall be enforceable under the applicable provisions of G.L. c. 184, as amended.
 - 2. The Planning Board shall require that the applicant comply with the mandatory provision of Affordable Housing Units and accompanying restrictions on affordability, including the execution of the Affordable Housing Restriction or regulatory agreement.
 - 3. All documents necessary to ensure compliance with this Section shall be subject to the review and approval of the Planning Board and, as applicable, Town Counsel. Such documents shall be executed prior to and as a condition of the issuance of any Certificate of Occupancy.

SECTION 9.0 SPECIAL DISTRICT REGULATIONS

- 9.1 GROUNDWATER PROTECTION OVERLAY DISTRICT (GPOD).
- **9.1.1 Purpose.** The purpose of the Groundwater Protection Overlay District (GPOD) is:

- 1. To promote the health, safety and general welfare of the Town by ensuring an adequate quality and quantity of drinking water for the residents, institutions, and businesses of Hamilton.
- 2. To preserve and protect existing and potential sources of drinking water supplies and recharge areas;
- 3. To conserve the natural resources of the Town; and
- 4. To prevent temporary and permanent contamination of the environment.
- **9.1.2 Overlay District.** The GPOD is an overlay district and shall be superimposed on the other zoning districts established by this By-law. This overlay district shall apply to all new construction, reconstruction or expansion of existing buildings, and new or expanded uses. Applicable activities or uses which fall within the Well Head Protection District must additionally comply with the requirements of the GPOD. Uses that are prohibited in the underlying districts shall not be permitted in the GPOD.
- 9.1.3 Location. The GPOD shall consist of those areas shown on the Hamilton Groundwater Protection Overlay District Map, dated May 1985, amended May 2000 to include the aquifer protection districts of neighboring communities that lie within the Town of Hamilton, amended October 2004, and amended Fall 2015 to incorporate Weston & Sampson's Zone II delineations map dated January 2013. Said map is hereby incorporated into the Zoning By-law by reference, and shall be on file with the Town Clerk.
- **9.1.4 Dimensional Requirements.** Regardless of the minimum Lot size of the underlying zone, there shall be a minimum lot area of eighty thousand (80,000) square feet for a building Lot in the GPOD. See Section 4.0 for Computation of Lot Area.
- 9.1.5 Lot Partially in the GPOD. Any Lot, which has one-third (1/3) or more of its total area falling in the GPOD, must meet all the requirements of the GPOD.
- 9.1.6 Boundary in Doubt. If the location of the GPOD boundary in relation to a particular parcel is in doubt, resolution of boundary disputes shall be through a special permit application to the Zoning Board of Appeals. Any application for a special permit for this purpose shall be accompanied by adequate documentation. The burden of proof shall be upon the owner(s) of the land to show where the bounds should be located. At the request of the owner(s), the Town may engage a professional engineer, hydrologist, geologist, or soil scientist to determine more accurately the boundaries of the GPOD with respect to individual parcels of land, and may charge the owner(s) for the cost of the investigation.
- **9.1.7 Permitted Uses.** Unless prohibited or restricted by the regulations of the GPOD or other state or local regulations, uses or activities permitted in the underlying district are controlled by the requirements of the underlying district.

- **9.1.8 Prohibited Uses.** The following uses are prohibited within the GPOD:
 - 1. Landfills and open dumps as defined in 310 CMR 19.006;
 - 2. Automobile graveyards and junkyards, as defined in G.L. c. 140B;
 - 3. Landfills receiving only wastewater and/or septage residuals including those approved by the Department of Environmental Protection (DEP) pursuant to G.L. c. 21, ss. 26 through 53; G.L. c. 83, ss. 6 and 7; and regulations promulgated thereunder;
 - 4. Facilities that generate, treat, store or dispose of hazardous waste subject to G.L. c. 21C and 310 CMR 30.00, except the following:
 - a. Very small quantity generators as defined under 310 CMR 30.00;
 - b. Household hazardous waste collection centers or events operated pursuant to 310 CMR 30.390;
 - c. Waste oil retention facilities required by G.L. c. 21, s. 52A;
 - d. Water remediation treatment works approved by the Massachusetts Department of Environmental Protection ("DEP") for the treatment of contaminated ground or surface waters;
 - 5. Storage of liquid hazardous materials, as defined in G.L. c. 21E and liquid petroleum products, including petroleum, fuel oil, heating oil bulk stations and terminals pursuant to 310 CMR 22.21(2)(a)(5), unless such storage:
 - a. Is the storage of (i) heating oil for consumptive use on the premises where stored, or (ii) motor oil/fuel for noncommercial purposes that is stored in farm or residential tanks of 1100 gallons capacity or less; and
 - b. Is in a fuel oil tank installed either within the Building, which it will heat, or above ground outside the Building, with surfaces underlying such tank being impermeable to fuel oil and enclosed by a permanent berm or dike of impermeable construction capable of containing ten (10) percent of the total volume of the tank (in the case of more than one (1) tank, the containment must hold either ten (10) percent of the total possible storage capacity of all containers or one hundred ten (110) percent of the largest container's storage capacity whichever is greater); and
 - c. To the extent that such storage involves any supply or return lines carrying fuel oil underground or under the basement floor surface, such lines shall be enclosed in conduit piping impervious to fuel oil to prevent leakage into the soil and in compliance with 527 CMR 4.0 et. seq. and any other applicable state regulations. All leaking tanks existing on the property must be repaired or emptied within

twenty-four (24) hours of leak detection. All State regulations pertaining to fuel tanks are also applicable to tanks governed by this Section.

- 6. Storage of sludge and septage, unless such storage is in compliance with 310 CMR 32.30 and 310 CMR 32.31;
- 7. Storage of deicing chemicals unless such storage, including loading areas, is within a Structure designed to prevent the generation and escape of contaminated runoff or leachate;
- 8. Storage of animal manure, unless covered or contained in accordance with the specifications of the United States Natural Resources Conservation Services;
- 9. Earth removal consisting of the removal of soil, loam, sand, gravel or any other earth material (including mining activities) to within four (4) feet of the historical high ground water table as determined from monitoring wells and historical water table fluctuation data compiled by the United States Geological Survey, except for excavations for building foundations, roads, or utility works;
- 10. Non-sanitary treatment or disposal works subject to 314 CMR 5.00, except for the following:
 - a. The replacement or repair of an existing treatment works that will not result in a design capacity greater than the design capacity of the existing treatment works;
 - b. Treatment works approved by the Department of Environmental Protection (DEP) designed for the treatment of contaminated ground or surface water and operating in compliance with 314 CMR 5.05(3) or 5.05(13); or
 - c. Publicly owned treatment works;
- 11. Stockpiling and disposal of snow and ice containing deicing chemicals if brought in from outside the GPOD;
- 12. Storage of commercial fertilizers as defined in G.L. c. 128,s. 64, unless such storage is within a Structure designed to prevent the generation and escape of contaminated runoff or leachate.
- **9.1.9 Uses and Activities Requiring a Special Permit.** The following uses and activities are permitted only upon the issuance of special permit by the Zoning Board of Appeals under such conditions as it may require:
 - 1. Enlargement or alteration of existing uses that do not conform to the GPOD;
 - 2. Activities that involve the handling of toxic or hazardous materials in quantities greater than those associated with normal household use, permitted in the underlying

district, except as prohibited herein. Such activities shall require a special permit to prevent contamination of groundwater;

- 3. Any use that will render impervious more than fifteen (15) percent or twenty five hundred (2,500) square feet of any Lot, whichever is greater. A system for groundwater recharge must be provided which does not degrade groundwater quality. For nonresidential uses, recharge shall be by storm water infiltration basins or similar system covered with natural vegetation, and dry wells shall be used only where other methods are infeasible. For all non-residential uses, all such basins and wells shall be preceded by oil, grease, and sediment traps to facilitate removal of contamination. Any and all recharge areas shall be permanently maintained in full working order by the owner.
- **9.1.10 Special Permit Application.** The applicant shall file eight (8) copies of a site plan and attachments. The site plan shall meet the requirements for site plan review and approval established by Section 10.6. Additional submittals shall include the following information where pertinent:
 - 1. A complete list of chemicals, pesticides, herbicides, fertilizers, fuels and other potentially hazardous material to be used or store on the premises in quantities greater than those associated with normal household use;
 - 2. For those activities using or storing such hazardous materials, a hazardous materials management plan shall be prepared and filed with the Hazardous Materials Coordinator, Fire Chief and Board of Health. The plan shall include:
 - a. Provisions to protect against the discharge of hazardous material or wastes to the environment due to spillage, accidental damage, corrosion, leakage or vandalism, including spill containment and clean-up procedures;
 - b. Provisions for indoor, secured storage of hazardous materials and wastes with impervious floor surfaces; and
 - c. Evidence of compliance with the Regulations of the Mass Hazardous Waste Management Act 310 CMR 30, including obtaining an EPA identification number from DEP.
 - 3. Proposed down-gradient location(s) for groundwater monitoring well(s), should the Board deem the activity a potential groundwater threat.
- **9.1.11 Procedures for Issuance of Special Permits.** The SPGA under this By-law shall be the Zoning Board of Appeals. A special permit may be granted if the Zoning Board of Appeals determines, in conjunction with the Board of Health, the Conservation Commission, the Planning Board and the DPW Superintendent that the intent of this Section, as well as its specific criteria are met. The Zoning Board of Appeals shall not grant a special permit under this Section unless the petitioner's application materials include, in the Zoning Board of Appeals' opinion, sufficiently detailed, definite and credible information to support positive findings in relation to

the standards given in this Section. The Zoning Board of Appeals shall document the basis for any departures from the recommendations of the other Town boards or agencies in its decision. Upon receipt of the special permit application, the Zoning Board of Appeals shall transmit one (1) copy to the Board of Health, Conservation Commission, Planning Board and DPW Superintendent for their recommendations. Failure to respond in writing within thirty five (35) days shall indicate approval or no desire to comment by said Board or official. The necessary copies of the application shall be furnished by the applicant.

- **9.1.12** Special Permit Criteria. The Zoning Board of Appeals may grant the required special permit only upon finding that the proposed use meets the following standards, those specified in Section 10.5 of this By-law, and any regulations or guidelines adopted by the Zoning Board of Appeals. The proposed use must:
 - 1. In no way, during construction or thereafter, adversely affect the existing or potential quality or quantity of water that is available in the GPOD; and
 - 2. Be designed to avoid substantial disturbance of the soils, topography, drainage, vegetation, and other water-related natural characteristics of the site to be developed.
- **9.1.13 Enforcement.** Written notice of any violations of this Section shall be given by the Building Commissioner to the responsible person as soon as possible after detection of a violation or continuing violation. Notice to the assessed owner of the property shall be deemed notice to the responsible person. Such notice shall specify the nature of the violation and may also identify the actions necessary to remove or remedy the violation, measures required for avoiding future violations and schedule of compliance. A copy of such notice shall be submitted to the Zoning Board of Appeals, Board of Health, Conservation Commission, Planning Board and DPW Superintendent and Water Department. The cost of containment, clean-up or other action for compliance shall be borne by the owner and operator of the premises.

9.2 FLOOD PLAIN OVERLAY DISTRICT (FPOD).

- **9.2.1 Purpose.** The purpose of the Floodplain Overlay District (FPOD) is:
 - 1. To ensure public safety through reducing the threats to life and personal injury;
 - 2. To eliminate new hazards to emergency response officials;
 - 3. To prevent the occurrence of public emergencies resulting from water quality, contamination, and pollution due to flooding;
 - 4. To avoid the loss of utility services which if damaged by flooding would disrupt or shut down the utility network and impact regions of the community beyond the site of flooding;
 - 5. To eliminate costs associated with the response and cleanup of flooding conditions; and

- 6. To reduce damage to public and private property resulting from flooding waters.
- 9.2.2 FPOD Boundaries. The FPOD is herein established as an overlay district. The FPOD includes all Areas of Special Flood Hazard within the Town of Hamilton designated as Zones A and AE on the Essex County Flood Insurance Rate Map (FIRM) issued by the Federal Emergency Management Agency (FEMA) for the administration of the National Flood Insurance Program. The map panels of the Essex County FIRM that are wholly or primarily within the Town of Hamilton are panel numbers 25009C0267F, 25009C0268F, 25009C0269F, 25009C0288F, 25009C0288F, 25009C0289F, 25009C0293F, 25009C0406F, 25009C0407F, 25009C0427F, 25009C0429F, 25009C0431F, and 25009C0433F dated July 3, 2012. The exact boundaries of the FPOD may be defined by the 100-year base flood elevations shown on the FIRM and further defined by the Essex County Flood Insurance Study (FIS) report dated July 3, 2012. The FIRM and FIS report are incorporated herein by reference and are on file with the Town Clerk, Board of Selectmen, Planning Board, and Conservation Commission.

9.2.3 Base Flood Elevation and Floodway Data.

- 1. Floodway Data. In Zones A and AE of FIRM, along watercourses that have not had a Regulatory Floodway designated, the best available Federal, State, local, or other floodway data shall be used to prohibit encroachments in floodways which would result in any increase in flood levels within the community during the occurrence of the Base Flood discharge.
- 2. Base Flood Elevation Data. Base Flood elevation data is required for subdivision proposals or other developments greater than 50 lots or 5 acres, whichever is the lesser, within unnumbered A zones.
- **9.2.4 Use Regulations.** All Development in the district, including structural and non-structural activities, whether permitted by right or by special permit must be in compliance with the Massachusetts Wetlands Act (G.L. c. 131, s. 40) and with the following:
 - 1. The section of the Massachusetts State Building Code which addresses floodplain and coastal high hazard areas (currently 780 CMR 120.G "Flood Resistant Construction and Construction in Coastal Dunes");
 - 2. The Wetlands Protection Regulations promulgated by the Massachusetts Department of Environmental Protection (DEP) (currently 310 CMR 10.00);
 - 3. The Inland Wetlands Restriction Regulations promulgated by DEP (currently 310 CMR 13.00);
 - 4. The Minimum Requirements for the Subsurface Disposal of Sanitary Sewage promulgated by the DEP (currently 310 CMR 15, Title 5); and

5. The Hamilton Board of Health Regulations, Chapter 6. Satisfactory evidence shall consist of any permit issued by the Board of Health for the project and/or a written statement from the Board of Health or its agent that no such permit is necessary for the project.

9.2.5 Other Requirements.

- 1. Any variances from the provisions and requirements of the above referenced State regulations may only be granted in accordance with the required variance procedures of these State regulations.
- 2. Within Zones AH and AO of the FIRM, adequate drainage paths must be provided around Structures on slopes to guide floodwaters around and away from proposed Structures.
- 3. In Zone AE of FIRM, along watercourses within the Town of Hamilton that have a Regulatory Floodway designated on FIRM, encroachments are prohibited in the Regulatory Floodway which would result in any increase in flood levels within the community during the occurrence of the Base Flood discharge.
- 4. All subdivision proposals must be designed to assure that such proposals minimize flood damage; all public utilities and facilities are located and constructed to minimize or eliminate flood damage; and adequate drainage is provided to reduce exposure to flood hazards.
- 5. Existing contour intervals of site and elevations of existing structures must be included on the development plan proposal.
- 6. There shall be established a "routing procedure" which will circulate or transmit one copy of the development plan to the Conservation Commission, Planning Board, Board of Health, Director of Public Works, and Building Commissioner for comments which will be considered by the appropriate permitting board prior to issuing applicable permits.
- **9.2.6 Permitted Uses.** The following uses of low flood damage potential and causing no obstructions to flood flows are permitted provided they are permitted in the underlying district and they do not require structures, fill, or storage of materials or equipment:
 - 1. Agricultural uses such as farming, grazing, truck farming, horticulture, etc.;
 - 2. Forestry and nursery uses;
 - 3. Outdoor recreational uses, including fishing, boating, play areas, etc.;
 - 4. Conservation of water, plants, wildlife;
 - 5. Wildlife management areas, foot, bicycle, and/or horse paths;

- 6. Temporary nonresidential structures used in connection with fishing, growing, harvesting, storage, or sale of crops raised on the premises; and
- 7. Buildings lawfully existing prior to the adoption of these provisions.
- **9.2.7 Notification of Watercourse Alteration.** In a riverine situation, the Conservation Commission shall notify the following of any alteration or relocation of a watercourse:

Adjacent Communities NFIP State Coordinator Massachusetts Department of Conservation and Recreation 251 Causeway Street, Suite 600-700 Boston, MA 02114-2104

NFIP Program Specialist Federal Emergency Management Agency, Region I 99 High Street, 6th Floor Boston, MA 02110

9.3 CONSERVANCY DISTRICT.

- **9.3.1 Purpose.** The Conservancy District is intended for the following purposes:
 - 1. The preservation and maintenance of groundwater quality and quantity;
 - 2. The protection of watershed resources upon which the inhabitants depend for water supply;
 - 3. For the enhancement of water quality by encouraging infiltration and percolation through natural soils;
 - 4. To protect the public health and safety, persons and property against hazards of flood water inundation;
 - 5. For the protection of the community against the costs which may be incurred when unsuitable development occurs in swamps, marshes, along watercourses, or in areas subject to floods;
 - 6. To preserve and increase the amenities of the Town; and
 - 7. To conserve natural conditions, wild life, and open spaces for the education, recreation and general welfare of the public.

- **9.3.2 Permitted Uses.** Only the following uses are permitted in the Conservancy District (Note: Construction or alteration in Conservancy District will usually also require application to the Hamilton Conservation Commission under the Wetlands Protection Act and the Hamilton Wetlands By-law):
 - 1. Conservation of water, plants, wildlife, and ponds;
 - 2. Passive recreation activities which do not require paving, filling, or construction of facilities that may degrade water quality, including but not limited to nature study, walking, pond skating, canoeing, fishing, horseback riding, and hunting where otherwise legally permitted;
 - 3. Agricultural uses, forestry, horticulture and floriculture including grazing of animals and harvesting of crops, provided that use of chemical fertilizers, pesticides or defoliants found by the Board of Health or the Conservation Commission to be inconsistent with the purposes of this Conservancy District may be regulated by said Boards. (Note: See also Board of Health Animal Regulations); and
 - 4. Religious and educational uses on land owned or leased by the Commonwealth of Massachusetts, or its agencies, subdivisions or bodies politic, or by a religious sect or denomination, or by a non-profit educational corporation. (Note: Site Plan Review may be required. See Section 10.6.).

(Note: Construction or alteration in Conservancy District will usually also require application to the Hamilton Conservation Commission under the Wetlands Protection Act and the Hamilton Wetlands By-law.)

- **9.3.3 Special Permit.** Subject to a special permit by the Zoning Board of Appeals under specified conditions as provided for in Section 10.5, and only when the Conservation Commission has reported to the Zoning Board of Appeals that such activities will not have adverse effects relating to water quality or recharge, upstream or downstream flooding and drainage, and water storage capacity in the Conservancy-District, the following uses are permitted:
 - 1. Nonresidential Buildings and Structures, such as boathouses, duckwalks and landings;
 - 2. Stands for sale of produce raised on the premises, provided that any such Building or Structure permitted by the Zoning Board of Appeals shall be designed, placed and constructed to offer minimum obstruction to the flow of water and shall be firmly anchored to prevent floating away and thus threatening other buildings or blocking of openings in restricted sections of the watercourse below (Note: Special permit and site plan review required per Sections 10.5 and 10.6.);
 - 3. Dams, changes in watercourses, or drainage works, only as part of an over-all drainage basin plan;

- 4. Uses, whether or not on the same parcel as activities permitted as a matter of right, which activities are necessary in connection with scientific research or scientific development or related production provided the Zoning Board of Appeals finds that the proposed use does not substantially derogate from the public good, and that the proposed use does not include impermeable ground cover in excess of 10% of the total lot area, or any storage of chemical substances which have the potential for degrading groundwater quality. (Site plan review under Section 10.6);
- 5. Wind Energy Facilities and Utility Scale Wind Energy Facilities, subject to the standards set forth in Section 7.3; and
- 6. When any land shown on the Zoning Map as being in the Conservancy District is proven to the satisfaction of the Zoning Board of Appeals, after reference of the proof to and report by the Conservation Commission, Planning Board and Board of Health, as being in fact not subject to flooding or not unsuitable because of drainage conditions for residential use, and that neither construction activities nor the use of such land for residence will interfere with the general purposes for which Conservancy Districts have been established, and will not be detrimental to the public health, safety or welfare, the Zoning Board of Appeals may, by grant of a special permit, permit the use of such land for Single Family Dwellings under all the provisions of this By-law applying to the Residence District in which the land lies.
- **9.3.4 Prohibited Uses.** Except as provided, above, the following are prohibited in the Conservancy District:
 - 1. Land filling and dumping of any natural or other materials;
 - 2. Construction of Buildings or Structures;
 - 3. Paving and other impermeable surfaces;
 - 4. Construction of roads and driveways;
 - 5. Permanent storage of materials or equipment:
 - 6. Storage of chemical substances which may degrade water quality;
 - 7. Commercial borrow operations:
 - 8. Construction of dams, drainage systems and changes in watercourses which may speed drainage through or around a natural watershed, thereby reducing opportunities for water quality renovation by percolation or infiltration through soils; and
 - 9. Alteration of terrain which may result in increased discharge of stormwater runoff into the District.

9.3.5 Reversion. If for any reason the restrictions or requirements contained in this Section shall be or become invalid as to any land shown on the Zoning Map as being in the Conservancy District, then such land shall thereafter be subject to the regulation for the district in which the land lies.

9.4 ESTATE OVERLAY DISTRICT (EOD).

- 9.4.1 Purpose. The purpose of the Estate Overlay District ("EOD") is:
 - 1. To allow for the preservation of the buildings and land of large estate properties while allowing for appropriate development which has a positive net fiscal benefit to the Town and is an alternative to the subdivision of an estate property;
 - 2. To allow for the efficient use of such land in harmony with the natural features of the district while preserving open space for conservation, agricultural, or recreational use, and providing appropriate public access to open space;
 - 3. To protect natural features that are important to the character of the town, including public vistas of fields, lakes, and rivers from town roads as well as the existing non-vehicular connections through trails and pathways; and
 - 4. To protect Hamilton's rural and agricultural New England character and provide for its continued uses by permitting a mix of commercial and residential development in a manner which is in harmony with Hamilton's historic development patterns.
- **9.4.2 Overview.** The Estate Overlay District By-law is a special district overlaying the R-la, R-1b and RA districts. The EOD has two (2) distinct subsections:
 - 1. Part A An Estate House Adaptive Reuse Permit, which allows for new commercial uses as well as currently allowed and expanded residential uses in existing Estate Houses; and
 - 2. Part B An Estate House Adaptive Reuse and New Floor Area Construction for Commercial Use Permit, which allows for new commercial and expanded residential uses in the existing Estate House and new construction to create space for commercial use in an addition or in an associated Building.

Both Part A and Part B are allowed by special permit. An applicant must apply for Part A first and choose to file a separate special permit application for Part B after an occupancy permit has been issued for Part A.

- **9.4.3 Applicability.** A property in the EOD is eligible for a Part A special permit if it has a minimum Lot size of fifteen (15) acres, and is eligible for a Part B special permit if it has a minimum Lot size of twenty (20) acres. In calculating the Lot size:
 - 1. Contiguous Lots may be combined provided that:

- a. At least one (1) of the Lots contains an Estate House of over five thousand (5,000) Residential Gross Square Feet of living area, built before December 31, 1950;
- b. The Lots as combined have the required acreage; and
- c. The Lots have been under common ownership prior to July 1, 2010.
- 3. Lots may be considered combined regardless of whether a private or public road dissects the parcels.
- 4. Land restricted by conservation easements/restrictions or wetlands will count toward the fifteen (15) acre and twenty (20) acre thresholds.
- **9.4.4** Access Requirements for the EOD. Access to the Lot must be through existing frontage on a public way or through existing frontage on a private way if all abutters to the entire length of the private way agree in writing to the development proposal.
- **9.4.5 Open Space Requirement.** An EOD project must contain a minimum of forty (40) percent open space. The following options shall be used to meet this requirement:
 - 1. A minimum of forty (40) percent of the parcel must already be or shall be conveyed to the Town and accepted by it for open space use;
 - 2. A minimum of forty (40) percent of the parcel must be made subject to a conservation restriction prepared in accordance with the provision of G.L. c. 184, ss. 31-33, running in favor of either the Town or, upon the approval of the Board of Selectmen with advice from the Planning Board, to a non-profit corporation, the principal purpose of which is the conservation of open space;
 - 3. Land that has already been made subject to a conservation restriction as delineated above shall count toward the forty (40) percent threshold;
 - 4. No more than fifty (50) percent of the required open space shall consist of resource areas subject to the Wetlands Protection Act, G.L. c. 131, s. 40, (excepting the resource area "isolated land subject to flooding") or land within the Conservancy District;
 - 5. Land used for common or shared septic systems may not be counted toward the minimum common open space requirement unless authorized by the Planning Board.
- **9.4.6 Open Space Access.** Unless the applicant can provide a compelling reason to the Planning Board why public access would be infeasible, a portion of the open space shall be available for public access and use, for example, wildlife corridors, water protection, view-sheds, trails or other access which provides a public benefit. The Planning Board shall take into account the size of the parcel, the customary and agricultural uses, and whether or not it can

provide adequate parking and other facilities to allow for public use. If the property size or structure cannot meet public parking requirements, or if public use does not fit with the character of the neighborhood, the Planning Board may restrict public use.

- 9.4.7 Review Criteria. The Planning Board is the SPGA for the EOD and shall apply the provisions of Section 10.5.2 consistent with the purposes set forth in Section 9.4.1.
- **9.4.8 Advisory Opinions.** Within ten (10) days of the receipt of a special permit application in the EOD, the Planning Board shall transmit copies of the application to the Economic Development Committee, Historical District Commission, Board of Health, Department of Public Works, Conservation Commission, Zoning Board of Appeals, and the Fire and Police Departments for their review and advisory opinions, which shall be submitted within thirty (30) days.
- **9.4.9 Phasing.** Phasing in the EOD, as approved by the Planning Board, shall be permitted either pursuant to phasing described in the initial special permit application or in a subsequent special permit application. The special permit approval shall not be deemed to have lapsed as long as the applicant shall have commenced use of the phasing time frames set forth in the special permit application. The Planning Board shall have the authority to require one or more performance bonds or other similar mechanisms if it determines that such mechanisms are necessary to ensure that the key components or phase of a project are satisfactorily completed.
- 9.4.10 Pre-Application Conference. The applicant shall request and attend an EOD preapplication review at a regular business meeting of the Planning Board. The Planning Board shall invite the Board of Health, Building Commissioner, Conservation Commission and Open Space Committee, Department of Public Works, Fire Chief, Police Chief, Historic District Commission, Office on Disability, and Zoning Board of Appeals to attend the pre-application conference. The primary purpose of the EOD special permit pre-application conference is to identify the site's natural or historically important features, riding or walking trails on the site and abutting parcels, public safety, traffic or infrastructure issues, and areas the Town prefers to see preserved as open space, for agricultural or equestrian uses. The secondary purpose of a preapplication review is to minimize the applicant's cost of engineering and other technical experts, and to commence discussions with the Planning Board at the earliest possible stage of development. Meetings may be held by mutual agreement of the Planning Board and the applicant. At the pre-application review, the applicant shall outline the proposed Estate Overlay District proposal, seek preliminary feedback from the Planning Board, and set a timetable for submittal of a formal application. At the expense of the applicant, the Planning Board may engage technical experts to review the informal plans of the applicant and to facilitate submittal of a formal application for an Estate Overlay District special permit.

9.4.11 EOD Special Permit Design Process.

1. *Identify Conservation Areas*. The first step in the design process requires identification of conservation areas on the site, including wetlands, areas located in the Hamilton Conservancy District, riverfront areas, and floodplains regulated by state, federal, or local law, unprotected natural landscape features such as steep slopes, mature

woodlands, prime farmland, meadows, wildlife habitats and wildlife corridors or connections thereto, cultural features such as historic and archeological sites and scenic views, and recreational features such as established trails used for horseback riding, walking and cross-country skiing. To the maximum extent feasible, conservation areas shall include areas identified by the Planning Board during the pre-application conference.

- 2. Identify and Delineate the Proposed Development Area. The second step in the design process is to define and delineate the area that will contain buildings, roadways, other site improvements and amenities for the development. To the maximum extent feasible, the proposed development area shall consist of land outside the identified conservation areas.
- 3. Locate Dwelling and Commercial Units. The third step in the design process is to identify and delineate the approximate location of Dwelling Units or commercial units in the existing Estate House or commercial units in the proposed development area. The location, height, density, and architectural treatment of all Buildings proposed for construction, renovation or re-use should be included.
- 4. Locate Open Space. The fourth step in the design process is to identify and delineate the open space to be dedicated along with any potential uses.
- 5. Roads and Trails. The fifth step in the design process is to identify and delineate the approximate location of roads, parking area and trails. Roads shall be aligned to access the Dwelling Units and commercial units. The layout of new trails should anticipate internal and external connections to existing and/or potential future roads, trails and sidewalks.
- 6. Utilities. The type and probable location of the proposed utilities should be included.
- **9.4.12 Certification.** The special permit plan should be accompanied by a certification or letter from the Hamilton Historical District Commission of all Historically, Culturally and/or Architecturally Significant Buildings, landscape features and supporting Structures located on the site if the applicant intends on using this determination as a way to obtain new floor area.
- 9.4.13 Special Permit Application. All special permit applications for an EOD shall be made and filed with the special permit application form. For an application to be considered complete, it shall provide all information required by the Planning Board's Rules and Regulations Governing Special Permits. The special permit application shall also be accompanied by certification from the Hamilton Historical Commission of all Historically, Culturally and/or Architecturally Significant Buildings (if applying for the bonus square footage for renovation under Part B), a plan of all existing and proposed Buildings, landscape features and supporting Structures located on the site and by nine (9) copies of a site development report. The site development report should include, at minimum, an inventory and analysis of:

- 1. Natural resource features, wildlife and their habitat;
- 2. All Buildings and Structures;
- 3. Pedestrian and vehicular access to the site;
- 4. Public safety issues;
- 5. Provision of landscaping/buffering;
- 6. Protection of wildlife habitats;
- 7. Provision of utilities;
- 8. Open space and recreation, including trail development and/or maintenance;
- 9. Water supply and drainage issues;
- 10. Layout and density of site development;
- 11. Preservation and rehabilitation of the exterior features, character and structural integrity of the estate house, and the open space, vistas, stonework, gardens, and other historic features and supporting Structures;
- 12. Building design and materials, including exterior elevations of existing and proposed buildings and additions;
- 13. Proposed uses for commercial units;
- 14. Trails or pathways provided or maintained to connect pedestrian, equestrian, non-motorized-bicycle riders or skiers to sites within the property and to other public trail systems if possible; and
- 15. To the extent possible, the information provided in the report shall be shown in plans and map form, accompanied by written narrative.
- **9.4.14** Criteria. In addition to the criteria set forth in Section 10.5.2, the Planning Board shall consider the feasibility of the project given the size, location and scope of the property, proximity to abutters, safety concerns, impact on the property, and whether the proposed use is compatible with the character of the Town of Hamilton and the neighborhood the property is located within.
- **9.4.15 Modifications.** Any new use or change of use will require an amendment of the EOD special permit by the Planning Board.

EOD Part A - Estate House Adaptive Reuse.

- **9.4.16 Part A.** Estate House Adaptive Reuse allows for the estate owner to apply for a special permit from the Planning Board to use the Estate House in accordance with one of the following adaptive reuse criteria. Special permits allow for each application to be considered individually based on their specific location and impact to the existing neighborhood. Throughout this Section, residential units are limited to two (2) bedrooms each. This is designed to achieve the goal of providing alternative housing for groups that do not currently have housing options within the Town. The Planning Board may, in its sole discretion, allow a developer to increase the number of bedrooms to a maximum of three (3), provided the developer can demonstrate a compelling reason why such increase serves the stated goals of the EOD.
- **9.4.17** Part A Permitted Uses. The following use or a combination of the following uses may be permitted in the original Estate House under Part A:
 - 1. Any uses listed in the Table of Uses as an allowable use in the zoning district in which the property currently exists, whether by special permit or otherwise;
 - 2. Multi-family condominium, cooperative, and apartment residential Dwelling Units, provided that residential units shall be limited to no more than two (2) bedrooms per unit; or to thre (3) bedrooms if permitted by the Planning Board pursuant to Section 9.4.16 above;
 - 3. For profit function halls, meeting rooms, inns, Bed and Breakfasts, spas, and Restaurants (excluding Fast Food or Drive In or Drive Through Establishments);
 - 4. For Profit Educational Facilities:
 - 5. Business or Professional Office:
 - 6. Child Care Centers, Adult Social Day Care age restricted to Elderly Persons, and Assisted Living Residence; and
 - 7. Museums.
- **9.4.18 Part A Dimensional Requirements.** A minimum vegetated buffer strip of three hundred (300) feet shall be provided unless the Building is nearer to the abutting property line or at least one hundred (100) feet, provided that the Planning Board may increase the vegetated buffer strip based on factors that shall include, but not be limited to: the size and location of the parcel, the character of the neighborhood and the intended use.
- **9.4.19 Part A Special Permit Application.** All Part A special permit applications shall include building plans prepared by a registered architect and filed with the special permit application form. For an application to be considered complete, it shall provide all information required by the Hamilton Planning Board Rules and Regulations Governing Special Permits.

- **9.4.20** Part A Required Information. The special permit application shall include a plan of the existing Estate House and nine (9) copies of a Site Re-use Report. The Site Re-use Report should include an outline of how the following issues and impacts will be addressed by the re-use:
 - 1. Pedestrian and vehicular access to the site including anticipated changes to vehicular traffic patterns;
 - 2. Public safety issues;
 - 3. Provision of landscaping/buffering;
 - 4. Open space and recreation, including trail development and/or maintenance;
 - 5. Water supply and drainage issues;
 - 6. Proposed uses for commercial units, if any;
 - 7. Conceptual floor plans for units;
 - 8. Trails or pathways will be provided or maintained to connect pedestrian, equestrian, non-motorized-bicycle riders or skiers to sites within the property and to other public trail systems if possible; and
 - 9. To the extent possible, the information provided in the report shall be shown in building and site plans, accompanied by a written narrative describing the proposed adaptive reuse.

EOD Part B - New Floor Area Construction

- **9.4.21 Part B.** New Floor Area construction allows for the estate owner to improve the existing Estate House with expanded residential and commercial uses while increasing floor space by construction of an addition to the original Estate House and/or constructing an associated Building for commercial uses as indicated below. The New Floor Area uses are limited to commercial only. The new uses may be allowed by special permit with the Planning Board acting as the SPGA. Special permits allow for each application to be considered individually based on their specific location and impact to the existing neighborhood.
- **9.4.22** Part B Permitted Uses. Along with those uses allowed in Part A, the following single commercial use or a combination of the following commercial uses may be permitted in the New Floor Area under Part B (note that residential use is not permitted in any New Floor Area under Part B):
 - 1. Outdoor recreational facility or center such as cross country skiing, golf, horseback riding, or non-motorized cross country bike riding facility including any on-site trails;

- 2. Long Term Care Facility providing continuing residential care for seniors or the infirm; and
- 3. Additional uses which are consistent with the objective and intention of this Section and the character of the neighborhood.
- **9.4.23** Part B Methods of Achieving New Floor Area. There are three methods to obtain New Floor Area to construct commercial space on the property in addition to re-use of the original Estate House. The methods are intended to promote development which encourages preservation of both Hamilton's historic buildings and open space.
 - 1. Historic Rehabilitation of the Original Estate House. An application that proposes to rehabilitate or renovate Buildings and supporting Structures certified by the Hamilton Historic District Commission as being Historically, Culturally, or Architecturally Significant may increase New Floor Space by three times the amount of square footage contained in all existing Buildings that are rehabilitated or renovated as part of the EOD development. The Planning Board shall refer to the Secretary of the Interior's Standards for Rehabilitation and Guidelines for Rehabilitating Historic Buildings for guidance when reviewing the Buildings which have been, or are proposed to be, rehabilitated or renovated.
 - 2. Non-historic Rehabilitation of Original Estate House. If the Planning Board determines that the reconstruction or replacement of existing Buildings or supporting Structures that have not been certified by the Hamilton Historic District Commission as being Historically, Culturally or Architecturally Significant are more consistent with the Building's rehabilitation or renovation, then the applicant may increase New Floor Area by the amount of square footage contained in all existing Buildings that are rehabilitated or reconstructed. The renovated or rehabilitated Building or supporting Structure may not be torn down.
 - 3. Yield Analysis Plan of Developable Land. For the purpose of this By-law, two thousand (2,000) square feet of New Floor Area may be constructed for each Lot that could be created on the property as shown by a Yield Analysis Plan with a maximum of thirty-thousand (30,000) of additional square feet of New Floor Area obtained from this subsection. The applicant shall provide with the application for special permit a Yield Analysis Plan together with verifiable soil tests indicating the number of buildable lots possible under single family zoning, the State Environmental Code, Title V, the requirements of the Board of Health, the Wetlands Protection Act, and the Hamilton Conservation By-law and Rules and Regulations. Such soil tests shall be conducted as if they were actually percolation tests in accordance with the above-referenced requirements and shall be verified and attested to by a registered professional engineer. Buildable lot calculation can only apply to land that is not under current restriction (excluding G.L. c. 61A or c. 61B).

- **9.4.24** Part B Dimensional Requirements. The following dimensional requirements apply in Part B:
 - 1. Newly constructed or renovated buildings in an EOD must be thirty-five (35) feet or lower in height unless the original Estate House is greater than thirty-five (35) feet high in which case, newly constructed or renovated additions or any other new Buildings will be limited to the height of the original Estate House, but in no case greater than four (4) stories;
 - 2. Newly constructed Buildings shall be clustered within one hundred (100) feet of the original Estate House unless the developer has a specific goal consistent with the purposes of this By-law that allows for placement of the newly constructed Buildings outside of this area that achieves the cohesive integration of the new Buildings into the whole;
 - 3. In no event will new construction within the three-hundred (300) foot buffer be permitted any nearer the property line than the greater of at least one hundred (100) feet or the distance from the property line of the current Building; and
 - 4. A vegetated buffer strip with a minimum depth of fifty (50) feet shall be provided between the Buildings of an EOD and any public or private way provided that the Planning Board may increase the vegetated buffer strip based on factors that shall include, but not be limited to the size and location of the parcel, the character of the neighborhood and the intended use. An entry drive, along with a gate house and appropriate signage, may be permitted within the buffer strip.

9.5 WILLOW STREET OVERLAY DISTRICT (WSOD).

- **9.5.1 Purpose.** The general purpose of the Willow St. Overlay District (WSOD) is to encourage planned and orderly development of land within the boundaries of the WSOD. These areas are designated for special land use, planning, environmental, and access considerations that warrant careful control of development under a special permit procedure using appropriate standards. The specific purposes of the WSOD are the following:
 - 1. To support land uses and activities that will promote economic development in Hamilton, thereby generating substantial increases in net tax revenue from new development and business growth as well as additional local jobs;
 - 2. To allow for the phased development of new uses and activities that are appropriate and desirable for the WSOD, based on planning and market oppoliunities identified in the Hamilton Master Plan;
 - 3. To provide for mitigation of any negative impacts caused by new development, including but not limited to transportation and environmental impacts;

- 4. To incorporate into the design review and permitting process input from Hamilton's land use boards, nearby residential abutters and citizens; and
- 5. To improve the aesthetics, vibrancy and quality of life for the neighborhoods that include the downtown business district.
- **9.5.2 Location.** The Willow Street Overlay District shall consist of an area within the Business District, west of the railroad tracks and along Willow Street from Asbury Street, including 203 Willow Street.
- **9.5.3 Applicability.** The WSOD provides for additional uses through a special permit process. All underlying uses remain available to property owners. If a property owner elects to apply for a special permit to develop in accordance with the WSOD, the provisions of this Section shall apply. All underlying zoning district requirements still apply, unless they are superseded by this Section. By filing an application for a special permit under this Section, the owner shall be deemed to accept and agree to the terms and provisions set forth herein.

If the owner elects to proceed under the zoning provisions of the underlying district, the zoning By-laws applicable in the underlying district shall control and the provisions of the Willow Street Overlay District shall not apply.

- **9.5.4 Permitted Uses.** In addition to the uses provided in the underlying zoning district, the following additional uses shall be permitted in the WSOD:
 - 1. Athletic facilities;
 - 2. Accessory infrastructure facilities;
 - 3. Retail or Restaurant;
 - 4. Multi-family residential;
 - 5. Mixed residential and commercial:
 - 6. Business or Professional Office:
 - 7. Ancillary parking and commercial services that serve the primary uses; and
 - 8. Public spaces.
- **9.5.5** Administration. The Planning Board shall be the SPGA in the WSOD. An applicant may file an application with the Planning Board in conformance with this Section and Planning Board WSOD Rules and Regulations.
- **9.5.6 Planning Board Rules and Regulations.** The Planning Board shall adopt and from time to time may amend Willow Street Overlay District Rules and Regulations to implement this

Section. Such regulations shall include but will not be limited to submission requirements, plan requirements, such as size, form, number and contents; development standards, site standards, and standards for building placement and design. Such rules and regulations are required and/or authorized under G.L. c. 40A, s. 9, and shall be adopted after vote by the Planning Board.

- 9.5.7 Pre-Application Conference. The applicant shall request and attend a pre-application review at a regular business meeting of the Planning Board. The Planning Board shall invite the Board of Health, Board of Selectmen, Building Commissioner, Department of Public Works, Fire Chief, Police Chief, Zoning Board of Appeals, and Economic Development Committee to attend. The primary purpose of the pre-application conference is to provide an information exchange and feedback regarding how the applicant intends to use the site. The secondary purposes of a pre-application review are to minimize the applicant's cost of engineering and other technical experts, to commence discussions with the Planning Board at the earliest possible stage of development, and to set a timetable for submittal of a formal application. Meetings may be held by mutual agreement of the Planning Board and the applicant.
- 9.5.8 Review Procedures for WSOD Special Permit. The applicant shall furnish a copy of the application to the Town Clerk. The applicant shall furnish sufficient copies of the special permit application to the Planning Board to distribute for review to the Board of Health, Board of Selectmen, Building Commissioner, Department of Public Works, Fire Chief, Police Chief, Zoning Board of Appeals, and Economic Development Committee. Reports from the abovenamed boards and officials shall be submitted to the Planning Board within thirty (30) days of receipt by the reviewing party. In the event that the public hearing by the Planning Board is held prior to the expiration of the thirty (30) day period, the Planning Board shall continue the public hearing to permit the formal submission of reports and recommendations within that thirty (30) day period.
 - 1. Planning Board members shall conduct a site visit with the applicant prior to or during the public hearing.
 - 2. Once a WSOD Special Permit application has been submitted, no tree removal, no utility installation, no ditching, grading or construction of roads, no grading of land or lots, no excavation, except for purposes of soil testing, dredging or filling and no construction of Buildings or Structures shall be done on any part of the site until the application has been reviewed and decided as provided by this Section.
- **9.5.9 General Requirements.** A proposal must comply with all other applicable Town Bylaws, and the applicable rules, regulations, and requirements of all departments, boards, and commissions.
 - 1. Low Impact Development. The use of low-impact development techniques is required, where applicable. The applicant shall employ meaningful low impact techniques which will result in less impervious area, direction of roof runoff toward rain gardens and swales, and plantings not limited to but including those indigenous to the area. The use of recycled or recaptured rainwater is encouraged.

- 2. Minimizing Disturbance. The applicant is urged to incorporate horticultural and landscape design that reduces the need for supplemental irrigation and chemical fertilization, e.g. minimizing lawn area. Streets within the site shall be designed and located in such a manner as to maintain and preserve natural topography, significant landmarks and trees; and to minimize cut and fill. Buildings shall be designed to relate harmoniously to the terrain.
- 3. Location of Wastewater Treatment. All wastewater shall be treated and discharged on-site. Off-site wastewater treatment may be allowed if a suitable tie in can occur with an available sewer system.
- 4. Paths and Trails. All existing paths and trails shall be connected, preserved, and increased when possible.
- 5. Visual Appearance from Offsite. The orientation of individual Buildings and Structures shall be designed to preserve and enhance views and vistas on or off the subject property.
- 6. *Inclusionary By-law*. In the case that proposed developments in the WSOD include multi-family units such as apartments or condominiums then the developments must comply with the Section 8.3, Inclusionary Housing, with regard to the residential component of the projects.
- 7. Sidewalks. Sidewalks will be created between the Buildings and the Street wherever possible.

9.5.10 Building and Design Standards. The following standards shall apply:

- 1. Setbacks. The setbacks of the underlying zone shall apply.
 - a. Internal Building Separation: the minimum distance between all freestanding Buildings in the WSOD shall be free of any obstruction which would interfere with the passage of a fire-safety vehicle to the complete site; and
 - b. The setback for height at the property line shall be no more than fifteen (15) feet high from the property line for a setback distance of at least ten (10) feet.
- 2. Height. Provided the use of the Building is either non residential or Mixed Use, building heights may be increased in the discretion of the Planning Board. Current zoning requires heights not to exceed thirty five (35) feet, except that spires, water tanks, communication towers, chimneys, exhaust stacks, flagpoles, mechanical penthouses and other structures normally built above the roof and not devoted to human occupancy may be erected to such heights as are necessary to accomplish the purpose they are normally intended to serve. In the WSOD, the Planning Board may, in its sole discretion, allow building height for nonresidential or mixed properties to be increased.

3. *Parking*. Within the WSOD District, off-street parking shall be provided sufficient to serve the needs of the various uses, based upon the nature of the use and the number of persons occupying and using the facilities and shall be in compliance with the parking requirements of the underlying business district unless the applicant can prove that mixed-use shared parking between the residential and commercial uses is appropriate.

Parking areas shall: (1) be located so as to serve the uses to which they are related; (2) be designed to serve more than one use or facility; (3) be located and laid out so as to provide turning radii, sight lines, and separation from through traffic, consistent with public safety; (4) be paved or, if approved by the Planning Board, surfaced with other non-dusting material; and, (5) unless otherwise approved by the Planning Board, be striped with spaces of at least eight (8) feet width and nineteen (19) feet length.

- 4. Lighting. All lighting fixtures shall be integrated into the architectural style of the development. All exterior structural and site lighting (not including access road lighting) shall be localized and of full cut-off design, with light which shall be retained on site and shall not create a nuisance to abutting properties and streets. All exterior light sources shall be appropriately shielded from off-premise viewing. Access road lighting shall be designed to prevent nuisance to abutting properties by reason of light and glare.
- 5. *Signs*. The Business District provisions of Section 6.3 shall apply for all building specific signage (tenants' identification, directory, parking, etc.) within the Willow Street Overlay District, subject to the following:
 - a. Site signage for developments with multiple structures within the Willow Street Overlay District. Site signage at the main entrance to the entire project for identification of the project is permitted, to be located on free standing structure. Such sign shall not exceed thirty-six (36) square feet in size and shall be mounted a maximum of two (2) feet to the underside of the sign and a maximum of ten (10) feet to the top of the sign measured from the average grade of the locus of the sign;
 - b. A single project tenant directory sign is permitted, to be located on a free standing structure. Such directory sign shall for each of the WSOD's commercial buildings' main tenants not exceed thirty-six (36) square feet in size and shall be mounted a maximum of two (2) feet to the underside of the sign and a maximum of ten (10) feet to the top of the sign measured from the average grade of the locus of the sign;
 - c. Site directional signage is allowed provided each such sign may be no larger than three (3) square feet in size. Such signs may be posted not closer than twenty (25) feet on center; and
 - d. No other additional signs or increases in sign dimensions are allowed except by special permit from the Planning Board.

- **9.5.11 Criteria.** The Planning Board may grant a special permit pursuant to Section 10.5.2, and after considering each of the following additional criteria:
 - 1. The degree to which the conceptual design and layout of the proposed development is projected to attain the purpose of generating positive net tax revenue and local employment;
 - 2. The degree to which the proposed development protects the local neighborhood;
 - 3. The degree to which the proposed development furthers the goals and policies of the Master Plan; and
 - 4. The degree to which the proposed development furthers the purposes of this Section.
- **9.5.12 Conditions.** The Planning Board may impose certain conditions, but shall not be limited to, safeguards, requirements, and other standards to:
 - 1. Mitigate impact to abutting land and natural resources by reason of air or water pollution, noise, dust, vibration, or stormwater runoff;
 - 2. Provide safe and convenient access to the site from existing or proposed roads, and to proposed structures thereon, with particular reference to pedestrian and vehicular safety, traffic flow and control, and access in case of fire or emergency;
 - 3. Provide for adequate capacity for public services, facilities, and utilities to service the proposed development such as water pressure and septic/sewer capacity;
 - 4. Provide for visual and noise buffering of the development to minimize impact to abutting properties; and
 - 5. Provide for the perpetual preservation and maintenance of any open space and recreation areas.
- **9.5.13** Lapse. A Willow Street Overlay District special permit shall lapse if (a) at least one (1) building permit has not been issued within two (2) years from the date of the approval of the special permit, except for good cause shown, (b) such a building permit has been issued and has expired, or (c) construction has commenced under the special permit but does not proceed continuously to completion.

9.6 COMMERCIAL OVERLAY DISTRICT (COD).

9.6.1 Purpose. The general purpose of the Commercial Overlay District (COD) is to encourage planned and orderly development of land within the boundaries of the COD. These areas are designated for special land use, planning, environmental, and access considerations that warrant careful control of development under a special permit procedure using appropriate standards. The specific purposes of the COD are the following:

- 1. To support land uses and activities that will promote economic development in Hamilton, thereby generating non-residential tax revenue from new business growth and additional local jobs;
- 2. To allow for the phased development of new uses and activities that are appropriate and desirable for the COD, based on planning and market opportunities identified in the Hamilton Master Plan, and consistent with the goals and objectives in the Hamilton Open Space Plan;
- 3. To provide for mitigation of any negative impacts caused by new development, including but not limited to transportation and environmental impacts;
- 4. To incorporate into the design review and permitting process input from Hamilton's land use boards, nearby residential abutters and citizens; and
- 5. To incorporate into the design review and permitting process the special provisions necessary to ensure that redevelopment of a former landfill site will not degrade important regional water resources.
- **9.6.2** Overlay District. The COD is an overlay district overlaying other districts. In the COD, all requirements of the underlying district shall remain in effect except where this Section provides an alternative to such requirements, in which case these regulations shall supersede those of the underlying district(s). In the event that an applicant wishes to develop in accordance with the COD, the rules and regulations of the COD shall apply, and if required, by filing an application for a special permit under this Section, the owner shall be deemed to accept and agree to them. In such event, where the provisions of the COD are silent on a zoning regulation that applies in the underlying district, the requirements of the underlying district shall apply.

If the proponent elects to proceed under the zoning provisions of the underlying district, the zoning By-laws applicable in the underlying district shall control and the provisions of the COD shall not apply.

- **9.6.3 Permitted Uses By Right in the COD.** Wind Energy Facilities shall be permitted by right subject to Site Plan Review under Section 10.6.
- **9.6.4** Uses Permitted by Special Permit in the COD. The following uses shall be permitted only by grant of a special permit from the Planning Board in the COD:
 - 1. Research and development;
 - 2. Light manufacturing and assembly;
 - 3. Business or Professional Office;
 - 4. Medical Clinic or Medical Office;

- 5. Life sciences:
- 6. Biotechnology;
- 7. Ancillary retail and commercial services that serve the primary uses;
- 8. Public spaces;
- 9. Athletic facilities or outdoor commercial recreation;
- 10. Accessory infrastructure facilities; and
- 11. Communication Towers and Telecommunications Antenna Facilities. The COD shall supersede Section 7.2.
- **9.6.5** Administration. The Planning Board shall be the SPGA (Special Permit Granting Authority) and Site Plan Review Authority in the COD. An applicant shall file an application with the Planning Board in conformance with this Section, and the Planning Board's COD Rules and Regulations.
- **9.6.6 Planning Board Rules and Regulations.** The Planning Board shall adopt and from time to time may amend Commercial Overlay District Rules and Regulations to implement this Bylaw section. Such regulations shall include but will not be limited to submission requirements, plan requirements, such as size, form, number and contents; development standards, site standards, and standards for Building placement and design. Such rules and regulations are authorized under G.L. C. 40A, s. 9, and shall be adopted after vote by the Planning Board.
- 9.6.7 Pre-Application Conference. The applicant shall request and attend a pre-application review at a regular business meeting of the Planning Board. The Planning Board shall invite the Board of Health, Board of Selectmen, Building Commissioner, Conservation Commission and Open Space Committee, Department of Public Works, Fire Chief, Police Chief, Zoning Board of Appeals, Economic Development Committee, and Landfill Steering Committee to attend. The primary purpose of the pre-application conference is to provide an information exchange and feedback regarding how the applicant intends to use the site. The secondary purposes of a pre-application review are to minimize the applicant's cost of engineering and other technical experts, to commence discussions with the Planning Board at the earliest possible stage of development, and to set a timetable for submittal of a formal application. Meetings may be held by mutual agreement of the Planning Board and the applicant.
- **9.6.8 Review Procedures for COD Special Permit.** The applicant shall furnish a copy of the application to the Town Clerk. The applicant shall furnish sufficient copies of the special permit application to the Planning Board to distribute for review to the Board of Health, Board of Selectmen, Building Commissioner, Conservation Commission, Open Space Committee, Department of Public Works, Fire Chief, Police Chief, Zoning Board of Appeals, Economic Development Committee, and Landfill Steering Committee. Reports from the above-named

boards and officials shall be submitted to the Planning Board within thirty (30) days of receipt by the reviewing party. In the event that the public hearing by the Planning Board is held prior to the expiration of the thirty (30) day period, the Planning Board shall continue the public hearing to permit the formal submission of reports and recommendations within that thirty (30) day period. Planning Board members shall conduct a site visit with the applicant prior to or during the public hearing.

- **9.6.9 Prohibition.** Once a COD Special Permit application has been submitted, no tree removal, no utility installation, no ditching, grading or construction of roads, no grading of land or lots, no excavation, except for purposes of soil testing, dredging or filling and no construction of Buildings or Structures shall be done on any part of the site until the application has been reviewed and decided as provided by these regulations.
- **9.6.10 General Compliance.** A proposal must comply with all other applicable Town By-laws, and the applicable rules, regulations, and requirements of all departments, boards, and commissions.

9.6.11 General Requirements.

- 1. Low Impact Development. The use of low-impact development techniques is required, where applicable. The applicant shall employ meaningful low impact techniques which will result in less impervious area, direction of roof runoff toward rain gardens and swales, and plantings not limited to but including those indigenous to the area. The use of recycled or recaptured rainwater is encouraged.
- 2. *Minimizing Disturbance*. The applicant is encouraged to maintain as much of the site as possible in its natural state. The applicant is urged to incorporate horticultural and landscape design that reduces the need for supplemental irrigation and chemical fertilization, e.g. minimizing lawn area. Streets within the site shall be designed and located in such a manner as to maintain and preserve natural topography, significant landmarks and trees; to minimize cut and fill. Buildings shall be designed to relate harmoniously to the terrain.
- 3. Location of Wastewater Treatment. All wastewater shall be treated and discharged on-site. Off-site wastewater treatment may be allowed if a suitable tie in can occur with an available sewer system.
- 4. Paths and Trails. All paths and trails shall be connected, preserved, and increased when possible.
- 5. Visual Appearance from Offsite. The orientation of individual Buildings and Structures shall be designed to preserve and enhance views and vistas on or off the subject property.

9.6.12 Building and Design Standards. The following standards shall apply:

- 1. Setbacks. All structures shall be so located as to be at least twenty-five (25) feet from the perimeter of the COD.
- 2. Internal Building Separation. The minimum distance between all freestanding Buildings in the COD shall be at least twenty (20) feet, free of any obstruction which would interfere with the passage of a fire safety vehicle.
- 3. Height. Buildings located in the COD shall not exceed fifty (50) feet in Building Height, except that spires, water tanks, communication towers, chimneys, exhaust stacks, flagpoles, mechanical penthouses and other structures normally built above the roof and not devoted to human occupancy may be erected to such heights as are necessary to accomplish the purpose they are normally intended to serve. Buildings located in excess of two hundred (200) feet from the lot lines of the COD may at the Planning Board's discretion be up to sixty (60) feet in Building Height, as measured above.
- 4. *Parking*. Within the COD District off-street parking shall be provided sufficient to serve the needs of the various uses, based upon the nature of the use and the number of persons occupying and using the facilities.
 - a. With respect to parking for all uses, with particular attention for the office, research, development and manufacturing, medical, life sciences, and biotechnology uses, parking shall be required in amounts that conform with the latest use-specific parking recommendations published by the Institute of Traffic Engineers (ITE).
 - b. Parking areas shall be located so as to serve the uses to which they are related, and may be designed to serve more than one use or facility; shall be located and laid out so as to provide turning radii, sight lines, and separation from through traffic, consistent with public safety; shall be paved or, if approved by the Planning Board, surfaced with other non-dusting material; and shall, unless otherwise approved by the Planning Board, be striped with spaces of at least nine (9) feet width and nineteen (19) feet length.
- 5. Lighting. All lighting fixtures shall be integrated into the architectural style of the development. All exterior structural and site lighting (not including access road lighting) shall be localized and of full cut-off design, with light which shall be retained on site and shall not create a nuisance to abutting properties and streets. All exterior light sources shall be appropriately shielded from off-premise viewing. Access road lighting shall be designed to prevent nuisance to abutting properties by reason of light and glare.
- 6. Water Supply. The project shall use either the Town of Hamilton and/or the Town of Manchester (if available and acceptable to the Town of Manchester) water supply and shall comply with any inter-town agreements that govern water access rights.

- 7. Buffer Areas. A vegetated buffer area of twenty-five (25) feet minimum may be required at the following locations:
 - a. Perimeter of the property where it abuts a lot in residential use;
 - b. Certain resource areas on or adjacent to the tract such as rock outcrops, ledge, agricultural or recreational fields, and land held for conservation purposes;
 - c. A landscape plan will be required and shall address buffers to ensure screening from abutting properties. Additions to existing plantings may be required by the Planning Board; and
 - d. The Planning Board may waive the buffer requirement when it determines that a small buffer (or no buffer) will suffice to accomplish the objective set forth herein.
- 8. Site Access, Roadway Design and Management. All Structures located within the Commercial Overlay District development shall be accessed by the project's proposed private access road. No access to or from the site will be permitted to the abutting residential area to the north along Chebacco Road. All access roadways and associated infrastructure improvements shall be private and shall not be owned or maintained by the Town of Hamilton.
- 9. *Traffic Study*. A traffic study shall be provided by the applicant to ensure public safety on and around Chebacco Road and to address the adequacy of the internal road and parking infrastructure.
- 10. Stormwater Management and Erosion Control. The peak rate of stormwater runoff from the Commercial Overlay District shall comply with Department of Environmental Protection's Stormwater Management Handbook, and the provisions General By-law XXIX, Stormwater Management By-law.
- 11. Soil Erosion. General soil erosion of the proposed development site shall be minimized by integrating the development into the existing terrain and by reasonably retaining natural grades and soil cover. During grading and construction of all improvements, including all Structures and infrastructure improvements, erosion of soil shall be minimized using best management practices.
- 12. Open Space and Natural Resources. A Commercial Overlay District development which includes any open space components must provide that land area as permanently protected, usable, common open space that is functional for purposes intended by this Section. The common open space shall have no Structures, parking, private Yards, patios, or gardens that are restricted for the exclusive or principal use by the tenants or users of the commercial properties.

- **9.6.13** Signs. The Business District provisions of Section 6.3 shall apply for all Building specific signage (tenant identification, directory, parking, etc.) within the Commercial Overlay District.
 - 1. Site signage at the main entrance to the entire project for identification of the Project is permitted, to be located on free standing Structure. Such sign shall not exceed thirty-six (36) square feet in size and shall be mounted a maximum of two (2) feet to the underside of the sign and a maximum of ten (10) feet to the top of the sign measured from the average grade of the locus of the sign.
 - 2. A single project tenant directory sign is permitted, to be located on a free standing Structure. Such directory sign shall for each of the COD's commercial buildings main tenants not exceed thirty (36) square feet in size and shall be mounted a maximum of two (2) feet to the underside of the sign and a maximum of ten (10) feet to the top of the sign measured from the average grade of the locus of the sign.
 - 3. Site directional signage is allowed provided each such sign may be no larger than three (3) square feet in size; such signs may be posted not closer than twenty (25) feet on center.
 - 4. No other additional signs or increases in sign dimensions are allowed except by special permit from the Planning Board.
- **9.6.14 Conditions of Approval.** The Planning Board may impose conditions, safeguards, requirements, and other standards as part of its approval.
- **9.6.15** Findings and Decision of the Planning Board for COD Special Permit. The Planning Board may grant a special permit pursuant to Section 10.5.2 and where the proposed development includes satisfactory provisions to:
 - 1. Mitigate impact to abutting land and natural resources by reason of air or water pollution, noise, dust, vibration, or stormwater runoff;
 - 2. Provide safe and convenient access to the site from existing or proposed roads, and to proposed structures thereon, with particular reference to pedestrian and vehicular safety, traffic flow and control, and access in case of fire or emergency;
 - 3. Provide for adequate capacity for public services, facilities, and utilities to service the proposed development such as water pressure and septic/sewer capacity;
 - 4. Provide for visual and noise buffering of the development to minimize impact to abutting properties;
 - 5. Provide for the perpetual preservation and maintenance of any open space and recreation areas;

- 6. Protect regional water resources;
- 7. Promote permanent preservation of open space, agricultural land, forestry land, existing and proposed trails, other natural resources including water bodies and wetlands; and
- 8. Further the goals and policies of the Master Plan and the purposes of this Section.
- **9.6.16** Lapse. A Commercial Overlay District special permit shall lapse if at least one (1) building permit has not been issued by two (2) years from the date of approval, except for good cause shown, (b) such a building permit has been issued and has expired, or (c) construction has commenced under the special permit but does not proceed continuously to completion.

SECTION 10.0 ADMINISTRATION AND ENFORCEMENT

10.1 ADMINISTRATION.

- **10.1.1 Building Commissioner.** The provisions of this By-law shall be administered and enforced by the Building Commissioner.
- 10.1.2 Building Permit Required. It shall be unlawful to construct, alter, reconstruct or relocate any Building or Structure or to institute a new or altered use of a Building, Structure or land without first obtaining a building permit from the Building Commissioner. A building permit shall expire one hundred eighty (180) days from its date of issue unless exercised. A special permit and/or site plan approval under Section 10.5 and/or 10.6 may be required for some uses prior to building permit issuance.
- 10.1.3 Application for Permit. Any application for a permit shall be accompanied by a specific reference to the subject Lot or group of Lots in the same ownership as recorded in the Registry of Deeds, and by copies of a plan of the proposed Lot, drawn to scale, showing the entire recorded ownership, all existing Structures, all abutting Streets, and the exact area and boundaries of the parcel to be assigned to the subject use, and the proposed outline and location of each and every proposed Structure on the subject Lot, and the location and length of any proposed Driveway from the access Street to the building site. There shall also be shown on said plan the existing or proposed sewerage system and all distances necessary to establish full compliance with this By-law, including Building Frontage, areas proposed to be occupied by Buildings or Structures, Yards and sideline distances. The accuracy of said plan and the information thereon shall be certified by the applicant and shall be staked out on the land and left in place until the final inspections are made.
- 10.1.4 Approval of Applications. The Building Commissioner shall not issue a permit for the construction, alteration, reconstruction or relocation of a Building or Structure if the Building or Structure as constructed, altered, reconstructed or relocated would be in violation of this By-law, or for a new or altered use of a Building, Structure or land if the use would be in violation of this

By-law, unless the applicant has secured a special permit or site plan approval or variance from the Board of Appeals or Planning Board, as designated herein.

10.1.5 Occupancy Permit. No Building erected, altered, or in any way changed as to construction or use, under a permit or otherwise, shall be occupied or used without an occupancy permit, signed by the Building Commissioner, which permit shall not be issued until the Building and its uses, and the uses incident thereto, comply in all respects with this By-law.

10.2 ENFORCEMENT AND PENALTIES.

- **10.2.1 Enforcement.** If the Building Commissioner is requested in writing to enforce this Bylaw against any person allegedly in violation of this By-law, and the Building Commissioner declines to act, he shall notify, in writing, the party requesting such enforcement of any action or refusal to act, and the reasons therefore, within fourteen (14) days of receipt of such request.
- 10.2.2 Appeal. Any person aggrieved by his inability to obtain a permit or enforcement action from the Building Commissioner and any person, including an officer or board of the Town, aggrieved by an order, decision or failure to act of the Building Commissioner or other official in violation of this By-law or Chapter 40A of the Massachusetts General Laws, may appeal to the Zoning Board of Appeals by filing a notice of appeal with the Town Clerk within thirty (30) days of the date of the action, order, or decision.

10.2.3 Violations and Penalty.

- 1. Whoever violates any provision of this By-law may be punished by a fine not exceeding three hundred (300) dollars for each offense. Each day or portion thereof that such violation continues shall constitute a separate offense.
- 2. Pursuant to G.L. c. 40, s. 21D, this By-law may also be enforced by the Building Commissioner by noncriminal complaint. Each day or portion thereof that such violation continues shall constitute a separate offense. The penalty for violation of any provision of this By-law shall be twenty-five (25) dollars for the first offense; fifty (50) dollars for the second offense; one hundred (100) dollars for the third offense; and two hundred (200) dollars for the fourth offense and each subsequent offense.

10.3 ZONING BOARD OF APPEALS

- **10.3.1 Composition of Board.** A Zoning Board of Appeals consisting of three (3) members and two (2) associate members, all citizens of the Town, shall be appointed by the Board of Selectmen, and shall act as the Zoning Board of Appeals under this By-law and under Massachusetts General Laws, Chapter 40A, Section 12.
- **10.3.2 Duties.** The Zoning Board of Appeals shall have all the powers and perform all of the duties conferred or imposed upon it under the applicable provisions of the General Laws of the Commonwealth of Massachusetts, and shall serve as the permit granting authority when so designated.

- 10.3.3 Powers. The Zoning Board of Appeals shall have the following powers:
 - 1. To hear and decide administrative appeals in accordance with G.L. c. 40A, ss. 8 and 15;
 - 2. To hear and decide applications to alter a nonconforming use or structure.
 - 3. To hear and decide applications for special permits as provided in this By-law.
 - 4. To grant upon appeal or upon petition with respect to particular land or structures a variance from the terms of the applicable zoning ordinance or By-law where such permit granting authority specifically finds that owing to circumstances relating to the soil conditions, shape, or topography of such land or Structures and especially affecting such land or Structures but not affecting generally the zoning district in which it is located, a literal enforcement of the provisions of the ordinance or this By-law would involve substantial hardship, financial or otherwise, to the petitioner or appellant, and that desirable relief may be granted without substantial detriment to the public good and without nullifying or substantially derogating from the intent or purpose of such ordinance or this By-law. The Board of Appeals shall not grant a use variance.
- 10.3.4 Regulations; Fees. The Board of Appeals shall adopt regulations, including fees, not inconsistent with this By-law as required by G.L. c. 40A, s. 12.
- 10.3.5 Public Notice. Wherever proceedings under this By-law require the giving of notice by publication in a newspaper, mailing or service by a civil officer, the costs thereof shall be borne by the applicant, and the Board of Appeals shall require estimated costs to be advanced by the applicant.

10.4 PLANNING BOARD.

- **10.4.1 Establishment.** The Planning Board shall consist of seven (7) elected regular members and two (2) associate members, all citizens of the Town.
- **10.4.2 Powers.** The Planning Board shall have the following powers:
 - 1. To hear and decide applications for special permits as provided in this By-law;
 - 2. To hear and decide applications for site plan approval as provided in Section 10.6 of this By-law;
 - 3. To conduct other business as provided for by Massachusetts law and this By-law.
- 10.4.3 Rules and Regulations; Fees. The Planning Board shall adopt rules and regulations, including fees, for the conduct of business and for the purpose of this By-law and Chapter 40A of the General Laws, and shall file a copy of such rules and fees with the Town Clerk.

- 10.4.4 Associate Members of the Planning Board. The position of Associate Member of the Planning Board is established for applications where the Planning Board has been designated as the SPGA. Two (2) Associate Members shall be appointed jointly by the Planning Board and the Board of Selectmen. The terms of appointment shall be two (2) years. The Associate Members shall be designated, in order of seniority, to act as provided in G.L. c. 40A, § 9.
- **10.4.5 Public Notice.** Wherever proceedings under this By-law require the giving of notice by publication in a newspaper, mailing or service by a civil officer, the costs thereof shall be borne by the applicant, and the Planning Board shall require estimated costs to be advanced by the applicant.

10.5 SPECIAL PERMITS

- 10.5.1 Special Permit Granting Authority. The Zoning Board of Appeals, Planning Board, and the Board of Selectmen shall be the SPGA as specified in the various sections of this By-law and shall hear and decide applications for special permits. Those uses requiring a special permit and the SPGA assigned to review the application are set forth in the Table of Use Regulations.
- 10.5.2 Criteria. Special permits shall be granted by the SPGA, unless otherwise specified herein, only upon its written determination that the adverse effects of the proposed use will not outweigh its beneficial impacts to the town or the neighborhood, in view of the particular characteristics of the site, and of the proposal in relation to that site. In addition to any specific requirements and factors that may be set forth in this By-law, the determination shall include consideration of each of the following:
 - 1. Social, economic, or community needs which are served by the proposal;
 - 2. Traffic flow and safety, including parking and loading;
 - 3. Adequacy of utilities and other public services;
 - 4. Neighborhood character and social structures;
 - 5. Impacts on the natural environment; and
 - 6. Potential fiscal impact, including impact on town services, tax base, property values in the same or adjoining districts, and employment taking into account any proposed mitigation.
- 10.5.3 Application. The SPGA may adopt additional rules relative to the issuance of special permits and shall file a copy with the Town Clerk. The rules shall prescribe a size, form, contents, style, and number of copies of plans and specifications and the procedure for submission and approval of such permits. The application shall be subject to any general or specific rules contained in this By-law and subject to any appropriate conditions, safeguards and limitations imposed by the SPGA. A special permit shall be issued only following public

hearing held within sixty-five (65) days after filing of an application with the Town Clerk, who shall transmit a copy thereof to the SPGA forthwith.

- **10.5.4 Conditions.** The SPGA may impose additional conditions and limitations as it may deem necessary.
- **10.5.5** Regulations. The SPGA may adopt rules and regulations for the administration of this Section.
- 10.5.6 Fees. The SPGA may adopt reasonable administrative fees and technical review fees for applications for special permits.
- 10.5.7 Lapse. Except as otherwise expressly provided in this By-law, special permits shall lapse if a substantial use thereof or construction thereunder has not begun, except for good cause, within twenty-four (24) months following the filing of the special permit approval (plus such time required to pursue or await the determination of an appeal referred to in G.L. c. 40A, s. 17, from the grant thereof) with the Town Clerk.

10.6 SITE PLAN REVIEW.

- 10.6.1 Purpose. Site plan review is intended to allow the Town to review and impose reasonable conditions on development, allowed as of right or by special permit, which by its nature or location has the potential for significant impacts on traffic circulation, public safety, public utilities, the environment and neighborhood character and appearance. The purpose is to provide a mechanism and a process for fitting the proposed development to neighborhood character, scale, and infrastructure, and to reasonably protect the Town and abutting property owners from potential negative impacts. These purposes are to be served by providing for consistent submissions of thorough data for each proposal for Town review; and for consideration by Town boards as specified below, in terms of specific criteria; and for issuance of a site plan decision setting whatever terms and conditions, if any, are necessary to achieve the purposes of site plan review, consistent with Town By-laws and regulations. Provision is made to exempt small modifications and improvements from portions of this Section, and to coordinate site plan review with other required review of special permits and alteration of nonconforming uses and structures.
- **10.6.2 Projects Requiring Approval.** No building permit shall be applied for or issued for any construction or alteration subject to this Section, as specified below, until a site plan has been approved or conditionally approved by the Planning Board as set forth herein. Except as exempted in the By-law, site plan review shall be required for the following construction, alterations, and uses:
 - 1. Any new construction or alteration or expansion of, or conversion to a business, office, industrial, institutional, or multi-family residence Building or Structure in any district;

- 2. Any alteration or extension of a nonconforming commercial, industrial or institutional structure in a Residential zone;
- 3. Parking lots over five (5) spaces serving any of the above-listed types of Buildings, schools, hospitals, nursing homes, long term care facility, life care facilities, children's camps, campgrounds, churches, multi-family structures for more than two (2) Families, government buildings and structures, farm stands, and Buildings in commercial, and/or office use;
- 4. Industrial and scientific research use under Section 3.2;
- 5. Stables providing boarding and/or services for more than six (6) horses other than those owned by the residents of the Lot on which the stable is located;
- 6. Greenhouses for commercial production of plants or produce;
- 7. Interior alterations which increase the commercial, industrial, institutional or multifamily floor area within an existing Building, and/or which change the number of separately leasable or saleable spaces within an existing Building; and
- 8. Wind Energy Facilities in the Commercial Overlay District.
- 10.6.3 Projects Requiring Abbreviated Site Plan Review. Proposed additions or alterations to existing Buildings subject to site plan review which would either add new floor area totaling less than ten (10) percent of the current ground floor area of the existing Building, or which are estimated to cost less than ten (10) percent of the current one hundred (100) percent assessed yaluation of the Building shall require only abbreviated site plan review as provided below. For purposes of applying this guideline, when ground floor area includes attached residential floor area or attached commercial greenhouses, the area of the residential and/or greenhouse portions shall be deducted before applying the ten (10) percent threshold factors.
- **10.6.4** Exemptions from Site Plan Review. The following activities shall not require site plan review or abbreviated site plan review:
 - 1. Interior alterations unless the alterations increase the floor area within the building devoted to any of the uses listed in Section 10.6.2 and/or change the number of separately leasable or saleable spaces within the Building. Examples of exempt interior alterations are: paneling, shelving, partitions other than those creating new commercial units, counters, flooring and general interior decoration;
 - 2. Repairs, repainting, residing, reroofing and window replacement unless the activity also increases floor area;
 - 3. Replacement of parts of a partially damaged Building that would normally require site plan review with generally identical portions. Total replacement of such a destroyed Building shall require site plan review;

- 4. Demolition of a Building or Structure;
- 5. Change of tenant of commercial building unless there is also new construction triggering site plan review;
- 6. Customary home occupations;
- 7. Townhouse or Multifamily housing developed under Sections 8.1 (OSFPD) or 8.2 (Senior Housing);
- 8. Uses permitted by special permit within the Commercial Overlay District; and
- 9. Entries B.1, B.3, and B.7 in the Table of Uses shall require site plan review under Section 10.7.
- 10.6.5 Site Plan Review; Procedures. The applicant shall file twenty-four (24) copies of all site plan review materials as specified below, accompanied by a fee and the documentation required by the Planning Board Instruction Sheet. The application shall be submitted to the Town Clerk. The application will not be considered legally submitted, for purposes of starting the review "clock", until it is complete and all copies are submitted.
 - 1. The Town Clerk shall thereafter forward ten (10) copies to the Planning Board, and one (1) copy each to the Building Commissioner, Department of Public Works, Board of Selectmen, Board of Health, Conservation Commission, Police Department, Fire Department, and Office on Disability for their comments, retaining one copy for the Town Clerk file.
 - 2. An electronic copy of the application, including plans and drawings, shall also be submitted at the time of application, unless waived by the Planning Board.
 - 3. The applicant is encouraged to meet informally with any of the above-listed boards, officials, or departments for their recommendation.
 - 4. The above-listed boards, officials, or departments shall transmit their comments in writing to the Planning Board within thirty-five (35) days from the date of application. Failure to submit comments shall be deemed a lack of opposition to the project.
- 10.6.6 Site Plan Review; Hearing and Decision. The Planning Board shall open a public hearing within sixty five (65) days of the date of an application. The Planning Board shall provide notice as set forth in G.L. c. 40A, s. 11. The Planning Board shall make its determination within sixty-five (65) days of the closing of the public hearing, unless the applicant specifically grants an extension of the review period in writing. Failure to decide within sixty-five (65) days of the closing of the public hearing shall constitute approval of the site plan by the Planning Board unless the applicant granted an extension of the review period.

- 1. When proposed new construction or alteration is subject to both site plan review and a Planning Board special permit, the Planning Board shall conduct both processes simultaneously, using the time limits of the special permit process to cover both reviews.
- 10.6.7 Abbreviated Site Plan Review; Procedures. If the proposed construction and/or change of use is eligible for abbreviated site plan review under Section 10.6.3, the applicant shall file eight (8) copies of the materials required below with the Town Clerk, along with the fee and accompanying documents as required by the Planning Board Instruction Sheet. The application will not be considered legally submitted until it is complete.
 - 1. The Town Clerk shall forward five (5) copies to the Planning Board, and one (1) copy to the Building Commissioner, retaining one copy for the Town Clerk file.
 - 2. The above-listed boards or officials shall transmit their comments in writing to the Planning Board within thirty (30) days from the date of application. Failure to submit comments shall be deemed a lack of opposition to the project.
- 10.6.8 Abbreviated Site Plan Review; Hearing and Decision. The procedures set forth in Section 10.6.6 shall apply.
- 10.6.9 Site Plan Review; Contents. Unless waived by the Planning Board, all site plans shall be prepared to scale and be of professional quality. The application shall be prepared by a Registered Architect or Professional Engineer for any proposal that requires such certification under the State Building Code. All site plans shall be standard seventeen (17) inch by twenty-four (24) inch or twenty-four (24) inches by thirty-six (36) inch sheets and shall be prepared at a sufficient scale to show the following required items:
 - 1. Location and boundaries of existing (and proposed) Lot, adjacent Streets or ways, including rights-of way and easements, and the location and owner names of all adjacent properties;
 - 2. Size of the Lot, Lot Frontage and Yards;
 - 3. All existing and proposed Buildings and Structures, including dimensions;
 - 4. Elevation drawings (architectural appearance) of the Building or Building with addition, from each side. For additions/alterations, show only the affected sides;
 - 5. Location and dimensions of all parking and loading areas, Driveways, walkways, access and egress points, specifically as they relate to Street traffic; number of parking spaces, with note comparing total to the requirements of Section 6.1; and indication of facilities for handicapped access per the requirements of the Architectural Access Board;
 - 6. If the project includes any grading and site work, topography of the site at two (2) foot intervals based on USGS data, showing both existing and proposed contours;

- 7. Location and description of all existing and proposed sewerage disposal and treatment systems, and underground storage tanks, water supply, storm drainage systems, utilities, and refuse disposal dumpsters;
- 8. Location, height, and type of all external lighting;
- 9. Location, height, dimensions and appearance of signs;
- 10. Proposed landscaping including the location and description of existing large trees;
- 11. Proposed screening, buffers, fencing, plantings, open spaces and recreation areas; and
- 12. Statement of any other local and/or state permits required for the project.

10.6.10 Site Plan Review; Other Materials. In addition to the site plan, the applicant shall submit the following materials:

- 1. Completed Site Plan Application form available in Town Clerk's office.
- 2. Unless waived, all of the information required to demonstrate compliance with the performance standards set forth in Section 6.4.
- 3. When the Planning Board deems it necessary, it may require the applicant to provide at his expense supplemental data on traffic impact, including estimated daily and peak hour vehicle trips to be generated by the site, estimated number of employees and truck delivery schedule and hours of business.
- 4. When deemed necessary by the Planning Board, the applicant shall provide at his expense supplemental data and analysis on potential environmental impacts of the proposed project on air quality, surface and groundwater quality, site and neighborhood drainage conditions. The Planning Board may require that such studies be prepared by registered engineers or other appropriately qualified individuals.
- 5. When deemed necessary by the Planning Board, the applicant shall pay a review fee to cover the reasonable costs of consultants engaged by said Board to assist in review of the proposed site plan, as authorized and limited by G.L. c. 44, s. 53G and the Rules and Regulations of the Planning Board.
- **10.6.11 Abbreviated Site Plan Review; Plans and Submission Contents.** For projects qualifying for abbreviated site plan review under Section 10.6.3, the basic plan standards shall apply, and the plan shall show items 1, 2, 3, 5, and 7 of the plan requirements of Section 10.6.9 and only those elements of the remaining items that will be changed by the proposed construction or alteration. Along with the abbreviated site plan, the applicant shall submit a completed Site Plan Application Form available from the Town Clerk's office.

- 10.6.12 Site Plan Standards. The following standards shall be considered in the review and evaluation of a site plan or abbreviated site plan:
 - 1. The plan shall show compliance with Zoning By-law requirements for parking, signage, loading, landscaping, performance standards, dimensional requirements, and all other applicable requirements of this By-law;
 - 2. The proposed project should be compatible with existing natural features of the site, and compatible in architecture and scale with the surrounding area;
 - 3. The plan shall provide landscaping and open spaces that enhance the proposed development from within and without, which provide screening and buffers as necessary and which maximize amenity for customers, neighbors, and the general public;
 - 4. The proposed project shall provide for safe and convenient vehicular and pedestrian movement within the site in relation to adjacent ways, and provide for compliance with handicapped access requirements, and provide for access and egress by emergency vehicles;
 - 5. The proposed project shall provide for location and screening of exposed storage areas, machinery, service areas, dumpsters, utility Buildings and Structures and other unsightly uses to maximize amenity for neighborhood. The Planning Board may require landscaping and/or fencing to provide needed buffers;
 - 6. The proposed plan shall be consistent with the capacity of local infrastructure, such as water supply, utilities, drainage and Streets, or shall provide for such improvements as necessary; and
 - 7. The plan shall provide for compliance with Board of Health, Conservation Commission, and other Town and state regulations. Where applicable, the approval may be made conditional upon Board of Health, Conservation Commission, State Department of Public Works or other approvals, which shall be specifically referenced in the decision.
- **10.6.13 Final Action.** With regard to both site plans and abbreviated site plans, the Planning Board conducting the site plan review shall take one (1) of the following final actions:
 - 1. A written approval of the plan, with a determination that the proposal meets the criteria set forth in this By-law; or
 - 2. A written denial of the application stating the reasons for such denial; or
 - 3. A written approval subject to any conditions, modifications, and restrictions as the Planning Board may deem necessary, including phasing of construction and a performance guarantee for any public improvements.

The decision shall be filed with the Town Clerk.

- **10.6.14** Implementation of Site Plans. Construction under an approved site plan or abbreviated site plan shall begin within one (1) year and be completed within two (2) years of the date of the filing of the decision with the Town Clerk, unless extension is granted by the Planning Board. Otherwise approval shall lapse.
- 10.6.15 Appeal. The decision of the Planning Board may be appealed to a court of competent jurisdiction as set forth in G.L. c. 40A, s. 17.
- 10.7 SITE PLAN REVIEW FOR RELIGIOUS AND EDUCATIONAL USES AND CERTAIN CHILD CARE CENTERS.
- **10.7.1 Purpose.** The purpose of this Section is to provide for site plan review of religious and educational uses and child care centers otherwise subject to G.L. c. 40A, s. 3.
- 10.7.2 Site Plan Review Required. Prior to the issuance of any building permit or certificate of occupancy, the establishment, alteration, change, extension, or reconstruction of uses B.1, B.3, and B.7 as set forth in the Table of Use Regulations shall require site plan approval from the Planning Board pursuant to this Section.
- 10.7.3 Scope of Site Plan Review. Under this Section, site plan review shall be limited to two (2) inquiries:
 - 1. Whether the use qualifies for protection under G.L. c. 40A, s. 3; and, if so,
 - 2. What reasonable regulations concerning the bulk and height of Structures and determining Yard sizes, lot area, setbacks, open space, parking and building coverage requirements, if any, should be imposed on the use.
- **10.7.4 Required Information**. All applications for site plan review shall be in writing and provide, at a minimum, the following information:
 - 1. Name and address of applicant person or entity;
 - 2. Name and address of property owner;
 - 3. Description of the proposed use and any documents necessary to establish threshold compliance with G.L. c. 40A, s. 3;
 - 4. Reason that relief is requested from otherwise applicable zoning requirements; and
 - 5. If necessary to reach a decision on the application, the Planning Board may request further information from the applicant consistent with G.L. c. 40A, s. 3, specifying in detail the information required.

- **10.7.5 Site Plan; Contents.** In addition, the applicant shall submit a site plan with the following information:
 - 1. Legend depicting all pertinent existing and proposed site features;
 - 2. The date and north arrow shall be shown on the plans;
 - 3. All site plans must be stamped by a Registered Professional Civil Engineer and a Professional Land Surveyor. The land surveyor shall perform an instrument boundary survey and shall certify the accuracy of the locations of the Buildings, setbacks, and all other required dimensions to property lines;
 - 4. Zoning Chart depicting "Required" vs. "Provided" for all applicable zoning criteria including Lot size, Lot Frontage, setbacks, Building Height, Lot coverage, parking spaces, and landscaping requirements;
 - 5. Locus map, at a scale of 1"=600' or suitable scale to accurately locate the site in Town, oriented on the plan in the same way as the large scale plan;
 - 6. The location, width, status (public or private), and name of all streets within 100' of the project;
 - 7. On-site and abutting lot lines. On site lot lines shall be described by bearings and distance. Abutting lot lines shall be shown in a general way;
 - 8. Zoning District lines, including overlay districts if applicable;
 - 9. The location of existing or proposed Building (s) on the Lot shall be shown with total square footage and dimensions of all Buildings;
 - 10. Any streams, brooks, or wetland resource area boundaries within 100' of the property lines;
 - 11. Information on the location, size and type and number of existing and proposed landscape features;
 - 12. Information on the location, size and capacity of existing and proposed on-site and abutting utilities, (water, sewer, drainage, natural gas, electrical cable, etc.) including utilities in abutting side Streets, if applicable;
 - 13. Detailed locations and dimensions of all existing and proposed Buildings and uses on site and on abutting properties, including exterior details relating to the Building footprint.
 - 14. All existing and proposed setbacks from property lines.

- 15. All minimum, or below minimum, setback distances;
- 16. Information and details for all site and directional on-site signage shall be submitted;
- 17. Elevation and facade treatment plans of all proposed structures;
- 18. Color renderings for new construction;
- 19. Information on the location, size and type of parking, loading, storage and service areas. A parking calculation schedule noting existing, required and proposed spaces for the entire site shall be provided; and
- 20. Details and specifications (if applicable) for proposed site amenities, including but not limited to fences, recreation facilities, walls or other barrier materials, and special paving materials.
- **10.7.6 Decision.** The Planning Board may approve, approve with conditions, or deny an application for site plan approval. In making its decision, the Planning Board shall be guided exclusively by G.L. c. 40A, s. 3. The Planning Board shall file a written decision with the Town Clerk within sixty (60) days of receipt of the application. Failure to file a decision within sixty (60) days shall constitute approval of the site plan.
- 10.7.7 Appeal. Any appeal of the Planning Board's decision shall be made pursuant to G.L. c. 40A, s. 17, to a court of competent jurisdiction.

10.8 REQUEST FOR REASONABLE ACCOMMODATION.

- 10.8.1 Purpose. Under the requirements of the Federal Housing Act ("FHA"), it is a discriminatory practice to refuse to make "a reasonable accommodation in rules, policies, practices, or services when such accommodation may be necessary to afford [a handicapped] person equal opportunity to use and enjoy a dwelling." 42 U.S.C. § 3604(f)(3)(B). The same standard applies under the Americans With Disabilities Act ("ADA"), which also addresses nonresidential facilities providing services to persons with disabilities. 42 U.S.C. §12112(b)(5). The purpose of this Section is to facilitate housing and/or services for persons with disabilities and to comply fully with the spirit and the letter of the FHA and, where applicable, the ADA.
- **10.8.2 Request.** Any person eligible under the FHA or any provider of housing to persons eligible under the FHA, or any person eligible to operate a nonresidential facility providing services to persons eligible under the ADA, may request a reasonable accommodation as provided by the FHA and/or the ADA. A request for a reasonable accommodation does not affect a person's or provider's obligations to act in compliance with other applicable laws and regulations not at issue in the requested accommodation.
- **10.8.3 Zoning Board of Appeals.** All requests for reasonable accommodation under the FHA and/or the ADA shall be submitted to the Zoning Board of Appeals.

- **10.8.4 Information.** All requests for a reasonable accommodation under the FHA and/or the ADA shall be in writing and provide, at a minimum, the following information:
 - 1. Name and address of person (s) or entity requesting accommodation;
 - 2. Name and address of property owner;
 - 3. Name and address of dwelling or facility at which accommodation is requested:
 - 4. Description of the requested accommodation and specific regulation or regulations for which accommodation is sought;
 - 5. Reason that the requested accommodation may be necessary for the person or persons with disabilities to use and enjoy the premises; and
 - 6. If the requested accommodation relates to the number of persons allowed to occupy a dwelling, the anticipated number of residents, including facility staff (if any).

If necessary to reach a decision on the request for reasonable accommodation, the Zoning Board of Appeals may request further information from the applicant consistent with the FHA and/or ADA, specifying in detail the information required.

- 10.8.5 Zoning Board of Appeals Procedures. The Zoning Board of Appeals shall hold a public hearing using the procedures, including notice, set forth in G.L. c. 40A, ss. 11 and 15. The deadlines imposed in G.L. c. 40A, s. 11 or s. 15 may be extended upon the request of the applicant and the approval of the Zoning Board of Appeals. The Zoning Board of Appeals may seek information from other Town agencies in assessing the impact of the requested accommodation on the rules, policies, and procedures of the Town. Upon written notice to the Zoning Board of Appeals, an applicant for a reasonable accommodation may withdraw the request without prejudice. The Zoning Board of Appeals shall consider the following criteria when deciding whether a request for accommodation is reasonable:
 - 1. Whether the requested accommodation would require a fundamental alteration of a legitimate Town policy; and
 - 2. Whether the requested accommodation would impose undue financial or administrative burdens on the Town government.
- **10.8.6 Decision.** After conducting an appropriate inquiry into the request for a reasonable accommodation, the Zoning Board of Appeals may:
 - 1. Grant the request;
 - 2. Grant the request subject to specified conditions; or
 - 3. Deny the request.

The Zoning Board of Appeals shall issue a written final decision on the request in accordance with G.L. c. 40A, s. 15. If the Zoning Board of Appeals fails to render its decision on a request for a reasonable accommodation within the time allotted by G.L. c. 40A, s. 15, the request shall be deemed granted. The Zoning Board of Appeals' decision shall be filed with the Town Clerk and sent to the applicant by certified mail.

- **10.8.7** Appeal. The Zoning Board of Appeals' decision pursuant to this Section may be appealed to a court of competent jurisdiction in accordance with G.L. c. 40A, s. 17 or otherwise.
- **10.8.8** File. The Zoning Board of Appeals shall maintain a file of all requests for reasonable accommodation under the FHA and/or the ADA and a file of all decisions made on such requests. The file(s) may be reviewed in the office of the Zoning Board of Appeals upon request during regular business hours.
- 10.8.9 Other Laws. While a request for a reasonable accommodation is pending, all laws and regulations otherwise applicable to the premises that are the subject of the request shall remain in full force and effect.
- **10.8.10.** Effective Date. The provisions of this Section shall apply only to requests for reasonable accommodation made after adoption of this By-law. Any person who has previously submitted a request for a reasonable accommodation may resubmit the request for processing pursuant to the procedures set forth in this Section.

SECTION 11.0 DEFINITIONS

In this By-law, the following terms shall have the meanings described below:

Accessory Use or Building: A use or building which is subordinate and customarily incidental to and located on the same lot with the principal use or building to which it is accessory, except uses accessory to scientific research, scientific development or related production, as set forth in this By-law, which uses need not be located on the same lot as the principal use.

Adult Entertainment Use: An establishment, a building or portion thereof, or a use of land having a substantial or significant portion of its business activity, stock in trade, or other matter or materials for sale, rental, distribution, or exhibition which is distinguished or characterized by sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31, or a depiction, description or representation thereof, or emphasis thereon, including but not limited to adult bookstores, adult cabarets, adult motion picture theaters, adult paraphernalia stores, and adult video stores as hereafter defined.

Adult Bookstore: An establishment having as a substantial or significant portion of its stock in trade, books, magazines, or other matter which are distinguished or characterized by their emphasis depicting, describing, or relating to sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31. For the purpose hereof, the words

"substantial or significant" shall mean more than fifteen percent (15%) of the establishment's inventory of stock, or more than fifteen percent (15%) of the establishment's gross floor area.

Adult Cabaret: A night club, bar, restaurant, tavern, dance hall, or similar commercial establishment which features: persons who appear in a state of nudity; or live performances which are characterized by sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31; or films, motion pictures, video cassettes, slides or other photographic reproductions which are characterized by the depiction or description of sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31.

Adult Motion Picture Theater: An enclosed building used for presenting material distinguished by an emphasis on matter depicting, describing, or relating to sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31.

Adult Paraphernalia Store: An establishment having as a substantial or significant portion of its stock devices, objects, tools, or toys which are distinguished or characterized by their association with sexual activity, including sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31. For the purposes hereof, the words "substantial or significant" shall mean more than fifteen percent (15%) of the establishment's inventory of stock or more than fifteen percent (15%) of the establishment's gross floor area.

Adult Video Store: An establishment having a substantial or significant portion of its stock in trade, videos, movies, or other film material which are distinguished or characterized by their emphasis depicting, describing, or relating to sexual conduct or sexual excitement as defined in G.L. c. 272, s. 31. For the purposes hereof, the words "substantial or significant" shall mean more than fifteen percent (15%) of the establishment's inventory of stock, or more than fifteen percent (15%) of the establishment's gross floor area.

Adult Social Day Care: An accessory use that focuses on social activities, therapeutic recreation, meals, and some health-related services such as medication monitoring and blood pressure checks. May also provide assistance with daily activities such as toileting and walking. but not an adult day health facility or site as regulated by 404 CMR 402.

Boarder: An individual other than a family member occupying a dwelling unit who, for consideration, is furnished sleeping accommodations, meals, and may be provided personal care or other services, but excluding guests at a bed and breakfast.

Building: A structure having a roof or cover and forming a shelter for persons, animals or property.

Building Height: Except as otherwise provided herein, building height shall be measured as the vertical distance from the average elevation of the finished lot grade at the front of the building

to the highest point of the top story in the case of a flat roof, and to the mean height between the plate and the ridge in the case of a pitched roof.

Business or Professional Office: A business establishment, which does not offer a product or merchandise for sale to the public but offers a professional service to the public. However, general or personal service establishments are not to be included in the definition of business offices.

Child Care Center: A child care center as that term is defined in G.L. c. 15D, s. 1A.

Commercial Overlay District: The following definition shall apply in the Commercial Overlay District:

Building Height: The height of a Building shall be the vertical distance measured from the mean finished grade of the ground adjoining the front of the building, as determined by the Building Commissioner, to the top of the structure of the highest occupied floor in the case of a flat roof, to the deck line of a mansard roof, and to the top of the plate of a gable, hip or gambrel roof.

Commercial Recreation, Outdoors: Golf, tennis, swimming, riding, polo, skiing, skating, and all other athletic and recreational activities.

Drive-In or Drive-Through Establishment: A business establishment that includes service that is provided from a drive-up or drive-through window or other similar arrangement that allows the service of a patron while the patron remains in a vehicle, whether parked or live parked. The term shall include eating establishments and service establishments such as banks, dry cleaners, pharmacies, photo shops and the like.

Driveway: Any improved access for regular use by vehicle from the roadway of a street to a dwelling or other main building on a lot, or for use for which site plan approval or a special permit is required. A track or path used primarily by animals, or primarily for access of farm equipment to a field or orchard, or primarily for wood cutting, or for maintenance of a utility line is not a driveway. A driveway includes all of its branches.

Dwelling: A building or part thereof designed, erected and used for continuous and permanent habitation for one family or individual, but not including trailers, however mounted, or commercial accommodations offered for periodic occupancy.

Dwelling Unit: One or more rooms designed, occupied or intended for occupancy as separate living quarters, with cooking, sleeping and sanitary facilities provided within the dwelling unit for the exclusive use of a single family maintaining a household.

Dwelling, Single family Detached: A dwelling that is not attached to any other dwelling by any means and is surrounded by open space or yards on all sides, and that is occupied by not more than one household.

Dwelling, Semi-attached: A building containing two dwelling units that are attached to each other by a common vertical wall, each dwelling unit having open space or yards on three sides and each dwelling unit having direct access to the ground, and each dwelling unit is occupied by not more than one household.

Dwelling, Two-family: A building containing two dwelling units, either side by side or one above the other under a single roof, and each dwelling unit is occupied by not more than one household.

Dwelling, Townhouse: A building containing three or more dwelling units in a row, in which each dwelling unit has its own front and rear access to the ground, no dwelling unit is located over another dwelling unit, each dwelling unit is separated from any other dwelling unit by one or more party walls, and each dwelling unit is occupied by not more than one household.

Dwelling, Three-family: A building containing three dwelling units, each of which has direct access to the outside or to a common hall that leads to the outside.

Dwelling, Four-family: A building containing four dwelling units, each of which has direct access to the outside or to a common hall that leads to the outside.

Dwelling, Multifamily: A building containing five or more dwelling units, except as otherwise defined in this By-law.

Elderly: For the purposes of this By-law, persons who are 55 years of age or older, except as otherwise defined.

Electric Charging Station, Level Two: A facility equipped with a compatible cable such as J-1772, for the recharging of the batteries of motor vehicles.

Essential Services: Services provided by public utility or governmental agencies through erection, construction, alteration, or maintenance of underground or overhead gas, electrical, steam, or water transmission or distribution systems and collection, communication, supply, or disposal systems, excluding power plants or transfer stations. Facilities necessary for the provisions of essential services include poles, wires, mains, drains, sewers, pipes, conduits, cables, fire alarm boxes, police call boxes, traffic signals, hydrants, and other similar equipment and accessories in connection therewith. Specifically excluded from this definition are buildings necessary for the furnishing of adequate service by such public utility or governmental agencies for the public health, safety and general welfare.

Estate Overlay District: For the purposes of Section 9.4, the following definitions shall apply:

Combined or Aggregated: Parcels combined prior to July 1, 2010 to increase acreage to qualify for the Estate Overlay District By-law.

Commercial Use: A use that is primarily for profit and a nonresidential use, unless used in a hotel, inn or bed and breakfast, or nursing, assisted living or long term care facility that provides residential units for commercial or profitable use. The renting of apartments will not constitute a commercial use.

Common Ownership: Adjacent lots are considered to be in common ownership or within the landowner's legal control or power if one or more is held in an entity such as a limited partnership, trust benefitting the owners or landowner's use of adjoining land, even if the most recent instrument of record prior to July 1, 2010, reveals that the lot was separately owned and a previously recorded plan may reveal that the lot was at one time part of land held in common ownership.

Estate House: A building constructed prior to December 31, 1950, originally used as a single family residential dwelling, containing more than 5,000 square feet of residential floor area. Additions to the original dwelling that were constructed after 1950 will count toward the 5,000 square foot threshold if they were permitted by the Building Department prior to July 1, 2010.

Historically, Culturally or Architecturally Significant: A determination made by the Hamilton Historic District Commission that defines the property as one worth preserving to the level of the Secretary of the Interior's Standards for Rehabilitation and Guidelines for Rehabilitating Historic Buildings.

New Floor Area: The addition of building area calculated in square footage that can be added to the original estate house either as an attached or detached building located on the same property. Bonus square footage can be accumulated through (i) a yield plan, (ii) rehabilitation of the estate house to historic standards, or (iii) renovation of the estate house. Bonus Square Footage space can only be used for commercial purposes.

Nursing Home or Assisted Living for Seniors or the Infirm: A commercial use for the residential care of those who are infirm or elderly.

Rehabilitation: The process of returning a property to a state of utility, through repair or alteration, which makes possible an efficient contemporary use while preserving those portions and features of the property which are significant to its historic, architectural, and cultural values.

Secretary of the Interior's Standards for Rehabilitation and Guidelines for Rehabilitating Historic Buildings: The Standards for Rehabilitation (codified in 36 CFR 67 for use in the Federal Historic Preservation Tax Incentives program), which address the most prevalent treatment.

Yield Analysis Plan: The Yield Analysis plan is a schematic representation of a conventional subdivision. It shall show the maximum number of lots (or dwelling units) that could be placed upon the site under current zoning for a conventional subdivision plan. That maximum number of lots shall constitute the Base Maximum Density. The

Yield Analysis shall determine Base Maximum Density for the proposed development, given the presence of natural building constraints on the site such as wetlands, floodplains, Conservancy Districts, steep slopes, and zoning requirements. The proponent shall have the burden of proof with regard to the Basic Maximum Density.

Family: Any number of persons living together as a single economic unit and ordinarily using a single cooking facility.

Family Child Care Home, Large: An accessory use as defined in G.L. c. 15D, s. 1A.

Family Child Care Home, Small: An accessory use as defined in G.L. c. 15D, s. 1A.

Finding: For the purposes of Section 5.0, a "finding" shall mean the affirmative vote of three (3) members of the Zoning Board of Appeals.

Floodplain Overlay District: For the purposes of Section 9.2, the following definitions shall apply:

Area of Special Flood Hazard: The land in the floodplain within a community subject to a one percent or greater chance of flooding in any given year. The area may be designated as Zone A, AO, AH, Al- 30, AE, A99, Vl-30, VE, or V.

Base Flood: The flood having a one percent chance of being equaled or exceeded in any given year.

Coastal High Hazard Area: An area of special flood hazard extending from offshore to the inland limit of a primary frontal dune along an open coast and any other area subject to high velocity wave action from storms or seismic sources. The area is designated on a FIRM as Zone V, V 1-30, or VE.

Development: Any manmade change to improved or unimproved real estate, including but not limited to building or other structures, mining, dredging, filling, grading, paving, excavation or drilling operations.

District: The Floodplain Protection Overlay District.

Federal Emergency Management Agency (FEMA): The agency that administers the national Flood Insurance Program. FEMA provides a nationwide flood hazard area mapping study program for communities as well as regulatory standards for development in the flood hazard areas.

Flood Insurance Rate Map (FIRM): An official map of a community on which FEMA has delineated both areas of special flood hazard and the risk premium zones applicable to the community.

Flood Insurance Study: An examination, evaluation, and determination of flood hazards, and, if appropriate, corresponding water surface elevations, or an examination, evaluation and determination of flood-related erosion hazards.

Floodway: The channel of a river or other watercourse and the adjacent land areas that must be reserved in order to discharge the base flood without cumulatively increasing the water surface elevation.

Lowest Floor: The lowest floor of the lowest enclosed area (including basement or cellar). An unfinished or flood resistant enclosure, usable solely for parking vehicles, building access or storage in an area other than a basement area is not considered a building's lowest floor, provided that such enclosure is not built so as to render the structure in violation of the applicable non-elevation design requirements of NFIP Regulations 60.3.

Manufactured Home: A structure, transportable in one or more sections, which is built on a permanent chassis and is designed for use with or without a permanent foundation when connected to the required utilities. For floodplain management purposes the term "manufactured home" also includes park trailers, travel trailers, and other similar vehicles placed on a site for greater than 180 consecutive days. For insurance purposes, the term "manufactured home" does not include park trailers, travel trailers, and other similar vehicles.

Manufactured Home Park or Subdivision: A parcel (or contiguous parcels) of land divided into two or more manufactured home lots for rent or sale.

New Construction: For floodplain management purposes, structures for which the "start of construction" commenced on or after the effective date of a floodplain management regulation adopted by a community. For the purpose of determining insurance rates, new construction means structures for which the "start of construction" commenced on or after the effective date of an initial FIRM or after December 31, 1974, whichever is later.

One Hundred Year Flood: See Base Flood.

Regulatory Floodway: See Floodway.

Special Flood Hazard Area: An area having special flood and/or flood-related erosion hazards, and shown on an FHBM or FIRM as Zone A, AO, Al -30, AE, A99, AH, V, V 1-30, VE.

Structure: For floodplain management purposes, a walled and roofed building including a gas or liquid storage tank, that is principally above ground, as well as a manufactured home.

Structure: For insurance coverage purposes, means a walled and roofed building, other than a gas or liquid storage tank, that is principally above ground and affixed to a

permanent site, as well as a manufactured home on foundation. For the latter purpose, the term includes a building while in the course of construction, alteration, or repair, but does not include building materials or supplies intended for use in such construction, alteration, or repair, unless such materials or supplies are within an enclosed building on the premises.

Substantial Damage: Damage of any origin sustained by a structure whereby the cost of restoring the structure to its before-damaged condition would equal or exceed 50 percent of the market value of the structure before the damage occurred.

Substantial Improvement: Any repair, reconstruction, or improvement of a structure, the cost of which equals or exceeds 50 percent of the market value of the structure either (a) before the improvement or repair is started, or (b) if the structure has been damaged and is being restored, before the damage occurred. For the purposes of this definition, "substantial improvement" is considered to occur when the first alteration of any wall, ceiling, floor, or other structural part of the building commences, whether or not that alteration affects the external dimensions of the structure.

Zone A: The 100-year floodplain area where the base flood elevation (BFE) has not been determined. To determine the BFE, use the best available federal, state, local, or other data.

Zone Al-30 and Zone AE (for new and revised maps): The 100-year floodplain where the base flood elevation has been determined.

Zone AH and Zone AO: The 100-year floodplain with flood depths of 1 to 3 feet where a clearly defined channel does not exist, where the path of flooding is unpredictable, and where velocity flow may be evident. Such flooding is characterized by ponding or sheet flow.

Zone A99: Areas to be protected from the 100-year flood by federal flood protection system under construction. Base flood elevations have not been determined.

Zones B, C and X: Areas identified in the community Flood Insurance Study as areas of moderate or minimal flood hazard. Zone X replaces Zones B and C on new and revised maps.

Zone V: A special flood hazard area along a coast subject to inundation by the 100-year flood with the additional hazards associated with storm waves. Base flood elevations have not been determined.

Zone VI-30 and Zone VE (for new and revised maps): A special flood hazard area along a coast subject to inundation by the 100-year flood with additional hazards due to velocity (wave action). Base flood elevations have been determined.

For Profit Educational Facility: A facility offering education in fields such as music, dance, martial arts massage therapy, computer technology, the trades, and the like, but not otherwise regulated by G.L. c. 40A, s. 3.

Funeral Home: Facility for the conducting of funerals and related activities such as embalming.

Garage, Private: Covered space for the housing of motor vehicles, no more than two of which belong to others than the occupants of the lot on which such space is located.

General Service Establishment: Shop for lawn mower or small appliance repair, upholstery or furniture repair, bicycle repair person, printer, blacksmith, builder, carpenter, caterer, electrician, lawn mower service person, mason, painter, or plumber.

G.L.: Reference to the General Laws of Massachusetts.

Gross Floor Area, Business: The floor area within the perimeter of the outside walls of the building under consideration, without deduction for hallways, stairs, closets, thickness of walls, columns, or other features.

Gross Floor Area, Residential: The area of a residential structure used for living space, excluding basement and attic area.

Groundwater Protection Overlay District: For the purposes of Section 9.1, the following definitions shall apply:

Toxic or Hazardous Material: Any substance or mixture of physical, chemical or infectious characteristics posing a significant, actual or potential hazard to water supplies or other hazards to human health if such substance or mixture were discharged to land or water of the Town. Toxic or hazardous material include, without limitation, synthetic organic chemicals, petroleum products, heavy metals, radioactive or infectious wastes, acids or alkalis, and all substances defined as Toxic or Hazardous under G.L. c. 21C and c. 21E and 31O CMR 30.00, and also include such products as solvents and thinners in quantities greater than normal household use.

Inclusionary Housing: For the purposes of Section 8.7, the following terms shall have the following meanings:

Affordable Housing Trust Fund: A fund account established and operated by the Town for the exclusive purpose of creating or preserving affordable housing opportunities in the Town of Hamilton.

Affordable Housing Unit: A dwelling, or a unit in an assisted living facility or congregate residence, that is affordable to and occupied by a low- or moderate-income

household and meets the requirements of the Local Initiative Program for inclusion on the Chapter 40B Subsidized Housing Inventory.

Affordable Housing Restriction: A contract, mortgage agreement, deed restriction, or other legal instrument, acceptable in form and substance to the Town of Hamilton, that effectively restricts occupancy of an affordable housing unit to qualified purchaser or qualified renter, and which provides for administration, monitoring and enforcement of the restriction during the term of affordability. An affordable housing restriction shall run with the land in perpetuity or for the maximum period of time allowed by law, so as to be binding on and enforceable against any person claiming an interest in the property. An affordable housing restriction shall be enforceable under the applicable provisions of G.L. c.184, as amended, and be approved by the Department of Housing and Community Development through the Local Initiative Program.

Local Initiative Program: A program administered by the Massachusetts Department of Housing and Community Development (DHCD) pursuant to 760 CMR 45.00 to develop and implement local housing initiatives that produce low and moderate income housing.

Low Income Household: A household with income at or below 50% of area median income, adjusted for household size, for the metropolitan or non-metropolitan area that includes the Town of Hamilton as determined annually by the United States Department of Housing and Urban Development (HUD).

Maximum Affordable Purchase Price or Rent: A selling price or monthly rent that does not exceed the maximum allowable purchase price or rent guidelines of the Local Initiative Program.

Moderate Income Household: A household with income at or below 80% of area median income, adjusted for household size, for the metropolitan or non-metropolitan area that includes the Town of Hamilton as determined annually by the United States Department of Housing and Urban Development (HUD).

Qualified Purchaser: A low- or moderate-income household that purchases and occupies an affordable housing unit as its principal residence.

Qualified Renter: A low or moderate-income household that rents and occupies an affordable housing unit as a tenant.

Subsidized Housing Inventory: The Department of Housing and Community Development Chapter 40B Subsidized Housing Inventory.

Kennel: An establishment as defined in G.L. c. 140, s. 137A.

Long-Term Care Facility: Any institution whether conducted for charity or profit which is advertised, announced or maintained for the express or implied purpose of providing three or more individuals admitted thereto with long-term resident, nursing, convalescent or rehabilitative care; supervision and care incident to old age for ambulatory persons; or retirement home care for elderly persons. Long-term care facility shall include convalescent or nursing homes, rest homes, infirmaries maintained in towns and charitable homes for the aged, as set forth in 105 CMR 150.001.

Lot: An area of land in one ownership with definite boundaries, used, or available for use, as site for one or more buildings.

Lot Frontage: Shall be measured along the property line of (I) a public way or a way which the Town Clerk certifies is maintained and used as a public way, or (ii) a way not less than fifteen feet (15) from the center line shown on a plan approved and endorsed in accordance with the subdivision control law, which shall thereafter be considered a street for the purpose of this By-law or (iii) a way in existence when the subdivision control law became effective in Hamilton (1955) having in the opinion of the Planning Board, sufficient width, suitable grades, and adequate construction to provide for the needs of vehicular traffic and to provide adequate access for fire, police, and emergency vehicle.

In the case of creating a new street, frontage will be measured along the property line of the existing right of way of the existing road from the side line of the lot up to the point where the curve, based on a 25' radius, begins to separate from the property line to meet the proposed right of way of the new street.

Manufacturing: A use engaged in the basic processing and manufacturing of materials, or the manufacture from previously prepared materials, of finished products or parts, including processing, fabrication, assembly, treatment, packaging, incidental storage, sales and distribution of such products.

Medical Clinic: A facility as defined in 105 CMR 145.020, including a mobile clinic.

Medical Office: A building designed and used as an office by physicians, dentists, or psychotherapists for the diagnosis and treatment of human patients that does not include overnight care facilities or licensing as a clinic.

Mining: The removal or relocation of geologic materials such as topsoil, sand, gravel, metallic ores or bedrock.

Mixed Use: Two or more uses on a single parcel as a part of a single development plan.

Motor Vehicle Hourly Rental Station: A facility at which, by contract, motor vehicles are made available for rent for a period not longer than 24 hours.

Motor Vehicle Light Service Station: A building or premises used for the dispensing, sales or offering for sale of motor fuels directly to users of motor vehicles. Other sales activities and any repairs shall be activities minor in scope and clearly subordinate to the sale of motor fuels, oils and lubricants. No drive-through facilities are permitted without further zoning relief.

Motor Vehicle Repair: A building or use which is designed or intended to be used for the storage, servicing, repair, maintenance, or cleaning of motor vehicle equipment.

Motor Vehicle Sales: Premises for the sale of used or new motor vehicles, including supplying of fuel, oil, lubrication, washing, or repair services, but not to include body work or painting.

Open Space and Farmland Preservation Development: For the purposes of Section 8.1, the following terms shall have the following meanings:

Contiguous Open Space: Open space suitable, in the opinion of the Planning Board, for the purposes set forth in Section 8.1.1, herein. Such open space may be separated by the road(s) constructed within the Development. Contiguous open space shall not include required yards.

Personal Service Establishment: A facility providing personal services such as hair salon, barber shop, tanning beds, dry cleaning, print shop, photography studio, personal fitness center, and the like.

Qualified Acre: Agricultural land on which the sale of products produced from the agriculture, aquaculture, silviculture, horticulture, floriculture or viticulture use on the parcel annually generates at least \$1,000 per acre based on gross sales dollars in area not zoned for agriculture, aquaculture, silviculture, horticulture, floriculture or viticulture.

Research Laboratory: Laboratory or research establishments including biotechnology companies, but excluding laboratories categorized as Level 4 by the National Institutes for Health.

Restaurant: A building, or portion thereof, containing tables and/or booths for at least two-thirds (2/3) of its legal capacity, which is designed, intended and used for the indoor sales and consumption of food prepared on the premises, except that food may be consumed outdoors in landscaped terraces, designed for dining purposes, which are adjuncts to the main indoor restaurant facility. The term "restaurant" shall not include "fast food establishments."

Restaurant, fast-food: Establishment for the immediate sale of food or drink prepared on or off premises and served in disposable containers or wrappers for consumption on or off premises unless such sales are wholly incidental to a conventional restaurant or other use such as a grocery or convenience store or food market or other use defined in this chapter. Service is usually cafeteria style or from a serving counter. Such establishment may include inside seating, but

table service is usually not provided or only incidental. All restaurant establishments providing in-car, drive-through service are included in this definition.

Retail: A facility selling new or used goods to an end user but not otherwise specifically listed in the Table of Use Regulations.

School Aged Child Care Program: A school aged child care program as that term is defined in G.L. c. 15D, s. 1A.

Site Plan Review: Review by the Planning Board to approve or approve with conditions those land use activities set forth in Section 10.6.

Story: Shall mean the portion of a building being between the upper surface of any floor and the upper surface of the floor next above, having more than one half of its height above the average elevation of the finished grade adjoining the building, provided that any part of a building between the topmost floor and the roof shall be deemed a half-story.

Street: A public way, a way opened or dedicated to public use or a way plotted and laid out for ultimate public use, whether or not constructed.

Street Line: The sideline of a street or way, as determined by deeds and plans recorded at the Registry of Deeds, or a building line laid out under G.L. c. 82, s. 37. Where no line is thus legally established, then a line parallel with and twenty-five feet distant from the center line of a traveled way.

Structure: Any construction, erection, assemblage or other combination of materials upon the land made in such a manner as to indicate a purpose that it remain in position indefinitely.

Temporary Additional Living Area: See Section 8.1.

Trailer or Mobile Home: Any structure or vehicle designed or adapted for human habitation which is capable of being moved from one place to another, whether by being towed or by being transported.

Truck Terminal: A facility for handling freight with or without maintenance facilities.

Veterinary facility or clinic: A place where animals or pets are given medical or surgical treatment and the boarding of animals is limited to short term care incidental to the use.

Warehouse: A building used primarily for the storage of goods and materials, for distribution, but not for sale on the premises.

Warehouse, Mini or Self Storage: Establishment providing individual storage units for long or short term rental to persons or businesses.

Wholesale: Sale of goods not at retail.

Wind Energy Conversion System: A device for converting wind motion into usable domestic energy.

Wind Energy Facilities: For the purposes of Section 7.5, the following definitions shall apply:

As of Right Siting: Development may proceed without the need for a special permit, variance, amendment, waiver, or other discretionary approval if the project is located in the designated location (Commercial Overlay District). As of right development will be subject to nondiscretionary site plan review to determine conformance with local zoning by-laws as well as state and federal law. As of right development projects that are consistent with this By-law and with state and federal law cannot be prohibited, but may be subject to reasonable conditions.

Designated Location: The Commercial Overlay District (COD). Said location is shown on the Hamilton Zoning Map, which may be revised from time to time. This map is part of the Zoning By-law and is on file in the Office of the Town Clerk.

Height: The height of a wind turbine measured from natural grade to the tip of the rotor blade at its highest point, or blade-tip height.

Rate Nameplate Capacity: The maximum rated output of electric power production equipment. This output is typically specified by the manufacturer with a "nameplate" on the equipment.

Site Plan Review Authority: When the wind energy facility is located within the Designated Location, the Site Plan Review Authority will be the Planning Board.

Special Permit Granting Authority: When the wind energy facility is located outside the Designated Location, the SPGA shall be the Zoning Board of Appeals.

Small Scale Wind Energy Facility: A wind energy facility with a blade-tip height up to 150 feet as measured from existing average grade.

Utility Scale Wind Energy Facility: A commercial wind energy facility, where the primary use of the facility is electrical generation which may or may not be sold to the wholesale electricity markets and has a blade-tip height over 150 feet as measured from existing average grade.

Wind Energy Facility: All of this equipment, machinery and structures together utilitized to convert wind to electricity. This includes, but is not limited to, transmission,

storage, collection and supply equipment, substations, transformers, service and access roads, and one or more wind turbines.

Wind Monitoring or Meteorological Tower: A temporary tower equipped with devices to measure wind speed and direction, to determine how much electricity a wind energy facility can be expected to generate.

Wind Turbine: A device that converts kinetic wind energy into rotational energy to drive an electrical generator. A wind turbine typically consists or a tower, nacelle body, and a rotor with two or more blades.

Yard: A space open to the sky, located between a building or structure and a lot line, unoccupied except by fences, walls, poles, paving, and other customary yard accessories.

Yard, front: A yard extending the full width of the lot and situated between the street line and the nearest point of the building.

Yard, rear: A yard the full width of the lot and situated between the rear line of the lot and the nearest part of the main building projected to the side line of the lot.

Yard, side: A yard situated between the nearest point of the building and the side line of the lot and extending from the front yard to the rear yard. Any lot line not a rear line or a